

Creada por Acuerdo Nº 5 de 1987 del Concejo Municipal

Dirección Secretaría de Servicios Administrativos

Coordinación
Archivo General
Alcaldía de Medellín

Medellín, Diciembre 30 de 2015

CONTENIDO	
EXECUTION VERSION - EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P.,	Pág. 2
as Borrower, THE LENDERS PARTY HERETO and THE BANK OF TOKYO-	
MITSUBISHI UFJ, LTD., as Administrative Agent	
AMENDMENT TO THE LOAN AGREEMENT	54
CONTRATO DE EMPRÉSTITO INTERNO PARA OPERACIÓN DE CREDITO	55
TRANSITORIO CELEBRADO ENTRE LA CENTRAL HIDROELECTRICA DE	
CALDAS S.A E.S.P. Y BANCO DAVIVIENDA S.A.	
RESOLUCIÓN 3952 de 2015 (Diciembre 29)	59
Por medio del cual se realizan unos ajustes a la resolución De la Secretaría del Medio	
Ambiente y se dictan otras disposiciones.	
RESOLUCIÓN 3635 DE 2015 (Diciembre 4)	67
Por medio del cual se modifican unos Actos Administrativos en el Municipio de	
Medellín	
RESOLUCIÓN 3838 DE 2015 (Diciembre 18)	72
Por medio del cual se aclaran los Artículos 405 y 406 de la Resolución 3728 de 2015,	
los Artículos 143 y 149 de la Resolución 3730 de Diciembre 14 de 2015 y se adiciona	
el Artículo 95 de éste último Acto Administrativo.	
RESOLUCIÓN 3945 DE 2015 (Diciembre 29)	73
Por medio del cual se realizan unos traslados en la Administración Municipal,	
y se dictan otras disposiciones.	

EXECUTION VERSION

U.S.\$1,000,000,000

CREDIT AGREEMENT

among

EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P., as **Borrower**,

THE LENDERS PARTY HERETO

and

THE BANK OF TOKYO-MITSUBISHI UFJ, LTD.,
as Administrative Agent

Dated as of December 29, 2015

BANK OF AMERICA, N.A.; and HSBC SECURITIES (USA) INC.

as Global Coordinators and Mandated Lead Arrangers and Joint Bookrunners
THE BANK OF TOKYO-MITSUBISHI UFJ, LTD.; and
SUMITOMO MITSUI BANKING CORPORATION
as Mandated Lead Arrangers and Joint Bookrunners
CITIGROUP GLOBAL MARKETS INC.;
JP MORGAN SECURITIES LLC.; and
MIZUHO BANK, LTD.
as Lead Arrangers

TABLE OF CONTENTS

		<u>Page</u>
SECTION 1.	DEFINITIONS AND PRINCIPLES OF CONSTRUCTION	1
1.01	Defined Terms	1
1.02	Principles of Construction	14
SECTION 2.	THE LOANS	15
2.01	Commitments	15
2.02	Mechanics for Requesting a Loan	15
2.03	Funding	15
2.04	Minimum Amounts for Each Disbursement	16
2.05	Termination and Reduction of Commitments	16
2.06	Notes	16
2.07	Interest	18
2.08	Fees	19
2.09	Additional or Increased Costs	20
2.10	Breakage Costs, Other Expenses and Losses	21
2.11	Illegality	21
2.12	Replacement of Lenders	21
SECTION 3.	PAYMENTS	22
3.01	Repayment; Time, Place and Manner	22
3.02	Voluntary Prepayment	22
3.03	Mandatory Prepayment	23
3.04	Payments; Pro Rata Treatment	23
3.05	Extension of Payment Dates	24
SECTION 4.	TAXES	24
4.01	Covered Taxes	24
SECTION 5.	CONDITIONS PRECEDENT	26
5.01	Conditions Precedent to Effectiveness	26
5.02	Conditions Precedent to Each Disbursement	28
SECTION 6.	REPRESENTATIONS AND WARRANTIES	28
6.01	Representations and Warranties of the Borrower	28
SECTION 7.	COVENANTS	32
7.01	Affirmative Covenants of the Borrower	32
7.02	Negative Covenants of the Borrower	35
SECTION 8.	EVENTS OF DEFAULT AND REMEDIES	36
8.01	Events of Default	36
8.02	Remedies	38
SECTION 9.	GOVERNING LAW AND JURISDICTION	39
9.01	Governing Law	39
9.02	Submission to Jurisdiction	39
9.03	Service of Process	39
9.04	Waiver of Immunity	40
9.05	Waiver of Security Requirements	41
9.06	No Limitation	41

SECTION 10.	IHE A	DMINISTRATIVE AGENT	41
10.01	Appoin	ntment	41
10.02	Nature	e of Duties	41
10.03	Lack o	f Reliance on the Administrative Agent	42
10.04	Relian	ce	43
10.05	Consu	Itation with Experts	43
10.06	Indemi	nification	43
10.07	The Ac	dministrative Agent in Its Individual Capacity	43
10.08	Resign	nation by the Administrative Agent; Successor Administrative Agent	43
10.09	No Am	endment to Duties of Administrative Agent Without Consent	44
SECTION 11.	MISCE	ELLANEOUS	44
11.01	Compu	utations	44
11.02	Notices	s	45
11.03	Benefit	t of Agreement; Assignment; Participations	46
11.04	No Wa	niver; Remedies Cumulative	48
11.05	Entire A	Agreement	48
11.06	Amend	dment or Waiver; etc.	49
11.07	Counte	erparts	49
11.08	Expens	ses; Indemnity	50
11.09	Judgm	ent Currency	51
11.10	English	h Language	52
11.11	Severa	ability	52
11.12	Waiver	r of Jury Trial	52
11.13	Captio	ns	52
11.14	Damag	ges Waiver	53
11.15	Confid	entiality	53
11.16	Surviva	al	53
11.17	No Fid	luciary Duty	54
11.18	PATRI	OT Act	54
11.19	Exequa	atur	54
11.20	Interna	ational Banking Facilities	54
SCHEDULES			
Schedule 1.01-	<u> </u>	Commitments	
Schedule 1.01-	<u>II</u> –	Existing Liens	
Schedule 6.01(<u>b)</u> –	Governmental Approvals	
EXHIBITS			
Exhibit A	_	Form of Assignment and Assumption Agreement	
Exhibit B	_	Form of Instructions Letter	
Exhibit C	_	Form of Loan Request	
Exhibit D	_	Form of Promissory Note	
Exhibit E	_	Form of Officer's Certificate	
Exhibit F	_	Form of Legal Opinion of Milhank, Tiyand Hadlay & McClay	
Exhibit G	_	Form of Legal Opinion of Milbank, Tweed, Hadley & McCloy	

INDEX OF DEFINED TERMS

	<u>Page</u>
Administrative Agent	1
Affiliate	1
Agent Fee Letter	1
Agreement	1
Agreement Date	1
Alternative Base Rate	1
Anti-Corruption Laws	2
Anti-Money Laundering Laws	2
Applicable Law	2
Applicable Margin	2
Assignment and Assumption Agreement	2
Authorized Officer	2
Availability Period	2
BASEL III	2
Borrower	1
Borrower Financial Statements	2
Business Day	3
Capital Adequacy Requirement	3
Capitalized Lease Obligation	3
Change of Control	3
Closing Date	26
Code	3
Code of Corporate Governance	3
Colombia	3
Colombian Central Bank	3
Colombian Government Entity GAAP	3
Colombian Peso	3
Commercial Code	4
Commitment	4
Confidential Information	4
Connection Income Taxes	4
Consolidated Total Assets	4
Control	4
Controlled	4
Controlling	4
Covered Taxes	4
Disbursement Date	4
Dollars	5
Environmental Report	5
Event of Default	36
Excluded Taxes	5
Existing Notes	18

External Indebtedness	5
Fair Market Value	5
FATCA	5
FCPA	6
Federal Funds Effective Rate	6
Fee Letters	6
Fitch	6
Good Faith Contest	6
Governmental Approval	6
Governmental Authority	6
IFRS	7
Impacted Interest Period	8
Indebtedness	7
Indemnitee	50
Instructions Letter	7
Interest Determination Date	7
Interest Payment Date	7
Interest Period	7
Internal Indebtedness	7
Interpolated Rate	8
Ituango Project	8
Judgment Currency	51
Judgment Currency Conversion Date	51
Law 550	32
Lender	1
Lender Fee Letter	8
Lenders	1
Liabilities	43
LIBO Screen Rate	8
LIBOR	8
Lien	8
Loan	15
Loan Documents	8
Loan Facility Account	15
Loan Request	15
Loans	8
Margin Stock	8
Material Adverse Effect	8
Material Subsidiary	9
Maturity Date	9
Ministry of Finance	9
Moody's	9
Note	9
Notices	45

OFAC	9	
Other Connection Taxes	9	
Other Taxes	9	
Participant Register	48	
PATRIOT Act	9	
Permitted Lien	9	
Person	12	
Potential Default	12	
Proportionate Share	12	
Register	48	
Regulation D	54	
Regulation U	12	
Regulatory Change	12	
Related Parties	13	
Replaced Lender	22	
Replacement Lenders	22	
Required Lenders	12	
S&P	13	
Sanctioned Country	13	
Sanctioned Person	13	
Sanctions	13	
Sanctions Authorities	13	
Subsidiary	13	
Tax	13	
Taxing Jurisdiction	13	
Total Commitment	14	
U.S.	14	
U.S. \$	5	
U.S. Dollars	5	
United States	14	
Wholly Owned	14	

This CREDIT AGREEMENT (together with the Schedules and Exhibits attached hereto, this "Agreement"), dated as of December 29, 2015 is made by and among EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P., a public utility services company organized and existing under the laws of the Republic of Colombia (the "Borrower"); each of the lenders identified under the caption "Lender" on the signature pages hereof or which, pursuant to Section 11.03 hereof, shall become a lender hereunder (each a "Lender" and, collectively, the "Lenders") and THE BANK OF TOKYO-MITSUBISHI UFJ, LTD., as administrative agent (in such capacity, the "Administrative Agent"). Capitalized terms used herein shall be defined as provided in Section 1.01.

BACKGROUND

WHEREAS:

- (A) by this Agreement, the Borrower has requested that the Lenders provide a term loan facility to it in accordance with the terms and conditions set forth herein for an aggregate principal amount up to one billion Dollars (U.S. \$1,000,000,000);
- $\ensuremath{(B)}$ the Lenders are willing to do so on the terms and conditions set forth herein; and
- (C) the Borrower has obtained all required approvals, including the approval from the Ministry of Finance through Resolution No. 1348 of May 17, 2012, Resolution No. 3071 of September 2, 2014 and Resolution No. 4783 of December 22, 2015, to enter into this Agreement.

NOW, THEREFORE, in consideration of the premises and the mutual covenants herein contained, the parties hereto agree as follows:

SECTION 1. DEFINITIONS AND PRINCIPLES OF CONSTRUCTION

1.01 Defined Terms

- . For the purposes of this Agreement, unless otherwise defined herein, the following terms have the meanings specified below.
- "Affiliate" means, with respect to any Person, any other Person that, directly or indirectly through one or more intermediaries, controls, or is Controlled by, or is under common control with, such first Person.
- "Agent Fee Letter" means the fee letter executed on or prior to the Agreement Date between the Borrower and the Administrative Agent.
- "Agreement Date" means the date first set forth above, such date being the date as of which this Agreement was executed and delivered by the parties hereto.
- "Alternative Base Rate" means for any day, a fluctuating rate per annum equal to the highest of (a) the Federal Funds Effective Rate plus 0.50%, (b) the prime rate published in *The Wall Street Journal* for such day; *provided* that, if *The Wall Street Journal* ceases to publish for any reason such rate of interest, "Alternative Base Rate" shall mean the prime lending rate as set forth on the Bloomberg page PRIMBB Index (or successor page) for such day (or such other service as determined by the Administrative Agent from time to time for purposes of providing quotations of prime lending interest rates) and (c) the LIBOR for an Interest Period of one month plus one percent (1%). Each change in the Alternative Base Rate shall be effective on the date such change is effective.
- "Anti-Corruption Laws" means (a) the FCPA, (b) the UK Bribery Act of 2010, and the rules and regulations promulgated thereunder, (c) Law 1474 of 2011 (*Estatuto Anticorrupción*) of Colombia, and (d) all other laws, rules, and regulations of any jurisdiction applicable to the Borrower or any of its Subsidiaries concerning or relating to bribery or corruption.
- "Anti-Money Laundering Laws" means all laws of any jurisdiction applicable to the Borrower or its Subsidiaries concerning or relating to anti-money laundering and anti-terrorism financing.
- "Applicable Law" means, with respect to any Person, any Colombian or other applicable constitution, statute, law, rule, regulation, ordinance, judgment, order, decree, or any published directive, guideline, requirement or other governmental rule or restriction which has the force of law, and any determination by, or interpretation of any of the foregoing by, any judicial authority or Governmental Authority, binding on a given Person whether in effect as of the date hereof or as of any date thereafter.

- "Applicable Margin" means 1.40% per annum.
- "Assignment and Assumption Agreement" means an assignment and assumption agreement in substantially the form of Exhibit A.
- "Authorized Officer" means, with respect to any Person, the chief executive officer, the president, any vice president, any assistant vice president, the chief financial officer or treasurer, the assistant treasurer or equivalent officers of such Person, and any other officer or representative of such Person, who is duly authorized to act under such Person's charter documents or Applicable Law, and to act in the capacity in which they are acting pursuant to the certificate referred to in Section 5.01(b).
- "Availability Period" means the period from and including the Agreement Date to and including the earlier of (i) the first anniversary of the date hereof and (ii) the date on which the Total Commitment is reduced to zero or otherwise terminated.
- "BASEL III" means (a) the agreements on capital requirements, a leverage ratio and liquidity standards contained in "Basel III: A global regulatory framework for more resilient banks and banking systems," "Basel III: International framework for liquidity risk measurement, standards and monitoring" and "Guidance for national authorities operating the countercyclical capital buffer" published by the Basel Committee on Banking Supervision in December 2010, each as amended, supplemented or restated and (b) any further guidance or standards published by the Basel Committee on Banking Supervision relating to Basel III.
- "Borrower Financial Statements" means the audited consolidated financial statements of the Borrower and its consolidated Subsidiaries dated December 31, 2014, and the unaudited unconsolidated financial statements of the Borrower and its consolidated Subsidiaries as of September 30, 2015.
- "Business Day" means any day, except a day which is a Saturday or a Sunday, on which (a) the Federal Reserve Bank of New York is open for business, (b) for purposes of determining LIBOR, dealings in U.S. Dollar deposits are carried on in the London interbank market and (c) commercial lenders in Bogota, Colombia and New York, New York (United States) are open for domestic and foreign exchange business.
- "Capital Adequacy Requirementb" means, with respect to any Person, any requirement of law or any regulation affecting the amount of capital required or expected to be maintained by such Person (or the lending office of such Person) or any Person Controlling such Person.
- "Capitalized Lease Obligation" means, with respect to any Person as of the date of determination, the obligation to pay rent or other amounts under a lease with respect to any property (whether real, personal or mixed) acquired or leased (other than leases for transponders) by

such Person and used in its business that is required to be accounted for as a liability on the balance sheet of such Person in accordance with Colombian Government Entity GAAP or IFRS, as applicable, and the amount of such Capitalized Lease Obligation shall be the amount so required to be accounted for as a liability.

"Change of Control" means an event or series of events by which the Municipality of Medellin and/or the Republic of Colombia ceases to own, directly or indirectly, beneficially and of record, at least 50.1% of the outstanding shares of voting stock of the Borrower or shall cease to Control the Borrower.

"Code" means the Internal Revenue Code of 1986 and the regulations promulgated and rulings issued thereunder, as amended.

"Code of Corporate Governance" means the Borrower's *Código de Buen Gobierno Corporativo* as in effect from time to time.

"Colombia" means the Republic of Colombia (República de Colombia).

"Colombian Central Bank" means the Central Bank of Colombia (Banco de la República de Colombia) or any other Governmental Authority of Colombia charged with the responsibility of issuing, managing and controlling legal currency in Colombia and determining Colombian foreign exchange policy.

"Colombian Government Entity GAAP" means accounting principles for government-owned entities issued by the National Accounting Office (Contaduría General de la Nación) and other applicable legal provisions in effect from time to time, as modified, amended and/or supplemented by IFRS upon adoption of IFRS by the Borrower and its Subsidiaries in accordance with Applicable Law.

"Colombian Peso" means the lawful currency of Colombia.

"Commercial Code" means the Colombian Commercial Code (Código de Comercio), as amended.

"Commitment" means, for each Lender, the obligation of such Lender to make Loans to the Borrower hereunder during the period commencing on the date hereof to, but excluding the date on which the Availability Period ends, in a principal amount up to but not exceeding the amount set forth opposite such Lender's name on <u>Schedule 1.01-1</u> hereto; *provided* that such amount shall be reduced by any reduction or termination of the Total Commitment by the Borrower pursuant to <u>Section 1.05</u>.

"Confidential Information" means information that the Borrower furnishes to the Administrative Agent or any Lender in writing, but does not include any such information (a) that is or becomes generally available to the public, (b) that is or becomes available to the Administrative Agent or any Lender from a source other than the Borrower on a non-confidential basis, (c) is in the Administrative Agent's or any Lender's possession including any such information in respect of which the Administrative Agent or Lender reasonably believes that it is not bound by any confidentiality obligations to the Borrower (but in any event excluding any information furnished by the Borrower that is designated as confidential in writing) or (d) is independently developed by the Administrative Agent or any Lender without reference to such information.

"Connection Income Taxes" means Other Connection Taxes that are imposed on or measured by net income (however denominated) or that are franchise Taxes or branch profits Taxes.

"Consolidated Total Assets" means, at any date, the total amount of assets of the Borrower and its consolidated Subsidiaries, as of the end of the last period preceding such date for which a balance sheet is prepared and published in accordance with Applicable Law, calculated on a consolidated basis as determined in accordance with Colombian Government Entity GAAP or IFRS, as applicable.

"Control" means, in relation to any specified Person, (a) holding, directly or indirectly, fifty percent (50%) or more of the outstanding voting securities or ownership interests of such specified Person or (b) having the contractual power to designate a majority of the directors of a corporation, or in the case of an unincorporated entity, a majority of the individuals who exercise similar functions of such specified Person (and "Controlled" and "Controlling" shall be construed accordingly).

"Covered Taxes" means, with respect to the Administrative Agent or any Lender, (a) Taxes, other than Excluded Taxes or Other Connection Taxes, imposed by a Taxing Jurisdiction on or with respect to any payment made by or on account of any obligation of the Borrower under any Loan Document, and (b) to the extent not otherwise described in (a), Other Taxes.

"Disbursement Date" means each date of disbursement of Loans specified in a Loan Request; *provided* that such Disbursement Date shall be a Business Day no later than the last day of the Availability Period.

"Dollars," "U.S. Dollars," or "U.S. \$" means the lawful currency of the United States of America.

"Environmental Report" means a report describing all material aspects concerning social and environmental matters related to the Ituango Project that may have risen during the immediately preceding fiscal year of the Borrower, including, but not limited to, (a) status of land negotiation, acquisition and compensation by the Borrower, (b) acceptance or opposition towards Project Ituango by communities within its area of influence or any representative organization and (c) compensation by the Borrower for ecological repairs.

"Excluded Taxes" means any of the following Taxes imposed on or with respect to any Lender or the Administrative Agent or required to be withheld or deducted from a payment to any Lender or the Administrative Agent, (a) Taxes imposed on or measured by net income (however denominated), franchise Taxes, and branch profits Taxes. in each case (i) imposed as a result of such Lender or Administrative Agent being organized under the laws of, or having its principal office or, in the case of any Lender, its applicable lending office located in, the jurisdiction imposing such Tax (or any political subdivision thereof) or (ii) that are Other Connection Taxes, (b) in the case of a Lender, withholding Taxes imposed on amounts payable to or for the account of such Lender with respect to an applicable interest in the Loan pursuant to a law in effect on the date on which (i) such Lender acquires such interest in the Loan (other than pursuant to an assignment request by the Borrower under Section 2.12) or (ii) such Lender changes its lending office, except in each case to the extent that, pursuant to Section 4.01, amounts with respect to such Taxes were payable either to such Lender's assignor immediately before such Lender became a party hereto or to such Lender immediately before it changed its lending office, (c) Taxes that would not have been imposed but for the Lender's or the Administrative Agent's failure to make any disclosure with the relevant taxing authority as required by Applicable Law, unless such Lender or Administrative Agent has reasonably and in good faith determined that any such disclosure would expose it to any material adverse effect, or to provide any reasonably requested documentation and/ or certification to the Borrower or any other Person or (d) any withholding Taxes imposed due to a Lender's failure to comply with FATCA.

"External Indebtedness" means Indebtedness of the Borrower other than Internal Indebtedness.

"Fair Market Value" means, with respect to any asset or property, the price which could be negotiated in an arm's-length transaction, for cash, between an informed and willing seller under no compulsion to sell and an informed and willing buyer under no compulsion to buy.

"FATCA" means (1) Sections 1471 through 1474 of the Code, as of the date of this Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with) and any current regulations or official interpretations thereof, (2) any treaty, law or regulation of any other jurisdiction or relating to an intergovernmental agreement between the U.S. and any other jurisdiction, which (in either case) facilitates the implementation of any law or regulation referred to in clause (1) above, or (3) any agreements entered into pursuant to Section 1471(b)(1) of the Code or pursuant to the implementation of any treaty, law or regulation referred to in clause (1) or (2) above with the U.S. Internal Revenue Service, the U.S. government or the government or tax authority of any other jurisdiction.

"FCPA" means the Foreign Corrupt Practices Act of 1977, as amended, and the rules and regulations thereunder.

"Federal Funds Effective Rate" means, for any day, the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers as published on the next succeeding Business Day by the Federal Reserve Bank of New York or, if such rate is not so published for any day that is a Business Day, the average (rounded upwards, if necessary, to the next 1/100 of 1%) of the quotations for such day for such transactions received by the Administrative Agent from three Federal funds brokers of recognized standing selected by it; *provided*, that, if the Federal Funds Effective Rate shall be less than zero, such rate shall be deemed to be zero for purposes of this Agreement.

"Fee Letters" means the Lenders' Fee Letter and the Agent Fee Letter.

"Fitch" means Fitch Ratings Limited or its affiliates.

- "Good Faith Contest" means, with respect to the payment of Taxes or any related claims or liabilities by any Person, the satisfaction of each of the following conditions:
- (a) the validity or amount thereof is being diligently contested in good faith by such Person by appropriate proceedings timely instituted;
- (a) in the case of Taxes or related claims and liabilities of the Borrower, the Borrower has established adequate cash reserves with respect to the contested items in accordance with the Colombian Government Entity GAAP or IFRS, as the case may be;
- ${\rm (b)} \qquad \text{during the period of such contest, the enforcement of any contested item is effectively stayed; and} \\$
- (c) such contest or proceedings and any resultant failure to pay or discharge the claimed or assessed amount do not and could not otherwise reasonably expected to result in a Material Adverse Effect.

"Governmental Approval" means any authorization, approval, consent, license, concession, ruling, permit, tariff, rate, certification, order, validation, exemption, waiver, variance, opinion of, or registration, filing or recording with, or report or notice to, any Governmental Authority.

"Governmental Authority" means any national, state, county, city, town, village, municipal or other local governmental department, commission, board, bureau, agency, authority or instrumentality of the United States, Colombia or any other national, multinational or international authority, or any political subdivision of any thereof, and any person exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to any of the foregoing entities, and in each case having jurisdiction over the Persons or matters in question.

"IFRS" means the International Financial Reporting Standards issued by the International Accounting Standards Board, as in effect from time to time, and as inter-

preted and applied by the Colombian National Accounting Office (*Contaduría General de la Nación*), on a basis consistent with the Borrower's operations and reflected in the Borrower's financial statements.

"Indebtedness" of any Person means, without duplication, (a) any indebtedness of such Person for borrowed money or evidenced by a note, debenture or similar instrument (including a purchase money obligation) given in connection with the acquisition of any property or assets, including securities, (b) obligations to pay the deferred purchase price of property or services, except accounts payable and accrued expenses arising in the ordinary course of business and payable within one hundred eighty (180) days, (c) any derivative transaction entered into in connection with, protection against, or benefit from fluctuation in any rate or price (provided that, for the calculation of the value of any derivative transaction, only the net market-tomarket value shall be taken into account), (d) Indebtedness of others described in clauses (a) through (c) above secured by (or for which the holder thereof has an existing right, contingent or otherwise, to be secured by) a Lien on the property of such Person, whether or not the respective Indebtedness so secured has been assumed by such Person, (e) any guarantee by such Person of any Indebtedness of others described in the preceding clauses (a) through (d) above, (f) any Capitalized Lease Obligations of such Person, and (g) any amendment, renewal, extension or refunding of any such Indebtedness.

"Instructions Letter" means the irrevocable instructions letter executed by the Borrower substantially in the form of $\underline{\text{Exhibit B}}$.

"Interest Determination Date" means the second Business Day prior to the commencement of any Interest Period relating to a Loan.

"Interest Payment Date" means the last day of each Interest Period.

"Interest Period" means (i) with respect to the first borrowing of the Loans hereunder, initially, the period commencing (and including) on the date on which such borrowing is made and ending (but excluding) on the date that is six months thereafter, and, thereafter, each period commencing on (and including) the last day of the immediately preceding Interest Period and ending on (but excluding) the date that is six months thereafter; and (ii) with respect to each additional borrowing of the Loans hereunder, the period commencing on (and including) the Disbursement Date corresponding to such borrowing and ending on (but excluding) the last day of the then current Interest Period pursuant to clause (i) above and, thereafter, each period commencing on (and including) the last day of the immediately preceding Interest Period and ending on (but excluding) the date that is six months thereafter; provided, however, that (x) if an Interest Period would end on a day other than a Business Day, such Interest Period shall be extended to the next succeeding Business Day unless such next succeeding Business Day would fall in the next calendar month, in which case such Interest Period shall end on the

next preceding Business Day, and (y) in no case shall any Interest Period end after the Maturity Date.

"Internal Indebtedness" means any Indebtedness payable to Colombian residents.

"Interpolated Rate" means, at any time, for any Interest Period, the rate *per annum* (rounded to the same number of decimal places as the LIBO Screen Rate) determined by the Administrative Agent (which determination shall be conclusive and binding absent manifest error) to be equal to the rate that results from interpolating on a linear basis between: (a) the LIBO Screen Rate (for the longest period for which the LIBO Screen Rate is available) that is shorter than the Impacted Interest Period; and (b) the LIBO Screen Rate is available) that exceeds the Impacted Interest Period, in each case, at such time.

"Ituango Project" means the Ituango hydroelectric plant to be located in the northern region of Antioquia, Colombia.

"Lender Fee Letter" means the fee letter executed on or prior to the Agreement Date between the Borrower and the Lenders.

"LIBOR" means, with respect to any Loans for any Interest Period, the London interbank offered rate as administered by ICE Benchmark Administration (or any other Person that takes over the administration of such rate for U.S. Dollars) for a period equal in length to such Interest Period as displayed on pages LIBOR01 or LIBOR02 of the Reuters screen that displays such rate (or, in the event such rate does not appear on a Reuters page or screen, on any successor or substitute page on such screen that displays such rate, or on the appropriate page of such other information service that publishes such rate from time to time as selected by the Administrative Agent in its reasonable discretion; in each case the "LIBO Screen Rate") on the Interest Determination Date; provided that if the LIBO Screen Rate shall be less than zero, such rate shall be deemed to be zero for the purposes of this Agreement; provided further that if the LIBO Screen Rate shall not be available at such time for such Interest Period (an "Impacted Interest Period") then LIBOR shall be the Interpolated Rate; provided that if any Interpolated Rate shall be less than zero, such rate shall be deemed to be zero for purposes of this Agreement.

"Lien" means any lien, lease, mortgage, pledge, hypothecation, or other encumbrance or security interest.

"Loan Documents" means this Agreement, the Notes, the Instructions Letter, the Fee Letters, and any other document the Borrower may from time to time designate as such with the prior approval of the Ministry of Finance to the extent required under Applicable Law.

"Loans" has the meaning ascribed to such term in <u>Section 2.01</u>. To the extent relating to any particular Lend-

er, references herein to the Loans mean the portion of the Loans allocable to such Lender.

"Margin Stock" means "margin stock" as defined in Regulations U and X of the Board of Governors of the U.S. Federal Reserve System (or any successor thereto), as the same may be modified and supplemented and in effect from time to time.

"Material Adverse Effect" means any event, circumstance, occurrence or condition that, as of any date of determination, results in or otherwise constitutes a material and adverse effect on: (a) the ability of the Borrower to perform any material obligations under the Loan Documents, (b) the validity or enforceability of any Loan Document or any material provision thereof, or (c) the financial condition, business or operations of the Borrower and its Subsidiaries (taken as a whole for purposes of this clause (c)).

"Material Subsidiary" means a Subsidiary of the Borrower which on any given date of determination accounts for more than 10% of the Borrower's Consolidated Total Assets.

"Maturity Date" means the fifth-year anniversary of the Agreement Date, *provided* that if such date is not a Business Day, the Maturity Date shall be the immediately preceding Business Day.

"Ministry of Finance" means the *Ministerio de Hacienda y Crédito Público de Colombia*.

"Moody's" means Moody's Investors Service, Inc.

"Note" means (a) a promissory note issued pursuant to <u>Section 2.06</u>, or (b) any replacement promissory note issued pursuant to this Agreement.

 $\mbox{``OFAC''}$ means the Office of Foreign Assets Control of the U.S. Department of the Treasury.

"Other Connection Taxes" means, with respect to any Lender or the Administrative Agent, Taxes imposed as a result of a present or former connection between such Lender or Administrative Agent and the jurisdiction imposing such Tax (other than connections arising from such Lender or the Administrative Agent having executed, delivered, become a party to, performed its obligations under, received payments under, engaged in any other transaction pursuant to or enforced any Loan Document, or sold or assigned an interest in any Loan Document).

"Other Taxes" means all present or future stamp, court or documentary, intangible, recording, filing or similar Taxes that arise from any payment made under, from the execution, delivery, performance, enforcement or registration of any Loan Document except any such Taxes that are Other Connection Taxes imposed with respect to an assignment (other than an assignment made pursuant to Section 2.12).

"PATRIOT Act" means the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001 (Title III of Pub. L. 107-56 (signed into law October 26, 2001)).

"Permitted Lien" means any of the following:

- (a) Liens arising by operation of law, such as merchants', maritime or other similar Liens arising in the ordinary course of business or Liens in respect of taxes, assessments or other governmental charges that are not yet delinquent or that are being contested in good faith by appropriate proceedings;
- (d) Liens arising in the ordinary course of business in connection with Indebtedness maturing not more than one (1) year after the date on which the Indebtedness was originally incurred and which is related to the financing of export, import or other trade transaction;
- (e) Liens resulting from the deposit of funds or evidences of Indebtedness in trust for the purpose of discharging or defeasing Indebtedness of the Borrower or any Material Subsidiary;
- (f) Liens on assets or property of a Person existing at the time such Person is merged into, consolidated with or acquired by the Borrower or any Material Subsidiary or becomes a Material Subsidiary; provided that any such Lien is not incurred in contemplation of such merger, consolidation or acquisition (unless such Lien was created to secure or provide for the payment of any part of the purchase price of such property or assets) and does not secure any property of the Borrower or any Material Subsidiary other than the property and assets subject to such Lien prior to such merger, consolidation or acquisition;
- $(g) \qquad \text{Liens existing as of the date hereof and set} \\ \text{forth on } \underline{\text{Schedule 1.01-II}};$
- Liens securing Indebtedness (including in (h) the form of Capitalized Lease Obligations and purchase money Indebtedness) incurred for the purpose of financing the cost (including the cost of design, development, site acquisition, construction, integration, manufacture or acquisition) of real or personal property (tangible or intangible) which is incurred contemporaneously therewith or within one hundred eighty (180) days thereafter; provided that (i) such Liens secure Indebtedness in an amount not in excess of the cost of such property (plus an amount equal to the reasonable fees and expenses incurred in connection with the incurrence of such Indebtedness) and (ii) such Liens do not extend to any property of the Borrower other than the property for which such Indebtedness was incurred;
- (i) Liens to secure the performance of statutory and common law obligations, bids, trade contracts, judgments, surety or appeal bonds, performance bonds or other obligations of a like nature incurred in the ordinary course of business;

- (j) Liens to secure debt securities;
- (k) Liens granted in favor of the Borrower and/ or any Wholly Owned Subsidiary to secure Indebtedness owing to the Borrower or such Wholly Owned Subsidiary;
- (l) Legal or equitable encumbrances deemed to exist by reason of the inclusion of customary negative pledge provisions in any financing document of the Borrower or any Subsidiary;
 - (m) Liens securing Internal Indebtedness;
- (n) Liens created in favor of a bank or financial institution which is party to a letter of credit transaction as account party on drafts, bills of lading and other documents which are the subject of such letter of credit transaction;
- (o) Liens on cash or cash equivalents to secure obligations under agreements or arrangements referred to in clause (c) of the definition of "Indebtedness";
- Any Lien in respect of Indebtedness rep-(p) resenting the extension, refinancing, renewal or replacement (or successive extensions, refinancings, renewals or replacements) of Indebtedness secured by Liens referred to in clauses (b), (d), (e), (f), (h), (i), (k), (r), (s) and (t); provided that the principal of the Indebtedness secured thereby does not exceed the principal of the Indebtedness secured thereby immediately prior to such extension, renewal or replacement, plus any accrued and unpaid interest or capitalized interest payable thereon, reasonable fees and expenses incurred in connection therewith, and the amount of any prepayment premium necessary to accomplish any refinancing; provided, further, that such extension, renewal or replacement shall be limited to all or a part of the property (or interest therein) subject to the Lien so extended, renewed or replaced (plus improvements and construction of such property);
- $\rm (q)$ Pledges and deposits made in the ordinary course of business in compliance with workers' compensation, unemployment insurance and other social security laws or regulations;
- (r) Easements, zoning restrictions, rights-ofway and similar encumbrances on real property imposed by law or arising in the ordinary course of business that do not secure any monetary obligations and do not materially detract from the value of the affected property or interfere with the ordinary conduct of business of the Borrower or any of its Subsidiaries;
- (s) Liens arising out of government concessions or licenses held by the Borrower or any of its Subsidiaries;
- (t) Liens to secure the purchase of, or created in connection with the financing of all or any part of the pur-

chase price or cost of the acquisition, purchase, construction, development, extension, expansion and/or improvement by the Borrower or any of its Subsidiaries of, any assets (or right of interest therein), provided that (i) such Liens cover only such assets (or right or interest therein, as the case may be), or any assets forming part of or connected with such assets (or any right or interest therein), or products or proceeds from such assets, or revenue or profit from such assets or such products or proceeds (or any right or interest therein), or the shares or other ownership interests in any Person substantially all of whose assets consist of such assets, revenue or profit, (ii) such Liens secure no more than the purchase price or other consideration paid for, and/or costs of construction, development, expansion, extension and/or improvement, of such assets (or any right or interest therein), including any financing or refinancing costs associated therewith, and (iii) such Liens granted in connection with any extension, expansion and/or improvement of assets cover only assets other than assets existing at the date of this Agreement;

- (u) Liens securing the Borrower's or any of its Subsidiaries' counter-guaranteeing obligations to the Republic of Colombia in respect of any External Indebtedness of the Borrower or any of its Subsidiaries (but only up to an amount equal to 120% of the sum of one principal and one interest payment in respect of such External Indebtedness) provided by multilateral or development organizations;
- (v) Liens in respect of Indebtedness the principal amount of which in the aggregate, together with all other Liens not otherwise qualifying as Permitted Liens pursuant to another part of this definition of Permitted Liens, does not exceed 15% of the Borrower's Consolidated Total Assets. For purposes of this definition, the value of any Lien securing Indebtedness will be computed on the basis of the lesser of (i) the outstanding principal amount of such secured Indebtedness and (ii) the higher of (x) the book value or (y) the Fair Market Value of property securing such Indebtedness; and
- $\label{eq:weights} (w) \qquad \text{Any extension, renewal or replacement of the foregoing.}$

"Person" means any individual, firm, company, limited liability company, corporation, partnership (including association and whether or not having separate legal personality), joint stock company, trust, unincorporated organization or any other enterprise, or a Governmental Authority.

"Potential Default" means an event that with the lapse of time (including any applicable grace period) or the giving of notice, or both, would become an Event of Default.

"Proportionate Share" means, with respect to each Lender and as of any date of determination:

(a) prior to the end of the Availability Period, the then-current ratio of such Lender's unutilized Commitment to the unutilized Total Commitment at that time; and

(x) thereafter, the then-current ratio of the principal amount of all outstanding Loans of such Lender to the principal amount of all outstanding Loans under this Agreement.

"Regulation U" means Regulation U of the Board of Governors of the Federal Reserve System.

"Regulatory Change" means the introduction or change after the date of this Agreement, of any United States or foreign national, state or municipal laws or regulations or in the interpretation or administration thereof, or the adoption or making after such date of any directives or requests (whether or not having the force of law) by any United States or foreign national, state, or municipal court or monetary authority, or other Governmental Authority; provided that notwithstanding anything herein to the contrary, (a) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines or directives thereunder or issued in connection therewith, and (b) all requests, rules, guidelines or directives promulgated by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities, in each case pursuant to Basel III, shall in each case be deemed to be a "Regulatory Change," regardless of the date enacted, adopted or issued.

"Required Lenders" means the Lender or Lenders (a) whose share in the outstanding Loans and whose undrawn Commitments represent at least 50.1% of the aggregate amount of Loans outstanding and undrawn Commitments; or (b) if there is no Loan then outstanding, whose Commitments are at least 50.1% of the Total Commitment.

"Related Parties" means, with respect to any specified Person, such Person's Affiliates and the respective directors, officers, employees, agents and advisors of such Person and such Person's Affiliates.

"**S&P**" means Standard & Poor's Ratings Services, a division of The McGraw-Hill Companies, Inc.

"Sanctioned Country" means a country or territory which is, or whose government is, the subject or target of country-wide or territory-wide Sanctions broadly restricting or prohibiting all dealings with such country or territory.

"Sanctioned Person" means any Person (a) listed in the Specially Designated Nationals and Blocked Person List or Sectoral Sanctions Identifications List maintained by OFAC, (b) listed in any other Sanctions-related list of designated Persons maintained by OFAC, the U.S. Department of State, the United Nations Security Council, the European Union or any EU member state, (c) organized, domiciled or resident in a Sanctioned Country or (d) 50% or more owned by any Person or two or more Persons in the aggregate, or Controlled by any Person, described in clause (a) through (c) above.

"Sanctions" means the sanctions laws, regulations, embargoes or restrictive measures administered, enacted or enforced by any of the Sanctions Authorities.

"Sanctions Authorities" means: (a) the United States of America, (b) the United Nations, (c) the European Union, (d) the United Kingdom, (e) Hong Kong, or (f) the respective Governmental Authorities of any of the foregoing, including, without limitation, OFAC, the US Department of State and Her Majesty's Treasury.

"Subsidiary" means, with respect to any Person, any other Person (a) the securities of which having ordinary voting power to elect a majority of the board of directors (or other persons having similar functions) or (b) the other ownership interests of which ordinarily constituting a majority voting interest, are at the time, directly or indirectly, owned or Controlled by such first Person, or by one or more of its Subsidiaries, or by such first Person and one or more of its Subsidiaries; unless otherwise specified, "Subsidiary" means a Subsidiary of the Borrower.

"Tax" means any present or future income, stamp or other taxes, levies, imposts, duties, charges, fees, assessments, deductions or withholdings (including value-added taxes) or other charges of whatever nature now or hereafter imposed by any jurisdiction or by any political subdivision or taxing authority thereof or therein with respect to payments made by or for the Borrower hereunder or under any Loan Document and all interest, penalties or similar liabilities with respect thereto.

"Taxing Jurisdiction" means, with respect to any Lender, any jurisdiction other than (a) a jurisdiction (other than New York) through which payments to such Lender by the Borrower under any Loan Document shall be made and (b) the jurisdiction under the laws of which such Lender is organized or established.

"Total Commitment" means, on any day, the aggregate Commitments on such day of all the Lenders. The original aggregate principal amount of the Commitments is U.S.\$1,000,000,000.

"U.S." or "United States" means the United States of America.

"Wholly Owned" means, with respect to any corporate entity, any Person of which 100% of the outstanding capital stock (other than qualifying shares, if any) having by its terms ordinary voting power (not dependent on the happening of a contingency) to elect the board of directors (or equivalent controlling governing body) of that Person, is at the time owned or Controlled directly or indirectly by that corporate entity, by one or more wholly owned Subsidiaries of that corporate entity or by that corporate entity and one or more wholly owned Subsidiaries.

- (a) The meanings set forth for defined terms in <u>Section 1.01</u> or elsewhere in this Agreement shall be equally applicable to both the singular and plural forms of the terms defined.
- (b) Unless otherwise specified, all references in this Agreement to Sections, Annexes, Exhibits, and Schedules are to Sections, Annexes, Exhibits, and Schedules in or to this Agreement.
- (c) The headings of the Sections in this Agreement are included for convenience only and shall not in any way affect the meaning or construction of any provision of this Agreement.
- (d) Acknowledging that the parties hereto have participated jointly in the negotiation and drafting of this Agreement, if any ambiguity or question of intent or interpretation arises as to any aspect of this Agreement, then this Agreement will be construed as if drafted jointly by each of parties hereto and no presumption or burden of proof will arise favoring or disfavoring any party by virtue of the authorship of any provision of this Agreement.
- (e) References in this Agreement or the Notes to any belief, decision, discretion or other action held, made, determined, exercised or taken by the Administrative Agent or any Lender shall mean any such belief, decision, discretion or other action to be held, made, determined, exercised or taken in good faith to the extent required under the laws of the State of New York.
- (f) Any reference herein to "including" shall mean "including without limitation."
- (g) References to any document or agreement, including this Agreement, shall be deemed to include references to such documents or agreements as amended, supplemented or replaced from time to time in accordance with its terms and (where applicable) subject to compliance with the requirements set forth therein.
- (h) Unless the context requires otherwise (i) any reference herein to any Person shall be construed to include such Person's successors and assigns, and (ii) any reference to any law or regulation herein shall, unless otherwise specified, refer to such law or regulation as amended, modified or supplemented from time to time or to the successor law or regulation.

SECTION 2. THE LOANS

2.01 Commitments

. Each Lender, severally and not jointly with the other Lenders, agrees to extend to the Borrower, subject to the terms and conditions of this Agreement, one or more term loans in U.S. Dollars (each, a "Loan" and collectively, the "Loans"), the aggregate principal amount of which shall not exceed the Total Commitment.

2.02 <u>Mechanics for Requesting a Loan</u>

. The Borrower shall request disbursement of the Loans in writing by using a request substantially in the form of Exhibit C hereto (the "Loan Request"). Each Loan Request shall be irrevocable and binding on the Borrower and shall be given to the Administrative Agent no later than 10:00 a.m. (New York time) three (3) Business Days before the proposed Disbursement Date (or such shorter period as may be agreed by the Administrative Agent acting upon the direction of all Lenders). The Administrative Agent shall promptly provide to the Lenders a copy of each Loan Request received from the Borrower.

2.03 Funding

(a) Subject to the terms of this Agreement, each Lender shall make available on the applicable Disbursement Date such Lender's Proportionate Share of the Loans requested in the corresponding Loan Request in immediately available funds to the Administrative Agent via Fed Wire or SWIFT, to arrive by 1:00 p.m. New York time to the following account, or such other account as notified from time to time by the Administrative Agent to the Lenders in writing (the "Loan Facility Account"):

Account Name: Loan Operations Depart-

ment

Reference: EMP/ EMPRESAS PÚBLICAS

DE MEDELLÍN E.S.P.,2015

Bank: The Bank of Tokyo-Mitsubishi UFJ,

Ltd.

ABA No.: 026-009-632 Account No: 9777-0191

SWIFT ID: BOTKUS33 Attention: Agency Desk

- (b) The Administrative Agent shall immediately transfer the proceeds funded pursuant to Section 2.03(a) in immediately available funds to the Borrower on the same day via Fed Wire or SWIFT to the account designated by the Borrower in the Loan Request.
- (c) On each day on which a disbursement is made by a Lender, the unutilized and available amount of the Commitment of such Lender shall be reduced by the amount of such disbursement. Notwithstanding any provision of this Agreement to the contrary, no Lender shall be required to make any disbursement hereunder if, as a result thereof, the unutilized and available amount of its Commitment would thereby be exceeded. Amounts borrowed and repaid or prepaid hereunder may not be reborrowed.
- (d) The rights and obligations of each Lender are several and not joint or joint and several. The failure of a Lender to fund its Proportionate Share of any disbursement shall not relieve any other Lender of its obligation under this Agreement to fund its Proportionate Share of any such disbursement on the relevant Disbursement Date. The failure by any Lender to perform its obligations

hereunder shall not affect the obligations of the Borrower towards any other party hereto nor shall any Lender or the Administrative Agent be responsible for the failure of any other Lender to advance its Proportionate Share of any disbursement.

(e) Each Lender at its option may make any Loan by causing any lending office of such Lender located outside Colombia to make such Loan; provided, however, that the exercise of such option shall not affect the obligation of the Borrower to repay such Loan in accordance with the terms of this Agreement and such Loan by such lending office shall not result in any obligation to reimburse or pay costs in connection with Section 2.09 or Section 4.01.

2.04 Minimum Amounts for Each Disbursement

The Borrower shall request that each disbursement under this Agreement (other than the final disbursement) be in a minimum amount of one hundred million U.S. Dollars (US\$100,000,000) or, if less, the remaining balance of the available amount of Total Commitments and in increments of five million U.S. Dollars (US\$5,000,000) in excess thereof. There shall be no more than five (5) disbursements for all of the Loans. A maximum of one (1) disbursement per calendar month shall be made by the Lenders during the Availability Period.

2.05 <u>Termination and Reduction of Commitments</u>

- (a) The Borrower may terminate the Total Commitment, or reduce the amount thereof, without payment of any premium or penalty thereof, by giving written notice to the Administrative Agent not later than 5:00 P.M. on the third Business Day prior to the date of such termination or reduction. Reductions of the Total Commitment shall be apportioned among the Lenders and shall reduce each Lender's Commitment according to such Lender's Proportionate Share and shall be in the amount of U.S.\$5,000,000 or in integral multiples of U.S.\$1,000,000 in excess thereof.
- (b) In the event a Change of Control occurs, the Commitments shall automatically and permanently terminate.

2.06 <u>Notes</u>

(a) The Borrower's obligation to pay the principal of and interest on the Loans to each Lender shall be evidenced by a promissory note with blank spaces and its corresponding letter of instructions (pagaré con espacios en blanco y carta de instrucciones) substantially in the form of Exhibit D (each, a "Note"). Each Note shall be valid and enforceable as to its principal amount at any time only to the extent of the amount disbursed and outstanding under the Loan evidenced thereby; and, as to interest, only to the extent of the interest accrued thereon. Each Note

shall be (i) payable to the order of the applicable Lender, (ii) dated the Agreement Date and (iii) payable at the date referred to in the corresponding Instructions Letter. On the Agreement Date, the Borrower shall provide to the Administrative Agent for distribution to each Lender (i) a duly executed Note for each Lender and (ii) a duly executed Instructions Letter substantially in the form of Exhibit B, pursuant to which the Borrower authorizes the applicable Lender to complete the Note issued in accordance with this Section 2.06.

In case of loss, theft, partial or (b) complete destruction or mutilation of a Note, the applicable Lender shall be entitled to request to the Borrower, and the Borrower shall promptly (but in any event within ten (10) Business Days of such notice) execute and deliver to such Lender in lieu thereof a new Note, dated the same date as the lost, stolen, destroyed or mutilated Note, in replacement of the Note; provided that, in the case of any mutilated Note, such mutilated Note shall be returned to the Borrower and, in the case of any lost, stolen or destroyed Note, the Borrower shall have first received evidence of such loss, theft or destruction as shall reasonably be considered satisfactory by the Borrower. Each Lender shall, prior to delivery of any replacement Note by the Borrower also comply with the procedures established by articles 802 to 821 of the Commercial Code and 398 of Law 1564 of 2012 of Código General del Proceso (once it is ruled effective by the Colombian Council of the Judiciary (Consejo Superior de la Judicatura)) or any other Applicable Law in connection with the case of loss, theft, partial or complete destruction or mutilation of a Note. In the event that any lost or stolen Note is subsequently found, such Lender shall immediately notify the Borrower in writing, cancel such Note and deliver such cancelled Note to the Borrower; provided further that the Borrower shall have already delivered a substitute Note to the Administrative Agent. Such Lender shall reimburse and indemnify the Borrower for and against any and all direct liabilities, obligations, losses, damages, penalties, claims, actions, judgments, suits, costs, expenses or disbursements of whatsoever kind or nature which may be imposed on, asserted against or incurred by the Borrower in any way relating to or arising out of such Lender's Note being lost, stolen or destroyed, excluding any such liabilities, obligations, losses, damages, penalties, claims, actions, judgments, suits, costs, expenses or disbursements caused by the Borrower. All replacement Notes issued in connection with this Agreement shall be signed by an Authorized Officer of the Borrower.

- (c) The payment of any part of the principal of any such Note shall discharge the obligation of the Borrower under this Agreement to pay principal of the Loan evidenced by such Note *pro tanto*, and the payment of any principal of a Loan in accordance with the terms hereof shall discharge the obligations of the Borrower under the Note evidencing such Loan *pro tanto*.

- (e) The Notes shall only be sold, assigned or transferred in accordance with the provisions of this Agreement and Applicable Law.
- (f) Each Lender agrees and covenants that it will not declare any amounts outstanding under any Note (or any interest thereon) to be due and payable prior to its stated maturity thereof except in accordance with the provisions of this Agreement.
- In the case of a permitted assignment pursuant to Section 11.03, (i) if requested by the assignee and if such assignment is of the entire Loan(s) held by the Lender, the Lender shall deliver to the Administrative Agent for further delivery to the Borrower concurrently with the execution and delivery by the Borrower to the Administrative Agent of the new Notes in the manner contemplated in clause (ii) below, the Note held by such assigning Lender evidencing such Loan (for any assigning Lender, together with the related Instructions Letter, the "Existing Notes" of such assigning Lender) and (ii) if requested by the assigning Lender or the relevant assignee, the Administrative Agent shall as promptly as reasonably practicable request that the Borrower, and the Borrower hereby agrees to, execute and deliver to the Administrative Agent as promptly as reasonably practicable (for further delivery to such assigning Lender (if applicable) and such assignee) a new Note or Notes (together with the related Instructions Letter) evidencing the Loan held by such assigning Lender (if applicable) and such assignee (in exchange for Existing Notes to the extent such assignment is of the entire Loan(s) held by the assigning Lender).

2.07 Interest

(a) <u>Interest</u>.

(i) The Borrower agrees to pay interest in respect of the unpaid principal amount of each Loan from the applicable Disbursement Date until the maturity thereof (whether by acceleration or otherwise), at a rate *per annum* equal to LIBOR for the Interest Period plus the Applicable Margin.

Notwithstanding the foregoing, if any principal of or interest on any Loan or any fee or other amount payable by the Borrower hereunder is not paid when due, whether at stated maturity, upon acceleration or otherwise, such overdue amount shall bear interest, after as well as before judgment, at a rate per annum equal to the rate which is 2.0% in excess of the rate then applicable to the Loan pursuant to Section 2.07(a)(i). The parties hereto acknowledge that, as of the Agreement Date, the payment of interest on interest (as described in this Section 2.07(a)(ii)), including default interest, is prohibited under Colombian law as currently in effect, subject to applicability of the rules set forth in Article 886 of the Commercial Code, it being understood that any outstanding interest obligations that remain unpaid for more than one (1) year may (upon the satisfaction of the rules set forth in Article 886 of the Commercial Code) be subject to interest on interest under Colombian law as currently in effect.

- (iii) Interest shall be payable on each Interest Payment Date, and upon the payment or prepayment thereof (but only on the amount paid or prepaid).
- (iv) All computations of interest hereunder shall be made on the actual number of days elapsed over a year of 360 days.
- (v) On each Interest Determination Date, the Administrative Agent shall determine LIBOR for the relevant Interest Period and shall promptly notify the Borrower and the Lenders thereof. Each such determination shall, absent manifest error, be final and conclusive and binding on all parties hereto.
- $(vi) \qquad \text{In no event shall the interest applicable under this Agreement exceed the interest rate limits set forth by Applicable Law.}$

(b) Alternate Rate of Interest.

- (i) If the Administrative Agent shall have determined (which determination shall be conclusive and binding for all purposes, absent manifest error), or the Required Lenders have advised the Administrative Agent, at any time that:
- (ii) on any Interest Determination Date, U.S. Dollar deposits of sufficient amount and maturity for funding a disbursement are not available to the Required Lenders in the London interbank market in the ordinary course of business; or
- (iii) on any Interest Determination Date, by reason of circumstances affecting the relevant market, adequate and fair means do not exist for ascertaining the rate of interest to be applicable to a Loan; or
- (iv) on any Interest Determination Date, the relevant rate of interest referred to in the definition of LIBOR that is to be used to determine the rate of interest for a Loan does not cover the funding cost to the Required Lenders of making or maintaining the Loan; or
- $$\left(v\right)$$ it has become unlawful at any time for any Lender to make or maintain Loans based upon LIBOR;

then the Administrative Agent shall notify the Borrower and the Lenders thereof.

Upon receipt of such notice, (A) the Borrower's right to request the making by any affected Lender of, and the affected Lender's obligations to make available or continue to maintain, its Proportionate Share of the affected Loan based upon LIBOR shall be suspended, and (B) each such Lender's Proportionate Share of the affected Loan shall instead bear interest at the rate per annum equal to the sum of (x) the Alternative Base Rate, plus (y) the Ap-

plicable Margin, it being expressly understood and agreed that the same Alternative Base Rate and Applicable Margin shall apply for all Lenders so affected effective as of the date of effectiveness of such event.

2.08 <u>Fees</u>

- (a) The Borrower agrees to pay to the Administrative Agent for the account of each Lender on each Interest Payment Date and on the last day of the Availability Period (or, if such day is not a Business Day, on the immediately preceding Business Day), a commitment fee (the "Commitment Fee"), at a rate *per annum* equal to 30% of the Applicable Margin on the daily average unused and uncancelled portion of the Commitment of such Lender. The Commitment Fee due to each Lender shall commence to accrue on the Closing Date, shall be payable in arrears and shall cease to accrue on the last day of the Availability Period.
- (b) The Borrower agrees to pay to the Administrative Agent for the account of the Administrative Agent, or the Lenders, as the case may be, in Dollars, any and all fees payable in the amounts and at the times set forth in the Fee Letters.
- (c) All fees payable hereunder, under the Fee Letters and under the other Loan Documents shall be paid on the dates due, in immediately available funds, to the Administrative Agent for distribution, if applicable, to the Lenders. Fees paid in accordance with this Agreement, the Fee Letters and the other Loan Documents shall not be refundable under any circumstances.

2.09 Additional or Increased Costs

If, due to any Regulatory Change that: (i) changes the basis of taxation of any amounts payable to any Lender (other than (A) Covered Taxes, (B) Taxes described in clauses (b) through (d) of the definition of Excluded Taxes, and (C) Connection Income Taxes); (ii) imposes, modifies or holds applicable any reserve, special deposit, deposit insurance or similar requirement, including any compulsory loan requirement, insurance charge or other assessment (other than, for any period for which the affected Lender is subject to a Capital Adequacy Requirement, the reserves against "Eurocurrency liabilities" under Regulation D of the Board of Governors of the United States Federal Reserve System) against assets of, deposits with or for the account of, or Loan extended by, such Lender; or (iii) imposes any other condition affecting this Agreement or the Note held by any Lender, and the effect of any of the foregoing is to increase the cost to such Lender of making or maintaining any of its Loan or to reduce any amount received or receivable by such Lender hereunder (whether of principal, interest or otherwise), then the Borrower shall from time to time, upon written demand by such Lender, pay to the Administrative Agent for the benefit of such Lender, additional amounts sufficient to compensate such Lender for such increased cost or reduction suffered.

- (b) Each demand for payment by a Lender under this Section 2.09, shall be accompanied by a certificate showing in reasonable detail the basis for the calculation of the amounts demanded in good faith, which certificate, in the absence of manifest error, shall be conclusive and binding for all purposes. The Borrower shall pay such Lender, as the case may be, the amount shown as due on any such certificate within twenty (20) Business Days after receipt thereof.
- (c) No Lender shall be entitled to demand or be compensated for any additional amounts under this <u>Section 2.09</u> (i) to the extent that such additional amounts relate to any period of time more than one hundred eighty (180) days prior to the date upon which such Lender first notifies the Borrower of such additional amounts, or (ii) if such Lender is causing the incremental cost to be incurred for a reason not provided for in <u>Section 2.09(a)</u> above.
- (d) If the Borrower is required to pay any amount to a Lender pursuant to this <u>Section 2.09</u>, it may prepay the portion of the Loan held by such Lender in accordance with <u>Section 3.02</u>. Notwithstanding anything to the contrary herein, the provisions of <u>Sections 3.02(a)</u>, <u>3.04(b)</u> and <u>3.04(c)</u> shall not apply to any such prepayment.

2.10 <u>Breakage Costs, Other Expenses and Losses</u>

The Borrower agrees to compensate each Lender, promptly upon its written request (which request shall set forth in reasonable detail the basis for requesting such compensation), for all reasonable and documented losses, expenses and liabilities which such Lender may sustain (including any loss, expense or liability incurred by reason of the liquidation or reemployment of deposits or other funds required by such Lender to fund its portion of the Loan, but excluding any loss of anticipated profits) if: (i) the Borrower fails to borrow in accordance with a Loan Request, (ii) the Borrower fails to make a voluntary prepayment of the Loan on an Interest Payment Date therefor in accordance with a prepayment notice given pursuant to Section 3.02, or (iii) the Borrower otherwise prepays the Loan on any date other than an Interest Payment Date therefor. Each Lender's calculation of the amount of compensation owing pursuant to this Section 2.10 shall be made in good faith and in a commercially reasonable manner. A Lender's basis for requesting compensation pursuant to this Section 2.10 and a Lender's calculation of the amount thereof made in accordance with the requirements of this Section 2.10 shall, absent manifest error, be final and conclusive and binding on all parties hereto. The Borrower shall pay such Lender, as the case may be, the amount shown as due on any such certificate within twenty (20) Business Days after receipt thereof.

2.11 <u>Illegality</u>

Notwithstanding any other provision herein, if after the Agreement Date the adoption of or any change in any Applicable Law or in the interpretation or application thereof

by a competent Governmental Authority shall make it unlawful for any Lender to make or maintain any of its Loans as contemplated by this Agreement and the Notes, such Lender shall give notice thereof to the Administrative Agent and the Borrower describing in reasonable detail the relevant provisions of such Applicable Law, following which (a) the Commitment of such Lender shall forthwith be suspended until such Lender notifies the Administrative Agent and the Borrower that the circumstances causing such suspension no longer exist and (b) if such Applicable Law shall so mandate, such Lender's Loans then outstanding shall be prepaid by the Borrower on or before the date required and permitted by Applicable Law, together with all accrued interest thereon (unless actions taken pursuant to Section 2.12 shall make such prepayment unnecessary).

2.12 Replacement of Lenders

(a) On and after the initial Disbursement Date, (i) upon the occurrence of any event giving rise to the operation of Section 2.09 with respect to any Lender that results in such Lender charging to the Borrower additional or increased costs, (ii) upon any adoption or change of the type described in Section 2.11, or (iii) the Borrower being required to compensate any Lender pursuant to Section 4.01, to the extent such compensation results solely from a Regulatory Change after the initial Disbursement Date, then such Lender shall use reasonable efforts to designate a different lending office for funding or booking its Loans hereunder or to assign its rights and obligations hereunder to another of its offices, branches or affiliates, if, in the reasonable judgment of such Lender, such designation or assignment (x) would eliminate or reduce amounts payable pursuant to Sections 2.09 or 4.01, as the case may be, in the future, or would eliminate or reduce the effect of any adoption or change described in Section 2.11 or (y) would not subject such Lender to any unreimbursed cost or expense and would not otherwise be disadvantageous to such Lender. The Borrower hereby agrees to pay all reasonable and documented out-of-pocket costs and expenses incurred by any Lender in connection with any such designation or assignment.

On and after the initial Disbursement Date, (i) upon the occurrence of any event giving rise to the operation of Section 2.09 with respect to any Lender that results in such Lender charging to the Borrower additional or increased costs, (ii) upon any adoption or change of the type described in Section 2.11, (iii) in the case of a refusal by a Lender to consent to a proposed change, waiver, discharge or termination with respect to this Agreement which has been approved by the Required Lenders as provided in Section 11.06(b), or (iv) the Borrower being required to compensate any Lender pursuant to Section 4, to the extent such compensation results solely from a Regulatory Change after the initial Disbursement Date, the Borrower shall have the right at its sole expense and effort, if (A) no Event of Default then exists or would exist after giving effect to such replacement, (B) in the case of any such assignment resulting from a claim for compensation under Section 2.09 or payments required to be made pursuant to Section 4, such assignment will result in a reduction in such compensation or payments and (C) in the case of any such assignment resulting from any adoption or change of the type described in Section 2.11, such assignment would eliminate or reduce the effect of any adoption or change described in Section 2.11, to replace such Lender (the "Replaced Lender") with one or more assignee Lenders (collectively, the "Replacement Lenders"); provided that, at the time of any replacement pursuant to this <u>Section 2.12</u>, each Replacement Lender shall enter into an Assignment and Assumption Agreement pursuant to Section 11.03(b) (ii), pursuant to which the Replacement Lender shall acquire the applicable portion of the outstanding Loan due to the Replaced Lender, and shall pay to the Replaced Lender in respect thereof an amount equal to the principal of, and all accrued interest on, the acquired portion of the Loan of the Replaced Lender plus all other amounts payable to Replaced Lender hereunder. A Lender shall not be required to make any such assignment and delegation if, prior thereto, as a result of a waiver by such Lender or otherwise, the circumstances entitling the Borrower to require such assignment and delegation cease to apply.

(c) Upon the execution of the respective Assignment and Assumption Agreement, the payment of any applicable amount referred to in the proviso to Section 2.12(b), recordation of the assignment on the Register by the Administrative Agent pursuant to Section 11.03(e) and delivery to the Replacement Lender(s) of the appropriate Note(s) executed by the Borrower, each Replacement Lender shall become a Lender hereunder and the Replaced Lender shall cease to constitute a Lender hereunder, except with respect to indemnification provisions under Sections 10.06 and 11.08 of this Agreement, which shall survive as to such Replaced Lender.

SECTION 3. PAYMENTS

3.01 Repayment; Time, Place and Manner

The entire outstanding principal amount of the Loan shall mature and become due and payable, and shall be unconditionally repaid by the Borrower to the Administrative Agent for the benefit of the Lenders, on the Maturity Date.

3.02 Voluntary Prepayment

The Borrower may from time to time, without premium or penalty, prepay all or any part of the Loans on any Interest Payment Date; *provided, however*, that: (a) any partial prepayment shall be in a minimum principal amount of U.S.\$10,000,000 or a multiple of U.S.\$5,000,000 in excess thereof; (b) the Borrower shall have given the Administrative Agent at least five (5) Business Days' prior written notice of the proposed prepayment date and the amount of principal to be prepaid (which notice shall be irrevocable) and (c) the Borrower shall have paid in full all amounts then due under this Agreement as of such prepayment date, including interest which has accrued to the prepayment date on the amount being prepaid and any amounts under <u>Section 2.10</u>.

In the event a Change of Control occurs, the Borrower shall prepay in full an amount equal to 100% of all of the then outstanding principal amount of the Loans within thirty (30) days from the occurrence of such Change of Control, together with accrued interest to the date of such prepayment. The Borrower shall notify the Administrative Agent of any prepayment pursuant to this Section 3.03 at least five days before the date such Change of Control is consummated. Promptly following the receipt of any such notice, the Administrative Agent shall advise the Lenders of the contents thereof.

3.04 Payments; Pro Rata Treatment

- (a) The Borrower shall make each payment hereunder or under any Note (including principal and interest) without set-off or counterclaim and not later than 11:00 a.m. (New York time) on the day when due, in Dollars to the Administrative Agent to the account designated by the Administrative Agent to the Borrower in writing prior to the date hereof (or to any other account outside of Colombia specified by the Administrative Agent in writing at least five (5) Business Days prior to the applicable payment date), in immediately available funds. Payments received by the Administrative Agent after 1:00 p.m. (New York time) on any Business Day shall be deemed to be received on the next Business Day.
- The Administrative Agent agrees that, promptly after its receipt of any payment from or on behalf of the Borrower pursuant to Section 3.04(a) above it shall (except as otherwise expressly provided in this Agreement) distribute such payment to the Lenders pro rata based upon their respective share, if any, of the obligations with respect to which such payment was received, provided that this clause (b) shall not apply to payments or prepayments to any Lender in connection with a claim under Sections 2.09, 2.10, 2.11 or Section 4. If at any time insufficient funds are received by and available to the Administrative Agent to pay fully all amounts of principal, interest and fees then due hereunder, such funds shall be applied (i) first, towards payment of interest and fees then due hereunder. ratably among the parties entitled thereto in accordance with the amounts of interest and fees then due to such parties, and (ii) second, towards payment of principal then due hereunder, ratably among the parties entitled thereto in accordance with the amounts of principal and then due to such parties.
- (c) If any of the Lenders shall, whether by voluntary payment, by realization upon security, through the exercise of any right of set-off or banker's lien, by counterclaim or cross action or by the enforcement of any right under this Agreement or any Note, receive payment of the aggregate amount of principal or interest then due and owing to the Lenders which is greater than the proportion due to such Lender, then such Lender receiving such proportionately greater payment shall (i) notify the Administrative Agent and each other Lender of the receipt of such payment and (ii) purchase (for cash at face value) participations in the Loans of other Lenders in accordance with Section 11.03(c) to the extent necessary so that the benefit of all such payments shall be shared by the Lenders ratably

in accordance with the aggregate amount of principal of and accrued interest on their respective Loans; provided that (i) if any such participations are purchased and all or any portion of the payment giving rise thereto is recovered. such participations shall be rescinded and the purchase price restored to the extent of such recovery, without interest, and (ii) the provisions of this paragraph shall not be construed to apply to any payment made by the Borrower pursuant to and in accordance with the express terms of this Agreement or any payment obtained by a Lender as consideration for the assignment of or sale of a participation in any of its Loans to any assignee or participant, other than to the Borrower or any Subsidiary or Affiliate thereof (as to which the provisions of this paragraph shall apply). The Borrower consents to the foregoing. This clause (c) shall not apply to payments or prepayments to any Lender in connection with a claim under Sections 2.09, 2.10, 2.11

3.05 Extension of Payment Dates

Unless otherwise provided herein, whenever any payment to the Administrative Agent or the Lenders under this Agreement or any Note shall be due (otherwise than by reason of acceleration) on a day that is not a Business Day, the date of payment thereof shall be extended to the next succeeding Business Day; *provided*, *however*, that with respect to the payment of principal of, and interest on, the Loans on the Maturity Date, if such date is not a Business Day, the date of payment thereof shall be made on the next preceding Business Day.

SECTION 4. TAXES

4.01 Covered Taxes

- Any and all payments by or on account of any obligation of the Borrower under any Loan Document shall be made without deduction or withholding for any Taxes, except as required by Applicable Law. If the Borrower shall be required by Applicable Law to withhold or deduct any Taxes from or in respect of any such sum payable to or for the benefit of the Administrative Agent or any Lender, then the Borrower shall be entitled to make such deduction or withholding and shall timely pay the full amount deducted or withheld to the relevant taxing authority or other authority in accordance with Applicable Law and, if such Tax is a Covered Tax imposed by a Taxing Jurisdiction, then the sum payable by the Borrower shall be increased as necessary so that after making all required deductions (including deductions applicable to additional sums payable under this <u>Section 4.01</u>) such Lender or the Administrative Agent receives an amount equal to the sum it would have received had no such deductions been made (subject, in any case, to Section 11.03 in case of an assignment of the Loans).
- (b) The Borrower shall indemnify each Lender or the Administrative Agent for the full amount of Covered Taxes imposed by a Taxing Jurisdiction that are paid by such Lender or the Administrative Agent (including any Covered Taxes imposed by a Taxing Jurisdiction

on amounts payable under this <u>Section 4.01</u> arising therefrom or with respect thereto), whether or not such Covered Taxes were correctly or legally asserted. Each Lender or the Administrative Agent shall give notice to the Borrower of the assertion of any claim against such Lender or the Administrative Agent relating to its Covered Taxes as promptly as possible (and in any event within thirty (30) days) after receipt of formal written notice of such assertion, *provided* that failure by a Lender or the Administrative Agent to provide any such notice within ninety (90) days shall relieve Borrower of its obligation to indemnify the Lender or the Administrative Agent pursuant to this <u>Section 4.01</u>. Within ten (10) Business Days of receipt of any such notice from a Lender or the Administrative Agent, the Borrower shall either:

(i) advise such Lender or the Administrative Agent that it intends to indemnify such Lender or the Administrative Agent in respect of such Covered Taxes pursuant to this paragraph (b), in which case it shall promptly indemnify in respect of such amounts, or

(ii) advise such Lender or the Administrative Agent that it intends to commence a Good Faith Contest with respect to such Covered Taxes at the Borrower's sole cost and expense, in which case it shall promptly commence such Good Faith Contest.

Except to the extent of any Good (c) Faith Contest with respect to Covered Taxes, an indemnity made by the Borrower pursuant to this indemnification shall be made within Thirty (30) Business Days after the date the relevant Lender makes written demand therefor, which demand shall be accompanied by a certificate describing in reasonable detail the basis thereof. If the Borrower shall have commenced a Good Faith Contest with respect to any such Covered Taxes and no indemnity payment has been made to the Lender, and such Covered Taxes are ultimately determined to be payable by the relevant Lender in a final judicial proceeding or otherwise, the Borrower shall indemnify such Lender or the Administrative Agent for such Covered Taxes and for any other liability including penalties and interest charged by the relevant taxing authority arising therefrom or with respect thereto.

(d) Within ten (10) days after the date of any indemnification of Covered Taxes by the Borrower, the Borrower shall furnish to the Lender or the Administrative Agent the original or a certified copy of a receipt evidencing indemnification thereof or, if later, promptly after the date on which it receives such receipt and the Borrower shall promptly furnish to the Lender or the Administrative Agent any other information, documents and receipts that the Lender or the Administrative Agent may from time to time reasonably request to establish to its satisfaction that full and timely indemnification has been made of all Covered Taxes required to be indemnified under this Section 4.01.

(e) If the Administrative Agent or any Lender determines in good faith that it has finally and irrevocably received or been granted a refund in respect of any Covered Taxes as to which indemnification has been made

by the Borrower pursuant to this <u>Section 4.01</u>, it shall within ten (10) days after the date the Administrative Agent or any Lender has received or been granted a refund, remit such refund (including any interest received in respect thereof). net of all reasonable Taxes and out-of-pocket costs and expenses payable as a result thereof, to the Borrower; provided, that the Borrower agrees to promptly return any such refund to the Administrative Agent or the relevant Lender in the event the Administrative Agent or such Lender is required to repay such refund to the relevant taxing authority. The Administrative Agent and each Lender shall provide the Borrower with a copy of any notice of assessment (or any similar documentation) from the relevant taxing authority (redacting any unrelated Confidential Information contained therein) requiring repayment of such refund. This paragraph shall not be construed to require the Administrative Agent or any Lender to make available its Tax returns (or any other information relating to its Taxes that it deems confidential) to the Borrower or any other Person.

If the Administrative Agent or any Lender is entitled to an exemption from or reduction of withholding Tax with respect to payments made under any Loan Document, it shall deliver to the Borrower and the Administrative Agent, at the time or times reasonably requested by the Borrower or the Administrative Agent, such properly completed and executed documentation reasonably requested by the Borrower or the Administrative Agent as will permit such payments to be made without withholding or at a reduced rate of withholding. In addition, the Administrative Agent or any Lender, if reasonably requested by the Borrower or the Administrative Agent, shall deliver such other documentation prescribed by applicable law or reasonably requested by the Borrower or the Administrative Agent as will enable the Borrower or the Administrative Agent to determine whether or not such Lender is subject to backup withholding or information reporting requirements. The Administrative Agent and each Lender agree that if any form or certification they previously delivered expires or becomes obsolete or inaccurate in any respect, they shall update such form or certification or promptly notify the Borrower and the Administrative Agent in writing of their legal inability to do so. Notwithstanding anything to the contrary in the preceding three sentences, the completion, execution and submission of such documentation shall not be required if in the Lender's reasonable judgment such completion, execution or submission would subject such Lender to any material unreimbursed cost or expense or would materially prejudice the legal or commercial position of such Lender.

SECTION 5. CONDITIONS PRECEDENT

5.01 Conditions Precedent to Effectiveness

This Agreement and the obligations of the Lenders to make their first Loan hereunder shall become effective on the date on which the Administrative Agent shall have received each of the following documents or each of the following conditions shall have been satisfied, as applicable, which date shall fall on or prior to January 12, 2016 (such date, the "Closing Date"), each of which shall be satisfactory to the Administrative Agent (and to the extent specified

below, to each Lender) in form and substance (or such condition shall have been waived):

- (a) <u>This Agreement</u>. This Agreement shall have been duly authorized, fully executed and delivered by the parties hereto, shall be in full force and effect and originals shall have been delivered to the Administrative Agent (with evidence that this Agreement has been registered with the appropriate authorities in Colombia, if such registration is required under Applicable Law).
- (b) Existence and Authority. The Administrative Agent shall have received a certificate signed by an Authorized Officer of the Borrower, dated as of the Closing Date, substantially in the form of Exhibit E, with respect to (i) the authority of the Borrower to execute, deliver, perform and observe the terms and conditions of the Loan Documents; (ii) the authority (including specimen signatures) for each Person who, on behalf of the Borrower, signed any Loan Document; and (iii) the Borrower's valid existence under the laws of Colombia.
- (c) <u>Governmental Approvals.</u> The Administrative Agent shall have received copies, certified as true copies by an Authorized Officer of the Borrower, of each consent, license, authorization or approval of, and exemption by, any Governmental Authority (including a resolution issued by the Ministry of Finance authorizing the execution of the Agreement), all of which are set forth on <u>Schedule 6.01(b)</u>: (i) for the execution, delivery, performance, and observance by the Borrower of the Loan Documents, including all approvals relating to the availability and transfer of U.S. Dollars required to make all payments due under the Loan Documents; and (ii) for the validity, binding effect, and enforceability of the Loan Documents.
- Legal Opinions. The Administrative Agent shall have received opinions dated as of the Closing Date of (i) Mosquera Abogados, Colombian counsel to the Borrower, substantially in the form of Exhibit F. (ii) Milbank, Tweed, Hadley & McCloy, New York counsel to the Borrower, substantially in the form of Exhibit G, (iii) Skadden, Arps, Slate, Meagher & Flom, New York counsel to the Administrative Agent and the Lenders, covering such matters relating to the transactions contemplated hereby as the Administrative Agent may reasonably request, and (iv) Brigard & Urrutia Abogados, Colombian counsel to the Administrative Agent and the Lenders, covering such matters relating to the transactions contemplated hereby as the Administrative Agent may reasonably request, and, in each case, addressed to the Lenders and the Administrative Agent.
- (e) <u>Appointment of Process Agent.</u> The Administrative Agent shall have received delivery of evidence that (i) the Borrower has irrevocably appointed as its agent for service of process the Person or Persons so specified in <u>Section 9.03</u>, and (ii) the designated agent has accepted the appointment (and been paid in full by the Borrower) for a term extending at least one year beyond the Maturity Date and has agreed to forward forthwith to the Borrower all legal process addressed to the Borrower that is received by such agent.

- (f) Fees. The Borrower shall have paid to the Administrative Agent and each Lender, as applicable, or shall have authorized deductions thereof from the proceeds of the first disbursement, the fees due under the Agent Fee Letter or under Section 11.08(a) (including fees and expenses of counsel).
- (g) <u>Borrower Financial Statements</u>. The Administrative Agent shall have received copies of the Borrower Financial Statements.
- (h) <u>Note</u>. The Administrative Agent shall have received a duly executed Note and an Instructions Letter with respect thereto for each Lender issued in accordance with Section 2.06.
- (i) <u>KYC</u>. To the extent requested at least three (3) Business Days prior to the Closing Date by the Administrative Agent or any Lender, the Administrative Agent or such Lender, as the case may be, shall have received such requested documents required to comply with their respective "know your customer" procedures.
- (j) <u>Closing Certificate</u>. The Administrative Agent shall have received a certificate signed by an Authorized Officer of the Borrower, dated as of the Closing Date, certifying (i) that the conditions specified in <u>Sections 5.02(b)</u> and (d) have been satisfied as of the Closing Date, and (ii) that there has been no event or circumstance since December 31, 2014 that has had or could be reasonably expected to have, either individually or in the aggregate, a Material Adverse Effect.

The obligation of each Lender to make any disbursement (including the first disbursement) shall be subject to the satisfaction (or waiver by the Administrative Agent acting on the instruction each of the Lenders), prior to the making of such disbursement, of each of the following conditions precedent:

- (a) <u>Closing Date</u>. The Closing Date shall have occurred.
- (b) <u>No Potential Default or Event of Default.</u> No Potential Default or Event of Default shall have occurred and be continuing both before and immediately after giving effect to the Loans and the application of the proceeds thereof.
- (c) <u>Loan Request</u>. The Administrative Agent shall have received a duly executed and completed Loan Request.
- (d) <u>Representations and Warranties</u>. The representations and warranties made by the Borrower in this Agreement shall be true and correct in all material respects on and as of the Disbursement Date both before and immediately after giving effect to the Loans and the ap-

plication of the proceeds thereof; *provided* that representations and warranties qualified as to materiality shall be true and correct in all respects as of such date.

SECTION 6. REPRESENTATIONS AND WARRANTIES

6.01 <u>Representations and Warranties of the Borrower</u>

The Borrower represents and warrants, as of the Agreement Date, the Closing Date, and as of each Disbursement Date, that:

- (a) <u>Existence and Authority</u>. The Borrower is duly organized and validly existing under the laws of Colombia, has all requisite power, authority and legal right to own its property and carry on its business as now conducted, and has taken all actions necessary to authorize it to execute, deliver, perform, and observe the terms and conditions of the Loan Documents.
- (b) <u>Governmental Approvals</u>. All consents, licenses, permits, authorizations and approvals of, and exemptions by, any Governmental Authority that are necessary: (i) for the execution, delivery, performance and observance by the Borrower of the Loan Documents, including approvals relating to the availability and transfer of U.S. Dollars required to make all payments due under the Loan Documents; and (ii) for the validity, binding effect and enforceability of the Loan Documents, have, in each case, been obtained and are in full force and effect, as set forth on <u>Schedule 6.01(b)</u>.
- (c) <u>Recordation</u>. To ensure the legality, validity, enforceability, priority or admissibility in evidence in Colombia of any of the Loan Documents, it is not necessary that any of the Loan Documents be registered, recorded, enrolled or otherwise filed with any court or Governmental Authority, except as set forth in <u>Schedule 6.01(b)</u>, or be notarized; or that any documentary, stamp or other similar tax, imposition or charge of any kind be paid on or in respect of any of the Loan Documents, except as set forth in <u>Schedule 6.01(b)</u>.
- Restrictions. The execution, deliv-(d) ery and performance or observance by the Borrower of the terms of, and consummation by the Borrower of the transactions contemplated by, this Agreement do not and will not conflict with or result in a breach or violation of: (i) the Estatutos Sociales of the Borrower; (ii) any law of Colombia or any other ordinance, decree, constitutional provision, regulation or other requirement of any Governmental Authority in effect as of the date on which this representation is made; or (iii) any order, writ, injunction, judgment, decree or award of any court or other tribunal. Further, the Borrower's execution and delivery of the Loan Documents, the performance and observance of its obligations thereunder, and the consummation of the transactions contemplated by this Agreement do not and will not conflict in any material respect with or result in a material breach of any material agreement or instrument to which the Borrower is a party

or to which it or any of its revenues, properties or assets may be subject, or result in the creation or imposition of any Lien upon any of the revenues, properties or assets of the Borrower pursuant to any such material agreement or instrument.

- (e) <u>Binding Effect</u>. The Borrower has duly executed and delivered each Loan Document on or before the Agreement Date. Each Loan Document constitutes a direct, general, and unconditional obligation of the Borrower that its legal, valid, and binding upon the Borrower and enforceable against the Borrower in accordance with its respective terms, except as such enforceability may be limited by Colombian public order laws, applicable insolvency, reorganization, liquidation, moratorium, readjustment of debt or other similar laws affecting the enforcement of creditors' rights generally, and by the application of general principles of equity, regardless of whether such enforceability is considered in a proceeding at law or in equity.
- (f) Choice of Law. Under the conflict of laws principles in Colombia, the choice of law provisions of this Agreement and any Note are valid, binding and not subject to revocation by the Borrower, and in any proceedings brought in Colombia for enforcement of this Agreement and any Note, the choice of the law of the State of New York as the governing law of such documents will be recognized and such law will be applied. Notwithstanding the foregoing or anything to the contrary in Section 9.01, all matters governing the authorization and execution of the Loan Documents by the Borrower are governed by and construed in accordance with the laws of Colombia.
- Commercial Activity. Except as (i) (g) provided for in Articles 192, 195, 298 and 299 of Law 1437 of 2011 (Código de Procedimiento Administrativo y de lo Contencioso Administrativo) and (ii) Articles 684 and 513 of the Colombian Civil Procedure Code (Código de Procedimiento Civil) (which will be gradually superseded by Articles 593, 594 and 595 et al subject to the entry into force of Law 1564 of 2012 (Código General del Proceso) pursuant to the terms of article 627, paragraph 6 thereof), the Borrower acknowledges that the execution and performance of this Agreement and each other Loan Document is a commercial activity and to the extent that the Borrower has or hereafter may acquire any immunity from any legal action, suit or proceedings, from jurisdiction of any court or from set-off or any legal process (whether service or notice, attachment prior to judgment, attachment in aid of execution of judgment, execution of judgment or otherwise) with respect to itself or any of its property or assets, whether or not held on its own account, the Borrower hereby irrevocably and unconditionally waives and agrees not to plead or claim such immunity in respect of its obligations under this Agreement or any other Loan Document.
- (h) <u>Legal Proceedings</u>. There are no actions, suits, litigation, arbitration or administrative proceedings pending or, to the best of the Borrower's knowledge and belief, threatened against the Borrower which are reasonably likely to be adversely determined and, if adversely determined, could reasonably be expected to have a Material Adverse Effect.

(i) Borrower Financial Statements.

The Borrower Financial Statements present fairly in all material respects the financial condition of the Borrower and its consolidated subsidiaries at the date of such statements and the results of the operations of the Borrower and its consolidated subsidiaries for the fiscal year or other time period to which such statements refer, in the case of unaudited Financial Statements, subject to changes resulting from audit and nominal year-end adjustments and the absence of footnotes. The Borrower Financial Statements were prepared in accordance with Colombian Government Entity GAAP or, for any financial statements in respect of periods commencing on or after January 1, 2015, IFRS consistently applied. Except as reflected in the Borrower Financial Statements, there are no liabilities or obligations with respect to the Borrower or any of its consolidated subsidiaries of any nature whatsoever (whether absolute, accrued, contingent or otherwise, and whether or not due) for the period to which the Borrower Financial Statements relate that, either individually or in the aggregate, would be materially adverse to the Borrower.

- (ii) Since the date of the latest audited financial statements of the Borrower delivered to the Lenders hereunder, there has been no event, condition or circumstance that has had or could reasonably be expected to have a Material Adverse Effect.
- (j) <u>Anti-Money Laundering Laws</u>. None of the Borrower, any of its Subsidiaries, or any director or officer or, to the Borrower's knowledge, employee or agent of the Borrower or any of its Subsidiaries, is violating any Anti-money Laundering Laws.
- (k) <u>Sanctions</u>. None of the Borrower, any of its Subsidiaries, or any director or officer or, to the Borrower's knowledge, employee or agent of the Borrower or any of its Subsidiaries, is a Sanctioned Person or has any commercial dealings with, or investments in, any Sanctioned Country.
- (l) <u>No Event of Default or Potential Default.</u> No Event of Default and no Potential Default has occurred and is continuing.
- (m) <u>No Material Adverse Effect</u>. The Borrower is not aware of any information or events that have resulted, or that could reasonably be expected to result, in a Material Adverse Effect relating to the Borrower.
- (n) <u>Compliance with Laws</u>. Each of the Borrower and its Material Subsidiaries is in compliance with all Applicable Laws (including environmental laws), all Governmental Approvals held by or binding upon the Borrower or its assets and all applicable restrictions imposed by all Governmental Authorities, domestic or foreign, except where the failure to comply with Applicable Laws (other than Anti-Corruption Laws, Anti-Money Laundering Laws, or Sanctions), such Governmental Approvals or re-

strictions, individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Effect.

(o) Regulation.

(i) <u>Investment Company</u>. The Borrower is not an "investment company" as defined in the Investment Company Act of 1940.

(ii) <u>Margin Stock</u>. The Borrower is not engaged principally, or as one of its important activities, in the business of extending credit for the purpose, whether immediate, incidental or ultimate, of buying or carrying "margin stock" within the meaning of Regulation U, and no part of the proceeds of the Loans will be used for the purpose, whether immediate, incidental or ultimate, of buying or carrying any such "margin stock".

(p) <u>Taxes</u>.

- (i) The Borrower has timely filed or caused to be filed all material Tax returns and reports required to have been filed and has paid or caused to be paid all material Taxes required to have been paid by it, except Taxes that are contested by the Borrower on a timely basis in good faith and by appropriate proceedings.
- (ii) There are no Taxes imposed by Colombia either (x) on or by virtue of the execution, enforcement or admissibility into evidence of any of the Loan Documents or any of the transactions thereby or (y) on any payment to be made pursuant to any Loan Document.
- (q) Validity under Colombian law. The Loan Documents are in proper form under Colombian law for their enforcement thereof. To ensure the legality, validity or enforceability of, and the priority of the obligations incurred by the Borrower under the Loan Documents in Colombia, or establish the admissibility into evidence of any of the Loan Documents in any court in Colombia, it is not necessary that any Loan Document be filed or recorded with any Colombian governmental agency or body or court, except for those referred to in Schedule 6.01(b), or that any stamp or similar tax be paid in Colombia on or in respect of any Loan Document for its enforcement in Colombia.
- final judgment for a fixed or readily calculable sum of money rendered by any court of the State of New York or the United States of America located in the State of New York based upon any of the Loan Documents (excluding the Notes) would be declared enforceable in the Republic of Colombia in the courts of the Republic of Colombia against the Borrower without reexamination, review of the merits of the cause of action in respect of which the original judgment was given or re-litigation of the matters adjudicated; provided, however, that recognition of such judgment in Colombia shall be subject to Colombian procedural Applicable Laws, in particular Colombia's Code of Civil Procedure (Código de Procedimiento Civil) (which will be abrogated by article 251 of Law 1564 of 2012 (Código General del Proceso)

once it is ruled effective by the Colombian Council of the Judiciary (*Consejo Superior de la Judicatura*)).

- (s) <u>Disclosure of Information</u>. No information that has been made available to the Administrative Agent or any Lender or the representatives or agents of the foregoing by or on behalf of the Borrower in connection with the transactions contemplated hereby, taken as a whole, contains any untrue statement of a material fact or omits to state a material fact necessary to make the statements made therein not misleading in light of the circumstances in which they are made; *provided* that with respect to any projected financial information, the Borrower represents only that such information was prepared in good faith based upon assumptions believed to be reasonable at the time.
- (t) <u>Properties</u>. The Borrower has good title to, or valid leasehold interests in, all material property necessary to conduct its business as conducted from time to time in good working order and condition, except to the extent failure to do so could not reasonably be expected to have a Material Adverse Effect.
- (u) <u>Insurance</u>. The Borrower maintains insurance with financially sound and reputable insurers against losses, damages or other risks (including, without limitation, risks and liability to Persons and property) to its assets and properties as are customarily maintained by prudent and experienced Persons engaged in the same or similar businesses operating in the same or similar jurisdictions and the Borrower deems, in its reasonable judgment, to be appropriate.
- (v) Ranking. The payment obligations of the Borrower under this Agreement and the Notes will at all times constitute the direct, general, unsecured, unsubordinated and unconditional obligations of the Borrower and rank in all respects at least *pari passu* in priority of payment with all other senior, unsecured and unsubordinated External Indebtedness of the Borrower, other than that which is preferred solely by the insolvency and/ or bankruptcy Applicable Laws of Colombia, including Law 550 of 1999 of Colombia ("Law 550").
- (w) International Banking Facility. The Borrower, an entity located outside the United States of America, understands that it is the policy of the Board of Governors of the U.S. Federal Reserve System that extensions of credit by international banking facilities, such as the Loan hereunder, may be used to finance the non-U.S. operations of the Borrower or the Borrower's Subsidiaries located outside the United States.
- (x) <u>Use of Proceeds</u>. 45% of the proceeds of the Loan shall be used to fund the investments required in connection with the Ituango Project, while the remaining 55% shall be used to fund the investments required in connection with the Borrower's 2015-2022 investment plan.

(y) Anti-bribery. None of the Borrower, any of its Subsidiaries, nor to the knowledge of the Borrower, any director, officer, agent, employee or other person acting on behalf of the Borrower or any of its Subsidiaries is aware of or has taken any action, directly or indirectly, that would result in a violation by such persons of any applicable Anti-Corruption Laws. Furthermore, the Borrower and its Subsidiaries have instituted and maintain policies and procedures designed to promote and achieve, and which are reasonably expected to continue to promote and achieve, continued compliance therewith.

SECTION 7. COVENANTS

7.01 Affirmative Covenants of the Borrower

- . The Borrower covenants and agrees that, until all amounts owing under the Loan Documents (other than contingent indemnification obligations) have been paid in full, the Borrower shall:
- (a) <u>Notice of Defaults</u>. Promptly but in no event later than ten (10) Business Days after the Borrower becomes aware of the occurrence of an Event of Default, furnish to the Administrative Agent written notice of the particulars of such occurrence and the corrective action proposed to be taken by the Borrower with respect thereto.
- Financial Reports. No later than (b) one hundred and eighty (180) days after the end of each of the Borrower's fiscal years, commencing with the fiscal year ending December 31, 2015, the Borrower shall furnish to the Administrative Agent a copy of the Borrower's annual consolidated financial statements, including its balance sheet, statement of income, and statement of cash flow for that fiscal year, all of which shall have been audited by an independent accounting firm of internationally recognized standing. The Borrower shall furnish to the Administrative Agent, no later than ninety (90) days after the end of each of the Borrower's fiscal quarters, a copy of the Borrower's quarterly unaudited unconsolidated interim financial statements. All financial reports to be submitted to the Administrative Agent shall be prepared in accordance with Colombian Government Entity GAAP or, for any financial statements in respect of periods commencing on or after January 1, 2015, IFRS, as the case may be, in each case as consistently applied, shall be in the English language (or accompanied by an accurate English translation), shall (in the case of the Borrower's annual consolidated financial statements) include the auditor's opinion and any accompanying notes, and shall fairly present in all material respects the financial condition of the Borrower and the results of its operations for the periods covered.
- (c) <u>Inspections</u>. The Borrower will permit, upon reasonable prior notice and during normal business hours, representatives of the Administrative Agent or any Lender, at their own cost and expense, to make no more than three (3) inspections *per annum* of the Borrower's books and records, and cause the officers and employees of the Borrower to give full cooperation and assistance in connection therewith; *provided* that, if an Event

- of Default has occurred and is continuing the Borrower will reimburse the Administrative Agent and any such Lender for such costs and expenses, and the numbers of inspections permitted pursuant to this <u>Section 7.01(c)</u> shall be as reasonably determined by the Administrative Agent or the Lender, as the case may be. The Administrative Agent and the Lenders shall use reasonable efforts to coordinate such visits and inspections in order to reduce their number, frequency and cost.
- (d) Government Approvals. Promptly obtain and maintain all consents, licenses, permits, authorizations and approvals of, and exemptions by, any Governmental Authority that are necessary: (i) for the execution, delivery, performance, and observance by the Borrower of the Loan Documents, including all approvals relating to the availability and transfer of U.S. Dollars required to make all payments due under the Loan Documents; and (ii) for the validity, binding effect and enforceability of the Loan Documents.
- (e) <u>Pari Passu</u>. Ensure that its payment obligations under this Agreement and the Notes will at all times constitute the direct, general, unsecured, unsubordinated and unconditional obligations of the Borrower and rank in all respects at least *pari passu* in priority of payment with all other senior, unsecured and unsubordinated External Indebtedness of the Borrower, other than that which is preferred solely by the insolvency and/or bankruptcy Applicable Laws of Colombia, including Law 550 of 1999 of Colombia.
- (f) Other Acts. From time to time, do and perform any and all acts and, subject to its having received any required prior approval of the Ministry of Finance, execute any and all documents as may be necessary or as reasonably requested by the Administrative Agent in order to effect the purposes of this Agreement and to protect the rights of the Administrative Agent and the Lenders hereunder and under the Notes (and execution of any such documents by the Borrower shall be deemed to be a certification that such Ministry of Finance approval has been obtained).
- (g) <u>Material Adverse Effect</u>. As soon as practicable, but in any event no later than ten (10) Business Days after it has knowledge of the same, provide notification to the Administrative Agent of any Material Adverse Effect.
- (h) <u>Compliance with Laws</u>. The Borrower will, and will use its reasonable best efforts to cause each of its Material Subsidiaries to, comply with all Applicable Laws (including environmental laws), all Governmental Approvals held by or binding upon it or its assets and all applicable restrictions imposed by all Governmental Authorities, domestic or foreign, in respect of the conduct of its business and the ownership of its property, except (unless such non-compliance relates to any Anti-Corruption Laws, Anti-Money Laundering Laws or Sanctions) to the extent any non-compliance is not reasonably expected to have a Material Adverse Effect.

- (i) <u>Maintenance of Existence</u>. (a) Except as otherwise permitted by <u>Section 7.02(b)</u>, the Borrower shall maintain its corporate existence, and (b) take all reasonable actions to maintain all rights, privileges and the like necessary or desirable in the normal conduct of business, activities or operations, except to the extent failure to do so could not reasonably be expected to have a Material Adverse Effect.
- (j) <u>Preservation of Assets</u>. The Borrower shall maintain all material property necessary to conduct its business as conducted from time to time in good working order and condition, except to the extent failure to do so could not reasonably be expected to have a Material Adverse Effect.
- (k) <u>Insurance</u>. The Borrower shall maintain insurance on its material property with financially sound and reputable insurance companies against such risks, of such types, on such properties and in such amounts as may from time to time be prudent for the Borrower's businesses, as determined by the Borrower in the exercise of its reasonable judgment.
- (l) <u>Use of Proceeds</u>. The Borrower shall use 45% of the proceeds of the Loan to fund the investments required in connection with the Ituango Project, and the remaining 55% to fund the investments required in connection with the Borrower's 2015-2022 investment plan. No part of the proceeds of the Loan will be used by the Borrower to directly or indirectly, immediately, incidentally or ultimately purchase or carry any Margin Stock or to extend loans to others for the purpose of purchasing or carrying any Margin Stock or to refinance or refund indebtedness originally incurred for such purpose or to fund the U.S. operations of the Borrower or the Borrower's affiliates.
- <u>Taxes</u>. The Borrower shall file all (m) tax returns required to be filed in any jurisdiction and pay and discharge all taxes shown to be due and payable on such returns and all other taxes, assessments, governmental charges, concession fees or levies imposed on it or any of its property, income or franchises, to the extent such taxes and assessments have become due and payable and before they have become delinquent, and all claims for which sums have become due and payable that have or might become a Lien (other than a Permitted Lien) on its Property except where (i) the amount, applicability or validity of such tax, assessment of claim is contested by the Borrower on a timely basis in good faith and by appropriate proceedings or (ii) the failure to do so could not reasonably be expected to have a Material Adverse Effect.
- (n) <u>Books and Records</u>. The Borrower shall keep proper books and records and accounts adequate to reflect truly and fairly in all material respects its financial condition and results of operations in conformity with Colombian Government Entity GAAP or IFRS (as applicable) and Applicable Law.
- (o) <u>Environmental Report</u>. No later than 90 days after the end of each of the Borrower's fiscal

years, commencing with the fiscal year ending December 31, 2015, the Borrower shall furnish to the Administrative Agent an Environmental Report.

7.02 <u>Negative Covenants of the Borrower</u>

The Borrower covenants and agrees that, until all amounts owing under the Loan Documents (other than contingent indemnification obligations) and the Notes have been paid in full.

- (a) <u>Liens</u>. The Borrower shall not, and shall not permit any of its Material Subsidiaries to, directly or indirectly create, incur, assume, permit or suffer to exist any Liens, except Permitted Liens.
- (b) <u>Merger, Consolidation, Dissolution, and Sale.</u>
- The Borrower shall not merge or consolidate with any other entity, dissolve or terminate its legal existence, or, directly or indirectly, sell, lease, transfer or otherwise dispose of, or permit any of its Material Subsidiaries to sell, lease, transfer or otherwise dispose of all or substantially all of the properties of the Borrower and its Material Subsidiaries (taken as a whole), unless in each case (1) for any such transaction involving the Borrower, the successor entity or entities, each of which shall be organized under the laws of Colombia or any country that is a member of the Organization for Economic Cooperation and Development (OECD), shall assume all the obligations of the Borrower under this Agreement and the Notes, (2) immediately after giving effect to the transaction, no Event of Default or Potential Default shall have occurred and be continuing, and (3) for any such transaction involving the Borrower, the Borrower delivers such certificates, opinions of its counsel and other documents regarding such transaction as may be required by the Administrative Agent in form and substance reasonably satisfactory to the Administrative Agent.
- (ii) For the avoidance of doubt: (A) in addition to the foregoing permitted transactions, the following transactions are expressly permitted under this Section 7.02(b): (1) mergers and consolidations of Subsidiaries into the Borrower, and (2) mergers, consolidations, sales, leases, transfers, divestitures or reorganizations among Subsidiaries; and (B) nothing in this Section 7.02(b) shall prohibit the Borrower or any Subsidiary from entering into build-lease-transfer, build-operate-transfer or similar project financing arrangements, provided that such arrangements are for (x) new greenfield projects or (y) the expansion of existing project assets or properties in which such arrangements extend only to the expansion assets and not in any substantial respect to the existing assets.
- (c) <u>Sanctions</u>. The Borrower shall not, directly or indirectly, use the proceeds of the Loans, or lend, contribute or otherwise make available such proceeds to any Subsidiary, joint venture partner or other Person, (i) to fund any activities or business of or with any Sanctioned

Person, or in any Sanctioned Country, in a manner that violates Sanctions, or (ii) in any other manner that would result in a violation of Sanctions by any party hereto, including any Person participating in the Loan.

- (d) Anti-Money Laundering Laws. The Borrower will not (i) directly or indirectly use the proceeds of the Loans, or (ii) lend, contribute or otherwise make available proceeds of the Loans to its Subsidiaries, any director, officer, employee, or agent of the Borrower or its Subsidiaries, joint venture partner or other Person, in any manner that would result in a violation of any Anti-Money Laundering Laws by any Person.
- (e) <u>Change in Business</u>. The Borrower shall not engage in any activities or businesses substantially different than any activities or businesses conducted by the Borrower or its Subsidiaries as of the Agreement Date or any activities or businesses reasonably ancillary, complementary or related thereto.
- (f) <u>Transactions with Affiliates</u>. The Borrower will not, and will not permit any Material Subsidiary to, enter into any transaction or series of related transactions with any Affiliate of the Borrower, other than (i) on terms and conditions substantially as favorable to the Borrower as would reasonably be obtained at that time in a comparable arm's-length transaction with a Person other than such Affiliate and (ii) as otherwise permitted or required under the Borrower's Code of Corporate Governance.
- (g) Regulation U. The Borrower shall not use the proceeds of any Loan, whether directly or indirectly, and whether immediately, incidentally or ultimately, to purchase or carry margin stock (within the meaning of Regulation U of the Board of Governors of the Federal Reserve System of the United States) or to extend credit to others for the purpose of purchasing or carrying margin stock or to refund indebtedness originally incurred for such purpose.
- (h) <u>Anti-Corruption</u>. The Borrower shall not use any part of the proceeds of the Loans, directly or indirectly, for any payment that would constitute a violation of any applicable Anti-Corruption Laws.

SECTION 8. EVENTS OF DEFAULT AND REMEDIES

8.01 Events of Default

Each of the following events or conditions shall be an "Event of Default":

(a) any failure by the Borrower to pay when due any amount owing under a Loan Document (unless in respect of any payment other than a payment of principal, the relevant amount is paid in full within five (5) Business Days of the due date thereof);

- (b) any failure by the Borrower to comply with its obligations under Sections 7.01(a), (e), (g), (i) or (l), or Section 7.02;
- (c) any representation or warranty made by the Borrower in any Loan Document or in connection therewith, or any statement made in any certificate, report or financial statement furnished by the Borrower, has been demonstrated to have been false or misleading in any material respect when made or deemed made, *provided* that such false or misleading statement shall not constitute an Event of Default if such condition or circumstance is (i) subject to cure and (ii) the facts or conditions giving rise to such misrepresentation or misstatement are cured in such a manner as to eliminate such misrepresentation or misstatement within 30 (thirty) days after the Borrower's having knowledge thereof;
- (d) any failure by the Borrower to perform or comply with any of the covenants or provisions set forth in any Loan Document (exclusive of any events specified as an Event of Default in any other subsection of this Section 8.01), which failure remains uncured for a period of thirty (30) days;
- (i) any event specified in any agreement or instrument under which there may be issued, or by which there may be secured or evidenced, External Indebtedness of the Borrower or any Material Subsidiary thereof shall occur and shall result in such External Indebtedness in an aggregate principal amount in excess of U.S.\$ fifty million (U.S.\$ 50,000,000) (or its equivalent) becoming or being declared due and payable prior to the date on which it would otherwise have become due and payable; or (ii) the Borrower fails to make any payment when due (whether by scheduled maturity, required payment, acceleration, demand, or otherwise) in respect of External Indebtedness in an aggregate principal amount in excess of U.S.\$ fifty million (U.S.\$ 50,000,000) (or its equivalent), and such failure shall continue after the applicable notice and grace period, if any, specified in the corresponding agreements or instruments;
- (A)(i) the Borrower shall be unable generally to pay its debts as they fall due or shall admit in writing its inability to pay its debts as they fall due or shall become insolvent; (ii) the Borrower shall apply for or consent to the appointment of any liquidator, receiver, trustee, síndico, conciliador or administrator for all or a substantial part of its business, properties, assets, or revenues; or (iii) a liquidator, receiver, trustee, or administrator shall be appointed for the Borrower and such appointment shall continue undismissed, undischarged or unstaved for a period of ninety (90) days; (B) the Borrower shall institute (by petition, application, answer, consent or otherwise) any bankruptcy, arrangement, readjustment of debt, dissolution, liquidation, proceso de reestructuración, proceso de reorganización, proceso de insolvencia, concurso mercantil, quiebra, or similar executory or judicial proceeding; (C) a bankruptcy, arrangement, readjustment of debt, dissolution, liquidation, proceso de reestructuración, proceso de reorganización, proceso de insolvencia, concurso mercantil, quiebra, or similar executory or judicial proceeding shall

- be instituted against the Borrower and such proceeding shall remain undismissed, undischarged or unstayed for a period of ninety (90) days; (D) the Borrower shall take any action seeking to take advantage of any other law relating to its bankruptcy, insolvency, liquidation, termination, dissolution, winding up, or composition, or readjustment of debts; or (E) the Borrower shall take any corporate or similar action for the purpose of effecting any of the foregoing;
- (g) any final, non-appealable judgment against the Borrower or any Material Subsidiary (i) shall have been entered on a claim not covered by insurance in an aggregate amount of U.S.\$ fifty million (U.S.\$ 50,000,000) (or its equivalent in another currency) or more, and (ii) such judgment has not been removed, vacated, discharged or satisfied for a period of sixty (60) days from the date of such final judgment;
- (h) any Governmental Authority shall have (i) condemned, seized or otherwise expropriated (either through a single act or a series of acts) all or substantially all of the property of the Borrower or (ii) taken any action that materially curtails the authority of the Borrower to conduct its business:
- (i) any authorization, approval, Governmental Approval, consent, license, exemption, filing, registration, notarization or other requirement of any Governmental Authority necessary to enable the Borrower to comply with its obligations under any Loan Document shall have been revoked, rescinded, suspended, held invalid or otherwise limited in effect in a manner that could reasonably be expected to have a Material Adverse Effect; or
- (j) any Loan Document ceases to be in full force and effect or is declared in a final, non-appeal-able judgment to be unenforceable against the Borrower (in each case, other than as a result of any action or inaction on the part of the Administrative Agent or any Lender), the validity or enforceability of any Loan Document at any time is challenged by the Borrower; or the Borrower repudiates any Loan Document, or does or causes to be done any act or thing evidencing an intention to repudiate any Loan Document.

8.02 Remedies

. If any Event of Default shall occur and be continuing, the Administrative Agent may (and shall, upon the direction of Required Lenders), by notice to the Borrower, declare (a) any and all amounts of principal outstanding under this Agreement and the Notes to be forthwith due and payable together with accrued interest and any and all other amounts payable or owing hereunder, whereupon the same shall become forthwith due and payable and (b) declare the Commitments to be terminated forthwith, whereupon the Commitments shall immediately terminate; provided that if such event is an Event of Default specified in Section 8.01(f) above, automatically the Commitments shall immediately terminate and the Loans hereunder (with accrued interest thereon) and all other amounts owing under this Agreement and the Notes shall immediately be-

come due and payable. Presentment, demand, protest or notice of any kind (other than the notice provided for in the first sentence of this paragraph) are expressly waived, anything in this Agreement to the contrary notwithstanding. The aforementioned right to accelerate is in addition to and not a substitute for any other rights and remedies available under this Agreement and under Applicable Law.

SECTION 9. GOVERNING LAW AND JURISDICTION

9.01 Governing Law

THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK.

Notwithstanding the foregoing, all matters governing the authorization and execution of the Loan Documents by the Borrower shall be governed by and construed in accordance with the laws of Colombia.

9.02 Submission to Jurisdiction

EACH OF THE PARTIES TO THIS AGREEMENT AGREES THAT ANY SUIT, ACTION OR PROCEEDING WITH RESPECT TO THIS AGREEMENT, ANY OTHER LOAN DOCUMENT OR ANY JUDGMENT ENTERED BY ANY COURT IN RESPECT THEREOF MAY BE BROUGHT IN THE UNITED STATES DISTRICT COURT FOR THE SOUTHERN DISTRICT OF NEW YORK AND, EXCEPT IN THE CASE OF ANY SUIT, ACTION, PROCEEDING OR JUDGMENT BROUGHT AGAINST THE LENDERS OR THE ADMINISTRATIVE AGENT, IN THE SUPREME COURT OF THE STATE OF NEW YORK SITTING IN NEW YORK COUNTY (INCLUDING ITS APPELLATE DI-VISION), OR IN ANY APPELLATE COURT THEREOF. EACH PARTY TO THIS AGREEMENT HEREBY IRREVO-CABLY SUBMITS TO THE NON-EXCLUSIVE JURISDIC-TION OF SUCH COURTS FOR THE PURPOSE OF ANY SUCH SUIT, ACTION, PROCEEDING OR JUDGMENT. EACH PARTY CONSENTS THAT ANY SUCH ACTION OR PROCEEDING MAY BE BROUGHT IN SUCH COURTS AND WAIVES ANY OBJECTION THAT IT MAY NOW OR HEREAFTER HAVE TO THE VENUE OF ANY SUCH ACTION OR PROCEEDING IN ANY SUCH COURT OR THAT SUCH ACTION OR PROCEEDING WAS BROUGHT IN AN INCONVENIENT COURT AND AGREES NOT TO PLEAD OR CLAIM THE SAME. EACH PARTY HERE-TO FURTHER SUBMITS, FOR THE PURPOSE OF ANY SUCH SUIT, ACTION, PROCEEDING OR JUDGMENT BROUGHT OR RENDERED AGAINST IT, TO THE AP-PROPRIATE COURTS OF THE JURISDICTION OF ITS DOMICILE. EACH OF THE PARTIES TO THIS AGREE-MENT AGREES THAT A JUDGMENT. AFTER EXHAUS-TION OF ALL AVAILABLE APPEALS, IN ANY SUCH AC-TION OR PROCEEDING SHALL BE CONCLUSIVE AND BINDING UPON IT, AND MAY BE ENFORCED IN ANY OTHER JURISDICTION, INCLUDING BY A SUIT UPON SUCH JUDGMENT, A CERTIFIED COPY OF WHICH SHALL BE CONCLUSIVE EVIDENCE OF THE JUDG-MENT.

9.03 Service of Process

- (a) In the case of the courts of the State of New York or of the federal courts sitting in the State of New York, the Borrower hereby designates, appoints and empowers CT Corporation System, located at 111 8th Avenue #13, New York, NY 10011, as its authorized agent to accept, receive, and acknowledge for and on behalf of the Borrower, its properties and revenues, service of any and all process that may be served in any action, suit or proceeding of the nature referred to above in the State of New York, which appointment shall be irrevocable until the appointment and acceptance of a successor authorized agent pursuant to the provisions of Section 9.03(d).
- The Borrower further agrees that such service of process may be made personally or by mailing or delivering a copy of the summons and complaint or other legal process in any such legal suit, action or proceeding to the Borrower, in care of its respective agent designated above at the aforesaid address (or, if the Borrower shall have designated a successor agent for service of process, the address of the successor agent for service of process), and each such agent is hereby authorized, respectively, to accept, receive, and acknowledge the same for and on behalf of the Borrower and to admit service with respect thereto. Service upon each such agent shall be deemed to be personal service on the Borrower and shall be legal and binding upon the Borrower for all purposes notwithstanding any failure to mail copies of such legal process to the Borrower or any failure on the part of the Borrower to receive the same, and shall be deemed completed upon the delivery thereof to such agent whether or not such respective agent shall give notice thereof to the Borrower or upon the earliest other date permitted by Applicable Law (including the United States Foreign Sovereign Immunities Act of 1976, as amended).
- (c) To the extent permitted by Applicable Law, including treaties by which the United States and Colombia are bound, the Borrower further irrevocably agrees to the service of process of any of the aforementioned courts in any suit, action or proceeding by the mailing of copies thereof by certified mail, postage prepaid, return receipt requested, to the Borrower at the address referenced in Section 11.02, such service to be effective upon the date indicated on the postal receipt returned from the Borrower.
- (d) The Borrower agrees that it will at all times continuously maintain an agent to receive service of process in the State of New York on behalf of itself and its properties and revenues, and, in the event that for any reason its agent designated above shall not serve as agent for the Borrower to receive service of process in the State of New York on its behalf, the Borrower shall promptly appoint a successor satisfactory to the Administrative Agent so to serve, advise the Administrative Agent thereof, and deliver to the Administrative Agent evidence in writing of the successor agent's acceptance of such appointment for a term extending at least one year beyond the Maturity Date and that such successor agent has been paid in full for such term. The foregoing provisions constitute, among

other things, a special arrangement for service between the parties to this Agreement for the purposes of 28 U.S.C. § 1608.

9.04 Waiver of Immunity

TO THE EXTENT THAT THE BORROWER MAY BE OR BECOME ENTITLED, IN ANY JURISDICTION IN WHICH JUDICIAL PROCEEDINGS MAY AT ANY TIME BE COMMENCED WITH RESPECT TO THIS AGREEMENT OR ANY OTHER LOAN DOCU-MENT, TO CLAIM FOR ITSELF OR ITS PROPERTIES OR REVENUES ANY IMMUNITY FROM SUIT, COURT JURISDICTION, ATTACHMENT PRIOR TO JUDGMENT, ATTACHMENT IN AID OF EXECUTION OF A JUDGMENT, EXECUTION OF A JUDGMENT OR FROM ANY OTHER LEGAL PROCESS OR REMEDY RELATING TO ITS OB-LIGATIONS UNDER THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT, AND TO THE EXTENT THAT IN ANY SUCH JURISDICTION THERE MAY BE ATTRIBUTED SUCH AN IMMUNITY (WHETHER OR NOT CLAIMED), THE BORROWER HEREBY IRREVOCABLY AGREES NOT TO CLAIM AND HEREBY IRREVOCABLY WAIVES SUCH IMMUNITY TO THE FULLEST EXTENT PERMIT-TED BY THE LAWS OF SUCH JURISDICTION, BUT (I) SUBJECT TO THE LIMITATIONS PROVIDED IN ARTI-CLES 192, 195, 298 AND 299 OF LAW 1437 OF 2011 (CÓ-DIGO DE PROCEDIMIENTO ADMINISTRATIVO Y DE LO CONTENCIOSO ADMINISTRATIVO) AND (II) ARTICLES 684 AND 513 OF THE COLOMBIAN CIVIL PROCEDURE CODE (CÓDIGO DE PROCEDIMIENTO CIVIL) (WHICH WILL BE GRADUALLY SUPERSEDED BY ARTICLES 593, 594 AND 595 ET AL SUBJECT TO THE ENTRY INTO FORCE OF LAW 1564 OF 2012 (CÓDIGO GENERAL DEL PROCESO) PURSUANT TO THE TERMS OF ARTICLE 627 THEREOF).

(b) THE BORROWER AGREES THAT THE WAIVERS SET FORTH IN PARAGRAPH (A) ABOVE SHALL HAVE THE FULLEST EFFECT PERMITTED UNDER THE FOREIGN SOVEREIGN IMMUNITIES ACT OF 1976 OF THE UNITED STATES OF AMERICA (28 U.S.C. §§ 1602-1611) AND ARE INTENDED TO BE IRREVOCABLE AND NOT SUBJECT TO WITHDRAWAL FOR PURPOSES THEREOF.

9.05 <u>Waiver of Security Requirements</u>

. TO THE EXTENT THE BORROWER MAY, IN ANY ACTION, SUIT OR PROCEEDING BROUGHT IN ANY OF THE COURTS REFERRED TO IN SECTION 9.02 OR OTHERWISE ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT, BE ENTITLED TO THE BENEFIT OF ANY PROVISION OF LAW REQUIRING THE ADMINISTRATIVE AGENT OR ANY LENDER IN SUCH ACTION, SUIT OR PROCEEDING TO POST SECURITY FOR THE COSTS OF THE BORROWER OR TO POST A BOND OR TO TAKE SIMILAR ACTION, AS THE CASE MAY BE, THE BORROWER HEREBY IRREVOCABLY WAIVES SUCH BENEFIT, IN EACH CASE TO THE FULLEST EXTENT NOW OR HEREAFTER PERMITTED UNDER APPLICABLE LAW.

9.06 No Limitation

. Nothing in this <u>Section 9</u> shall affect the right of the Administrative Agent or any Lender to serve process in any other manner permitted by Applicable Law or to commence legal proceedings or otherwise proceed against the Borrower in Colombia or in any other jurisdiction.

SECTION 10. THE ADMINISTRATIVE AGENT

10.01 Appointment

. Each Lender hereby designates and appoints The Bank of Tokyo-Mitsubishi UFJ, Ltd. as Administrative Agent to act as specified herein. The Bank of Tokyo-Mitsubishi UFJ, Ltd., as Administrative Agent, is authorized to execute and deliver this Agreement and to take such action on behalf of the Lenders under the provisions of this Agreement and to exercise such powers and to perform such duties hereunder as are specifically delegated to or required of the Administrative Agent by the terms hereof and thereof and such other powers as are reasonably incidental thereto.

10.02 Nature of Duties

- (a) The Administrative Agent shall:
- (i) promptly inform each Lender of the contents of any notice or document received by it from the Borrower; and
- $\label{eq:promptly} (ii) \qquad \text{promptly provide each notice as may be required of it pursuant to this Agreement.}$
- (b) The Administrative Agent may perform any of its duties hereunder by or through its officers, directors, agents or any other authorized employee. The exculpatory provisions of this <u>Section 10</u> shall apply to any such sub-agent and to the Related Parties of the Administrative Agent and any such sub-agent.
- The Administrative Agent shall (c) have no duties or responsibilities except those expressly set forth in this Agreement. The duties of the Administrative Agent to the Lenders shall be mechanical and administrative in nature; the Administrative Agent shall not have by reason of this Agreement a fiduciary relationship in respect of any Lender; and nothing in this Agreement, expressed or implied, is intended to or shall be so construed as to impose upon the Administrative Agent any obligations in respect of this Agreement or the Loan except as expressly set forth herein or therein. Without limiting the generality of the foregoing, (i) the Administrative Agent shall not be subject to any fiduciary or other implied duties, regardless of whether a Potential Default or Event of Default has occurred and is continuing, (ii) the Administrative Agent shall not have any duty to take any discretionary action or exercise any discretionary powers, except discretionary rights and powers expressly contemplated hereby that the Administrative Agent is required to exercise in writing as directed by the Required Lenders (or such other number or percentage of

the Lenders as shall be necessary under the circumstances as provided herein), and (iii) except as expressly set forth herein, the Administrative Agent shall not have any duty to disclose, and shall not be liable for the failure to disclose. any information relating to the Borrower or any of its Subsidiaries that is communicated to or obtained by the bank serving as Administrative Agent or any of its Affiliates in any capacity. The Administrative Agent shall not be liable for any action taken or not taken by it with the consent or at the request of the Required Lenders (or such other number or percentage of the Lenders as shall be necessary under the circumstances as provided herein) or in the absence of its own gross negligence or willful misconduct. The Administrative Agent shall be deemed not to have knowledge of Event of Default unless and until written notice thereof is given to the Administrative Agent by the Borrower or a Lender, and the Administrative Agent shall not be responsible for or have any duty to ascertain or inquire into (A) any statement, warranty or representation made in or in connection with this Agreement, (B) the contents of any certificate, report or other document delivered hereunder or in connection herewith, (C) the performance or observance of any of the covenants, agreements or other terms or conditions set forth herein, (D) the validity, enforceability, effectiveness or genuineness of this Agreement or any other agreement, instrument or document, or (E) the satisfaction of any condition set forth in Section 5 or elsewhere herein, other than to confirm receipt of items expressly required to be delivered to the Administrative Agent.

10.03 <u>Lack of Reliance on the Administrative</u> <u>Agent</u>

Independently and without reliance upon the Administrative Agent, each Lender, to the extent it deems appropriate, has made and shall continue to make (i) its own independent investigation of the financial condition and affairs of the Borrower in connection with the Loan and the taking or not taking of any action in connection herewith and (ii) its own appraisal of the credit worthiness of the Borrower, and the Administrative Agent and each other Lender shall have no duty or responsibility, either initially or on a continuing basis, to provide any Lender with any information with respect to this Agreement or the Borrower other than as expressly provided herein. Each Lender shall, independently and without reliance upon the Administrative Agent or any other Lender and based on such documents and information (which may contain material, non-public information within the meaning of the United States securities laws concerning the Borrower and its Affiliates) as it shall from time to time deem appropriate, continue to make its own decisions in taking or not taking action under or based upon this Agreement, any related agreement or any document furnished hereunder or thereunder and in deciding whether or to the extent to which it will continue as a Lender or assign or otherwise transfer its rights, interests and obligations hereunder.

10.04 Reliance

The Administrative Agent shall be entitled to rely, and shall be fully protected in relying, upon any writing, resolution, notice, statement, certificate, telex, teletype or telecopier message, cablegram, radiogram, order or other document or telephone message signed, sent or made by any Person that the Administrative Agent believed to be the proper Person, and, with respect to all legal matters pertaining to this Agreement and its duties hereunder, upon advice of counsel selected by it.

10.05 Consultation with Experts

The Administrative Agent may consult with legal counsel, independent certified public accountants and other experts selected by it in connection with the performance of its duties hereunder and shall not be liable for any action taken or omitted to be taken by it in good faith in accordance with the advice of such counsel, accountants or experts.

10.06 Indemnification

To the extent the Administrative Agent is not reimbursed and indemnified by the Borrower, each Lender will reimburse and indemnify the Administrative Agent in proportion to its respective interest (determined at the time such reimbursement or indemnity is sought) for and against any and all liabilities, obligations, losses, damages, penalties, claims, actions, judgments, suits, costs, expenses or disbursements of whatsoever kind or nature (collectively, "Liabilities") which may be imposed on, asserted against or incurred by the Administrative Agent in performing its duties hereunder or in any way relating to or arising out of this Agreement; provided that no Lender shall be liable for any portion of such Liabilities resulting from the Administrative Agent's gross negligence or willful misconduct.

10.07 <u>The Administrative Agent in Its Individual</u> <u>Capacity</u>

The Administrative Agent and each Lender may accept deposits from, lend money to, and generally engage in any kind of banking, trust, financial advisory role or other business with the Borrower or any affiliate thereof as if it were not performing the duties specified herein, and may accept fees and other consideration from the Borrower for services in connection with this Agreement and otherwise without having to account for the same to the other Lenders. Notwithstanding anything to the contrary expressed or implied herein, the Administrative Agent and the Lenders shall not be bound to: (a) account to any other Lender for any sum or the profit element of any sum received by it for its own account or (b) disclose to any other Person any information relating to the Borrower if such disclosure would or might in its opinion constitute a breach of any law or regulation or be otherwise actionable by suit of any Person.

10.08 Resignation by the Administrative Agent; Successor Administrative Agent

(a) The Administrative Agent may resign from the performance of all its functions and duties hereunder at any time by giving thirty (30) calendar days' prior written notice to the Borrower and the Lenders. Such resignation shall take effect upon the appointment of a successor Administrative Agent pursuant to clauses (b) or (c)

below or as otherwise provided below; *provided* that, in the event the Borrower determines in good faith that the Administrative Agent's gross negligence or willful misconduct has resulted in a breach of the Administrative Agent's obligations under this Agreement, the Borrower may, by written notice to the Administrative Agent and the Lenders, require the Administrative Agent to resign in accordance with this paragraph, which notice shall (without any further action) be deemed to be a notice of resignation delivered by the Administrative Agent to the Lenders. Upon any such resignation, the Lenders shall have the right to appoint a successor, who shall be a bank with an office in New York, New York, or an Affiliate of any such bank.

- (b) Upon any such notice of resignation (other than in the course of a deemed resignation pursuant to clause (a) above), the Lenders shall appoint a successor Administrative Agent hereunder or thereunder who shall be a bank with an office in New York, New York, or an Affiliate of any such bank.
- (c) In the event of a resignation (other than in the course of a deemed resignation pursuant to clause (a) above) if a successor Administrative Agent shall not have been so appointed within such thirty (30) calendar day period, the Administrative Agent, with the consent of the Borrower (not to be unreasonably withheld), may then appoint a successor Administrative Agent who shall serve as Administrative Agent hereunder or thereunder until such time, if any, as the Lenders appoint a successor Administrative Agent as provided above.
- (d) If the Administrative Agent becomes insolvent or unable to meet its debts as they mature or if a receiver of it or of all or any substantial part of its property shall be appointed or if an order of any court of competent jurisdiction shall be entered approving any petition filed by or against it under the provisions of any applicable bankruptcy or insolvency law, or if any public officer shall take charge or control of the Administrative Agent or of its property or affairs, for the purpose of rehabilitation, conservation or liquidation, the Lenders shall appoint a successor Administrative Agent in accordance with clause (b) above. If a successor Administrative Agent shall not have been so appointed pursuant to clause (b) above, the Lenders shall appoint a successor Administrative Agent pursuant to clause (c) above, as the case may be.

10.09 <u>No Amendment to Duties of Administrative</u> <u>Agent Without Consent</u>

The Administrative Agent shall not be bound by any waiver, amendment, supplement or modification of this Agreement which affects its rights or duties under this Agreement unless it shall have given its prior written consent, as Administrative Agent, thereto.

SECTION 11. MISCELLANEOUS

11.01 <u>Computations</u>

. Each determination of an interest rate or fee by the Administrative Agent or any Lender pursuant to any provision of this Agreement or any Note, in the absence of manifest error, shall be conclusive and binding on the Borrower. All computations of interest and fees hereunder and under any Note shall be made on the basis of a year of three hundred sixty (360) days and actual days elapsed. All such calculations shall include the first day and exclude the last day of the period of calculation.

11.02 Notices

(a) All notices or other communications required or permitted to be given hereunder (the "**Notices**") may be given to the following addresses:

If to the Borrower: EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P.

Carrera 58 # 42-125, Medellín Attn: Diana Rúa Jaramillo Phone: + 57 4 380 4200 Email: diana.rua@epm.com.co

If to the Administrative Agent: THE BANK OF TO-KYO-MITSUBISHI UFJ, LTD.

1251 Avenue of the Americas, New York, NY 10020

Attn: Lawrence Blat

Phone: +1 (212) 782 4310 / +1 (212) 782

6687

Email: agencydesk@us.mufg.jp / lblat@

us.mufg.jp

If to any Lender, as set forth on the signature page or in its Assignment and Assumption Agreement, as applicable.

Any party shall have the right to change its address for Notice hereunder to any other location by giving ten (10) days' written Notice to the other parties in the manner set forth herein above.

- (b) All Notices shall be in writing and shall be considered as properly given (A) if delivered in person, (B) if sent by overnight delivery service (including Federal Express, United Parcel Service and other similar reputable overnight delivery services), (C) in the event reputable overnight delivery services are not readily available, if mailed by first class mail, postage prepaid, registered or certified with return receipt requested, (D) if transmitted by facsimile confirmed by telephone or (E) if transmitted by electronic communication as provided in Section 11.02(d).
- (c) Notices delivered in person or by overnight courier service, or mailed by registered or certified mail, shall be effective upon receipt by the addressee. Notices transmitted by facsimile shall be deemed to have been given when transmitted, if confirmation of a successful transmission has been received (except that, in all in-

stances, if not given during normal business hours on a Business Day for recipient, shall be deemed to have been given at the opening of business on the next Business Day for the recipient). Notices delivered through electronic communications to the extent provided in <u>Section 11.02(d)</u> shall be effective as provided in such Section.

(d) Notices hereunder may be delivered or furnished by electronic communication (including email) pursuant to procedures approved by the relevant recipient.

(i) Unless the relevant recipient otherwise prescribes, Notices sent to an email address shall be deemed received upon the sender's receipt of an acknowledgement from the intended recipient (such as by the "return receipt requested" function, as available, return email or other written acknowledgement); provided that if such Notice is not sent during the normal business hours of the recipient, such Notice shall be deemed to have been sent at the opening of business on the next Business Day of the recipient.

11.03 <u>Benefit of Agreement; Assignment; Participations</u>

(a) This Agreement shall be binding upon and inure to the benefit of and be enforceable by the respective successors and assigns of the parties hereto; provided, however, that (1) the Borrower may not assign or transfer any of its rights, obligations or interest under any Loan Document, except with the prior written consent of each of the Lenders and (2) no Lender may assign or transfer any of its rights, obligations or interest under any Loan Document except in accordance with this Section 11.03 and the requirements of Applicable Law.

(b) Commencing on the thirtieth day following the Closing Date, any Lender may:

(i) assign all or a portion of the Loan due to it to (1) its parent company and/or any Affiliate of such Lender which is at least 50% owned by such Lender or its parent company, (2) one or more other Lenders or (3) in the case of any Lender that is a fund or trust that invests in bank loans or that manages or advises (directly or through an Affiliate) any fund or trust that invests in bank loans, any fund or trust that invests in bank loans and is managed or advised by the same investment advisor as a Lender, by an Affiliate of such investment advisor or by a Lender, as the case may be, provided that the assigning Lender shall give notice to the Borrower of any such assignment (which notice shall include the identity of the proposed assignee) five days prior to the effective date of such assignment in order for the Borrower to complete its internal and regulatory processes; provided, further, that failure by the assigning Lender to give such notice or failure by the Borrower to complete any such process shall not affect the validity of such assignment; or

 $(ii) \qquad \text{assign all, or if less than all, a portion equal to at least U.S.$5,000,000 of outstand- } \\$

ing principal amount of Loans due to it to one or more assignees (treating (x) any fund that invests in bank loans and (y) any other fund that invests in bank loans and is managed or advised by the same investment advisor as such fund or by an Affiliate of such investment advisor, as a single assignee); and each such assignee shall become a party to this Agreement as a Lender by execution of an Assignment and Assumption Agreement;

provided that:

(x) in each case, the assigning Lender shall surrender to the Borrower the old Note(s) held by it (or furnish a standard indemnity letter from such Lender in respect of any lost Note(s) reasonably acceptable to the Borrower), and new Notes will be issued, at the Borrower's expense, to such new Lender and to the assigning Lender (in the case of a partial assignment);

(y) so long as no Event of Default has occurred and is continuing, written consent of the Borrower shall be required in connection with any assignment pursuant to clause (ii) of this <u>Section 11.03(b)</u> (which consent, in each case, shall not be unreasonably withheld or delayed); *provided* that, the Borrower shall be deemed to have consented to an assignment unless it shall have objected thereto by written notice to the Administrative Agent within ten (10) Business Days after having received notice thereof:

(z) if an Event of Default has occurred and is continuing and the Administrative Agent has delivered notice thereof to the Borrower, the Borrower shall be deemed to have consented to any assignment following the date of such notice pursuant to clause (ii) of this Section 11.03(b); provided that, the assigning Lender shall be required to give notice to the Borrower of any such assignment (which notice shall include the identity of the proposed assignee) five days before such assignment in order for the Borrower to complete its internal and regulatory processes; provided, further, that failure by the assigning Lender to give such notice or failure by the Borrower to complete any such process shall not affect the validity of such assignment; and

(aa) in each case, the Administrative Agent shall receive at the time of each assignment, from the assigning or assignee Lender, the payment of a non-refundable assignment fee of U.S.\$ 3,500, which fee shall be paid by the assignee Lender.

Notwithstanding the foregoing provisions of this <u>Section 11.03(b)</u>, no such transfer or assignment will be effective until recorded by the Administrative Agent on the Register pursuant to <u>Section 11.03(e)</u> below and the Administrative Agent shall provide notice to the Borrower of any such transfer or assignment within three Business Days thereof. To the extent that an assignment of all or any portion of a Lender's Loan pursuant to this <u>Section 11.03(b)</u> would, due to circumstances existing at the time of such assignment, result in increased costs under <u>Section 2.09</u>, 2.10 or 4.01 from those being charged by the respective

assigning Lender prior to such assignment, then the Borrower shall not be obligated to pay such increased costs (although the Borrower shall be obligated to pay any other increased costs of the type described above resulting from changes after the date of the respective assignment).

In addition to the foregoing, any Lender may grant participations in its rights hereunder without the consent of the Borrower; provided, however, that such Lender shall remain a "Lender" for all purposes hereunder and the participant shall not constitute a "Lender" hereunder and, provided further, that no Lender shall grant any participation under which the participant shall have direct or indirect rights to approve any amendment to or waiver of this Agreement except to the extent such amendment or waiver would (i) extend the Maturity Date, or reduce the rate or extend the time of payment of interest or fees thereon (except in connection with a waiver of applicability of any post-Default increase in interest rates) or reduce the principal amount thereof, or (ii) consent to the assignment or transfer by the Borrower of any of its rights and obligations under any Loan Document. In the case of any such participation, the participant shall not have any rights under any Loan Document (the participant's rights against such Lender in respect of such participation to be those set forth in the agreement executed by such Lender in favor of the participant relating thereto) and all amounts payable by the Borrower hereunder shall be determined as if such Lender had not sold such participation.

- (d) Nothing in this Agreement shall prevent or prohibit any Lender from at any time pledging or assigning a security interest in all or any portion of its Loans and Notes hereunder to secure obligations of such Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank. No pledge or assignment pursuant to this clause (d) shall release the transferor Lender from any of its obligations hereunder.
- (e) The Administrative Agent shall maintain at one of its offices in the United States a copy of each Assignment and Assumption Agreement delivered to it and a register for the recordation of the name and address of each Lender, and principal amounts (and stated interest) of the Loan owing to each Lender pursuant to the terms hereof from time to time (the "Register"). The entries in the Register shall be conclusive absent manifest error, and the Borrower or the Lenders shall treat each Person whose name is recorded in the Register pursuant to the terms hereof as a lender hereunder for all purposes of this Agreement. The Register shall be available for inspection by the Borrower and the Lenders, at any reasonable time and from time to time upon reasonable prior notice.
- (f) Each Lender that sells a participation shall, acting solely for this purpose as a non-fiduciary agent of the Borrower, maintain a register on which it enters the name and address of each participant and the principal amounts (and stated interest) of each participant's interest in the Loans or other obligations under the Loan Documents (the "Participant Register"); provided that no Lender shall have any obligation to disclose all or any portion of the Participant Register (including the identity of

any participant or any information relating to a participants interest in any commitments, loans, letters of credit or its other obligations under any Loan Document) to any Person except to the extent that such disclosure is necessary to establish that such commitment, loan, letter of credit or other obligation is in registered form under Section 5f.103-1(c) of the United States Treasury Regulations. The entries in the Participant Register shall be conclusive absent manifest error, and such Lender shall treat each Person whose name is recorded in the Participant Register as the owner of such participation for all purposes of this Agreement notwithstanding any notice to the contrary. For the avoidance of doubt, the Administrative Agent (in its capacity as Administrative Agent) shall have no responsibility for maintaining a Participant Register.

11.04 No Waiver; Remedies Cumulative

No failure or delay on the part of the Administrate Agent or any Lender in exercising any right, power or privilege under any Loan Document and no course of dealing between or among the Borrower and the Administrative Agent or any Lender shall operate as a waiver of such right, power or privilege; nor shall any single or partial exercise of any right, power or privilege hereunder, under any Loan Document preclude any other right, power or privilege hereunder or thereunder. The rights and remedies expressly provided herein are cumulative and not exclusive of any rights or remedies that the Administrative Agent or any Lender would otherwise have. No notice to or demand on the Borrower in any case shall entitle the Borrower to any other or further notice or demand in similar or other circumstances or constitute a waiver of the rights of the Administrative Agent or any Lender to any other or further action in any circumstances without notice or demand.

11.05 Entire Agreement

The Loan Documents contain the entire agreement among the parties hereto regarding the Loan.

11.06 Amendment or Waiver; etc.

Neither this Agreement nor any terms hereof may be changed, waived, discharged or terminated unless such change, waiver, discharge or termination is in writing signed by the Borrower, and approved by the Ministry of Finance to the extent required by Colombian public indebtedness regulations, and the Required Lenders; provided that no such change, waiver, discharge or termination shall, without the consent of each Lender (i) extend any scheduled date of payment or the Maturity Date, extend the expiration date of any Lender's Commitment, or reduce the rate or extend the time of payment of interest (other than as a result of any waiver of the applicability of any post-Default increase in interest rates) thereon, or reduce the fees payable hereunder or principal amount of the Loan, (ii) amend, modify or waive any provision of this Section 11.06, (iii) amend or modify the definition of Required Lenders, (iv) amend or modify Section 3.04 in a manner that would alter the pro rata sharing of payments required thereby, (v) consent to the assignment or transfer

by the Borrower of any of its rights and obligations under any Loan Document (except in accordance with the terms hereof), (vi) change Section 3.04(b) or 3.04(c) in a manner that would alter the pro rata sharing of payments or setoffs required thereby or any other provision in a manner that would alter the pro rata allocation among the Lenders without the written consent of each Lender directly affected thereby, or (vii) amend, modify or waive any provision of Section 5; provided further, that no such change, waiver, discharge or termination shall (x) increase the Commitment of any Lender without the written consent of such Lender, or (y) without the consent of the Administrative Agent, amend, modify or waive any provision of Section 10 as same applies to the Administrative Agent or any other provision as same relates to the rights or obligations of the Administrative Agent.

If, in connection with any proposed change, waiver, discharge or termination of or to any of the provisions of this Agreement requiring the consent of Reguired Lenders and for which the consent of the Reguired Lenders is obtained but the consent of one or more other Lenders is not obtained, then the Borrower shall have the right, so long as all non-consenting Lenders are treated as described in clauses (A) or (B) below, to either (A) replace each such non-consenting Lender or Lenders with one or more Replacement Lenders pursuant to Section 2.12 so long as, at the time of such replacement, each such Replacement Lender consents to the proposed change, waiver, discharge or termination for which consent of the Required Lenders was obtained, or (B) repay the portion of the outstanding Loan due to such non-consenting Lender, so long as all Loans, together with accrued and unpaid interest and all other amounts, owing to such Lender are repaid concurrently with the effectiveness of such termination and at such time, unless the respective Lender continues to have outstanding Loans hereunder, such Lender shall no longer constitute a "Lender" for purposes of this Agreement, except with respect to indemnification under Section 10.06, which shall survive as to such repaid Lender; provided that, unless the Loans which are repaid pursuant to preceding clause (B) are immediately replaced in full at such time through the addition of new Lenders and/or outstanding Loans of existing Lenders (who in each case must specifically consent thereto), then in the case of any action pursuant to preceding clause (B), the Required Lenders (determined after giving effect to the proposed action) shall specifically consent thereto.

11.07 Counterparts

This Agreement may be executed in any number of counterparts and by the different parties hereto on separate counterparts, each of which, when executed and delivered, shall be effective for purposes of binding the parties hereto, but all of which shall together constitute one and the same instrument. Delivery of an executed counterpart of a signature page of this Agreement by telecopy, emailed pdf. or any other electronic means that reproduces an image of the actual executed signature page shall be effective as delivery of a manually executed counterpart of this Agreement. The words "execution," "signed," "signature," "delivery," and words of like import in or relating to any document to be signed in connection with this Agreement and

the transactions contemplated hereby shall be deemed to include electronic signatures, deliveries or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature, physical delivery thereof or the use of a paper-based recordkeeping system, as the case may be, to the extent and as provided for in any applicable law, including the Federal Electronic Signatures in Global and National Commerce Act, the New York State Electronic Signatures and Records Act, or any other similar state laws based on the Uniform Electronic Transactions Act.

11.08 Expenses; Indemnity

The Borrower shall pay all reasonable and documented out-of-pocket expenses incurred by the Administrative Agent and its Affiliates, in connection with the syndication of the credit facilities provided for herein, the preparation, negotiation, execution, delivery and administration of this Agreement and the other Loan Documents or any amendments, modifications or waivers of the provisions hereof or thereof (whether or not the transactions contemplated hereby or thereby shall be consummated); provided that unless otherwise agreed by the parties hereto, the amount of fees and expenses of counsel that the Borrower is required to pay or reimburse in connection with the syndication of the credit facilities provided herein, the preparation, negotiation, execution and delivery of this Agreement and the other Loan Documents shall be subject to the provisions of the Agent Fee Letter. The Borrower shall pay all documented out-of-pocket expenses incurred by each of the Administrative Agent or any Lender, including the documented fees, charges and disbursements of any counsel for the Administrative Agent or any Lender, in connection with the enforcement or protection of its rights in connection with the Loan Documents, including its rights under this Section.

The Borrower shall indemnify the (b) Administrative Agent, each Lender, their respective Affiliates, and their respective directors, officers, employees, attorneys and agents (each such Person being called an "Indemnitee") against, and hold each Indemnitee harmless from, any losses, claims, damages, liabilities and related expenses, which expenses shall include reasonable attorney's fees and expenses for no more than one Colombian counsel and one New York counsel to the Lenders and one Colombian counsel and one New York counsel to the Administrative Agent (and in the case of a conflict of interest where an entity affected by such conflict notifies the Borrower of the existence of such conflict, of another international and another Colombian counsel of any such affected entity), in each case, arising out of, in connection with, or as a result of: (i) the execution or delivery of the Loan Documents, any demand for payment, other presentation or request under the Loan Documents, (ii) any Loans or the use or proposed use of the proceeds therefrom, (iii) any payment or other action taken or omitted to be taken in connection with the Loan Documents, (iv) any actual environmental liability related in any way to the Borrower or any of its Material Subsidiaries, and (v) any actual or prospective claim, litigation, investigation or proceeding relating to any of the foregoing whether based on contract, tort or any other theory, whether brought by a third party or by the Bor-

rower or any of its Subsidiaries and regardless of whether any Indemnitee is a party thereto; provided that such indemnity shall not, as to any Indemnitee, be available to the extent that such losses, claims, damages, liabilities or related expense (x) are determined by a court of competent jurisdiction by final and non-appealable judgment to have resulted from the gross negligence or willful misconduct of such Indemnitee or (y) result from a claim brought by the Borrower against an Indemnitee for breach in bad faith of such Indemnitee's obligations under any Loan Document, if the Borrower has obtained a final non-appealable judgment in its favor on such claims as determined by a court of competent jurisdiction. This Section 11.08 shall not apply with respect to Taxes (which shall be covered by Section 4.01) other than any Taxes that represent losses, claims, damages, liabilities and related expenses arising from any non-Tax claim.

- (c) To the extent that the Borrower fails to pay any amount required to be paid by it to the Administrative Agent under clause (a) of this Section, each Lender severally agrees to pay to the Administrative Agent such Lender's Proportionate Share (determined as of the time that the applicable unreimbursed expense or indemnity payment is sought) of such unpaid amount; *provided* that the unreimbursed expense or indemnified loss, claim, damage, liability or related expense, as the case may be, was incurred by or asserted against the Administrative Agent.
- $\mbox{\rm (d)}$ $\mbox{\rm All}$ amounts due under this Section shall be payable not later than thirty (30) days after written demand therefor.
- (e) The agreements in this Section shall survive the resignation of the Administrative Agent, the replacement of any Lender, the termination of the Commitments and the repayment, satisfaction or discharge of all the other obligations under this Agreement.

11.09 Judgment Currency

The Borrower's obligations under (a) the Loan Documents to make payments in Dollars shall not be discharged or satisfied by any tender or recovery pursuant to any judgment expressed in or converted into any currency other than Dollars, except to the extent that such tender or recovery results in the receipt by the Administrative Agent or any Lender of the full amount of Dollars expressed to be payable to Administrative Agent or such Lender under the Loan Documents. If for the purpose of obtaining or enforcing judgment against the Borrower in any court or in any jurisdiction outside of Colombia, it becomes necessary to convert into or from any currency other than Dollars (such other currency being hereinafter referred to as the "Judgment Currency") an amount due in Dollars, the conversion shall be made at the rate of exchange determined, in each case, as of the day on which the judgment is given (such Business Day being hereinafter referred to as the "Judgment Currency Conversion Date"). If, for the purpose of obtaining or enforcing judgment against the Borrower in any court in Colombia, it becomes necessary to convert into or from any Judgment Currency an amount due in Dollars, the conversion shall be made at the Colombian market representative rate certified by the Colombian Superintendence of Finance for the date of payment.

If there is a change in the rate of exchange prevailing between the Judgment Currency Conversion Date and the date of actual payment of the amount due by the Borrower, the Borrower covenants to pay, or cause to be paid, such additional amounts, if any (but in any event not a lesser amount), as may be necessary to ensure that the amount paid by the Borrower in the Judgment Currency, when converted at the rate of exchange prevailing on the date of payment to the Administrative Agent or such Lender by the Borrower, shall produce the amount of Dollars which could have been purchased with the amount of Judgment Currency stipulated in the judgment or judicial award at the rate of exchange prevailing on the Judgment Currency Conversion Date. Any such amount of Dollars not discharged by such Borrower payment of Judgment Currency shall continue to be due as an outstanding and unpaid obligation under this Agreement and shall accrue interest at the rate then applicable to the Loan in accordance with Section 2.07 until paid in full. If there is a change in the rate of exchange prevailing between the Judgment Currency Conversion Date and the date of actual payment by the Borrower of the amount due to the Administrative Agent or such Lender that results in the Borrower paying an amount in excess of that necessary to discharge or satisfy any judgment, the Administrative Agent or such Lender shall transfer or cause to be transferred to the Borrower the amount of such excess (net of any Taxes and reasonable and customary costs incurred in connection therewith).

(c) For purposes of applying the rate of exchange under this <u>Section 11.09</u>, the amount of Judgment Currency converted shall include any premium and costs payable in connection with the purchase of Dollars.

11.10 English Language

All documents (except the Notes) to be delivered by any party hereto pursuant to the terms hereof shall be in the English language or, if originally written in another language, shall be accompanied by an accurate English translation, upon which the other parties hereto shall have the right to rely for all purposes under any Loan Document. If any Loan Document (or any provision of any Loan Document) other than the Notes is translated in any language other than English, the version which is in English shall prevail in case of any discrepancy with any version in any other language.

11.11 Severability

To the extent permitted by Applicable Law, the illegality or unenforceability of any provision of this Agreement shall not in any way affect or impair the legality or enforceability of the remaining provisions of this Agreement.

11.12 Waiver of Jury Trial

EACH PARTY HERETO HEREBY IRREVOCABLY WAIVES TO THE FULLEST EXTENT PERMITTED BY

APPLICABLE LAW ALL RIGHT EACH MAY HAVE TO A TRIAL BY JURY IN ANY ACTION, SUIT, PROCEEDING OR COUNTERCLAIM DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, THE TRANSACTIONS CONTEMPLATED HEREBY OR ANY OTHER LOAN DOCUMENT TO WHICH IT IS A PARTY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY).

11.13 Captions

The table of contents and captions and section headings appearing herein are included solely for convenience of reference and are not intended to affect the interpretation of any provision of this Agreement.

11.14 Damages Waiver

To the fullest extent permitted by Applicable Law, no party shall assert, and each hereby waives, any claim, on any theory of liability, for indirect, special, punitive, consequential or exemplary damages (as opposed to direct or actual damages) arising out of, in connection with, or as a result of, this Agreement, any other Loan Document or any agreement or instrument contemplated hereby or thereby, the transactions contemplated hereby or thereby, any Loan or the use of the proceeds thereof.

11.15 Confidentiality

Each of the Administrative Agent and each of the Lenders agrees to maintain the confidentiality of the Confidential Information, except that Confidential Information may be disclosed (a) to its and its Affiliates' directors, officers, employees and agents, trustees, including accountants, legal counsel and other advisors (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Confidential Information and instructed to keep such Confidential Information confidential), (b) to the extent requested by any Governmental Authority purporting to have jurisdiction over it (including any self-regulatory authority, such as the National Association of Insurance Commissioners), (c) to the extent required by Applicable Laws or regulations or by any subpoena or similar legal process, (d) to any other party to this Agreement, (e) in connection with the exercise of any remedies hereunder or any suit, action or proceeding relating to this Agreement or the enforcement of rights hereunder, (f) subject to an agreement containing provisions substantially the same as those of this Section, to (i) any assignee of or participant in, or any prospective assignee of or participant in, any of its rights or obligations under this Agreement or (ii) any actual or prospective counterparty (or its advisors) to any swap or derivative transaction relating to the Borrower or its obligations, (g) with the consent of the Borrower or (h) to the extent such Confidential Information (i) becomes publicly available other than as a result of a breach of this Section, (ii) becomes available to the Administrative Agent or any Lender on a non-confidential basis from a source other than the Borrower, or (iii) is not, in the reasonable belief of the Administrative Agent including any such information in respect of which the Administrative Agent or Lender reasonably believes that it is not bound by any confidential obligation (but in any event excluding any information furnished by the Borrower that is designated as confidential in writing). Any Person required to maintain the confidentiality of Confidential Information as provided in this Section shall be considered to have complied with its obligation to do so if such Person has exercised the same degree of care to maintain the confidentiality of such Confidential Information as such Person would accord to its own confidential information. The Borrower hereby acknowledges that any Confidential Information permitted to be shared hereunder pursuant to clauses (a), (d) or (f) may be made available to the Persons permitted hereby, by posting such Confidential Information on IntraLinks, Debtdomain, Syndtrak, Clear-Par, or another similar electronic system.

Each of the Administrative Agent and each Lender acknowledges that (a) the Confidential Information may include material non-public information concerning the Borrower or any of its Subsidiaries and (b) it has developed compliance procedures regarding the use of material non-public information and for the handling of such material non-public information in accordance with Applicable Law, including United States Federal and state securities laws.

11.16 Survival

The expiration or termination of this Agreement does not terminate or affect any obligation hereunder that either expressly or by its nature survives the expiration or termination of this Agreement. Such obligations include, but are not limited to, those described in Sections 2.09, 2.10, 4.01, 9, 10.06, 11.08, 11.09 and 11.14, which shall survive expiration or termination of this Agreement; provided, however, that no change in Applicable Law after the expiration or termination of this Agreement shall increase the Borrower's obligations with respect to amounts paid prior to such expiration or termination.

11.17 No Fiduciary Duty

In connection with all aspects of each transaction contemplated hereby (including in connection with any amendment, waiver or other modification hereof or of any other Loan Document), the Borrower acknowledges and agrees that: (a) (i) the arranging and other services regarding this Agreement provided by the Administrative Agent and the Lenders are arm's-length commercial transactions between the Borrower and its Affiliates, on the one hand, and the Administrative Agent, and the Lenders, on the other hand, (ii) the Borrower has consulted its own legal, accounting, regulatory and tax advisors to the extent it has deemed appropriate, and (iii) the Borrower is capable of evaluating, and understands and accepts, the terms, risks and conditions of the transactions contemplated hereby and by the other Loan Documents; (b) (i) the Administrative Agent and each Lender is and has been acting solely as a principal and, except as expressly agreed in writing by the relevant parties, has not been, is not, and will not be acting as an advisor, agent or fiduciary for the Borrower or any of its Affiliates, or any other Person and (ii) neither the Administrative Agent nor any Lender has any obligation to the Borrower or any of its Affiliates with respect to the transactions contemplated

hereby except those obligations expressly set forth herein and in the other Loan Documents; and (c) the Administrative Agent and the Lenders and their respective Affiliates may be engaged in a broad range of transactions that involve interests that differ from those of the Borrower and its Affiliates, and neither the Administrative Agent nor any Lender has any obligation to disclose any of such interests to the Borrower or its Affiliates. To the fullest extent permitted by law, the Borrower hereby waives and releases any claims that it may have against the Administrative Agent or any Lender with respect to any breach or alleged breach of agency or fiduciary duty in connection with any aspect of any transaction contemplated hereby.

11.18 PATRIOT Act

Each Lender that is subject to the requirements of the PATRIOT Act hereby notifies the Borrower that pursuant to the requirements of the PATRIOT Act, it is required to obtain, verify and record information that identifies the Borrower, which information includes the name and address of the Borrower and other information that will allow such Lender to identify the Borrower in accordance with the PATRIOT Act.

11.19 Exequatur

A judgment obtained in a New York Court arising out of or relating to this Agreement, or the transactions contemplated thereby, will be enforced against the Borrower in the courts of Colombia; *provided* that such judgment has previously obtained an Exequatur, which is regulated by Colombian law.

11.20 <u>International Banking Facilities</u>

Certain Lenders intend to book extensions of credit hereunder on the books and records of their respective international banking facilities, which will constitute an extension of credit within the meaning of Section 204.8(a)(3) of Regulation D of the Board ("Regulation D"). Accordingly, pursuant to Regulation D, the Borrower acknowledges that such Lenders have notified that it is the policy of the Board that extensions of credit by international banking facilities may be used only to finance operations outside of the United States of borrowers or their affiliates which are located outside the United States. The Borrower agrees to limit its use of the net proceeds of the Loans to the financing of its operations outside the United States.

[Remainder of Page Intentionally Left Blank]

IN WITNESS WHEREOF, each of the parties hereto has caused this Agreement to be duly executed and delivered as of the date first above written.

Title:

Borrow	EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P., as er
Name: Title:	By:
as Adm	THE BANK OF TOKYO-MITSUBISHI UFJ, LTD., inistrative Agent
Name: Title:	Ву:
Name: Title:	By:
	HSBC BANK USA, N.A., as Lender
Name: Title:	By:
	BANK OF AMERICA, N.A., as Lender
Name: Title:	By:
Name: Title:	By:
as Lend	THE BANK OF TOKYO-MITSUBISHI UFJ, LTD., der
Name: Title:	By:
Name: Title:	By:
NEW Y	SUMITOMO MITSUI BANKING CORPORATION, ORK BRANCH, as Lender
Name: Title:	By:
Name:	Ву:

	CITIBANK, N.A., ACTING THROUGH ITS INTER-		By:
NOTTAN	NAL BANKING FACILITY, as Lender	Name:	•
		Title:	
	By:		
Name:			MIZUHO BANK, LTD., as Lender
Title:			_
	Dv:	Name:	Ву:
Name:	By:	Title:	
Title:			
			Ву:
	JP MORGAN CHASE BANK, N.A., as Lender	Name:	,
		Title:	
	By:		
Name:			
Title:			

SCHEDULE 1.01-I

COMMITMENTS

Lender	Commitment
HSBC SECURITIES (USA) INC.;	\$175.000.000
BANK OF AMERICA, N.A.;	\$175.000.000
THE BANK OF TOKYO-MITSUBISHI UFJ, LTD.;	\$175.000.000
SUMITOMO MITSUI BANKING CORPORATION;	\$175.000.000
CITIGROUP GLOBAL MARKETS INC.;	\$100.000.000
JP MORGAN SECURITIES LLC.; and	\$100.000.000
MIZUHO BANK, LTD.	\$100.000.000
Total	
	\$1.000.000
Figures in United States Dollars	

SCHEDULE 1.01-II EXISTING LIENS

Name of Lender	Nature of the lien	Specifications of the Lien
Inter-American Development Bank	Counter guarantee in connection with certain IADB loans	Certain loans granted to the Borrower by the Inter-American Development Bank are guaranteed by the Republic of Colombia. Pursuant to Colombian law, whenever the Republic of Colombia issues a guarantee to secure the obligations of a government-owned company in respect of third parties, government-owned companies are required to issue a counter-guarantee in favor of the Republic of Colombia. Under the counter-guarantee granted by the Borrower under the loans entered with the IADB, the Borrower agreed to grant a lien on a portion of its revenues, resulting from the rendering of public utility services equal to 120% of the debt corresponding to a sixmonth period.

SCHEDULE 6.01(b)

GOVERNMENTAL APPROVALS

- (a) Approval from the Ministry of Finance through Resolution No. 1348 of May 17, 2012; No. 3071 of September 2, 2014 and No. 4783 of December 22, 2015
- (b) Authorization to execute the Agreement issued by the Ministry of Finance given by means of a Resolution, prior approval of the minutes of the Loan Documents by the General Directorate of Public Credit (*Dirección General de Crédito Público*) (Article 6 of Law 781 of 2002 and article 2.2.1.2.1.4 of Decree 1068 of 2015) as evidenced by the publication of the Resolution in the Official Diary, which will be satisfied by the order of the

General Directorate of Public Credit and National Treasury of the Ministry of Finance pursuant to article 18 of Law 185 of 1995.

- (c) Registration of the Agreement before the Colombian Central Bank by means of Form No. 6 "Información de Endeudamiento Externo otorgado a Residentes".
- (d) The favorable opinion from the National Planning Department (*Departamento Nacional de Planeación*), prior to Resolutions No. 1348 and 3071.

EXHIBIT A

FORM OF ASSIGNMENT AND ASSUMPTION AGREEMENT

ASSIGNMENT AND ASSUMPTION

This Assignment and Assumption (the "Assignment and Assumption") is dated as of the Effective Date set forth below and is entered into by and between [Insert name of Assignor] (the "Assignor") and [Insert name of Assignee] (the "Assignee"). Capitalized terms used but not defined herein shall have the meanings given to them in the Credit Agreement identified below (as amended, the "Credit Agreement"), receipt of a copy of which is hereby acknowledged by the Assignee. The Standard Terms and Conditions set forth in Exhibit 1 attached hereto are hereby agreed to and incorporated herein by reference and made a part of this Assignment and Assumption as if set forth herein in full.

For an agreed consideration, the Assignor hereby irrevocably sells and assigns to the Assignee, and the Assignee hereby irrevocably purchases and assumes from the Assignor, subject to and in accordance with the Standard Terms and Conditions and the Credit Agreement, as of the Effective Date inserted by the Administrative Agent as contemplated below (i) all of the Assignor's rights and obligations in its capacity as a Lender under the Credit Agreement and any other documents or instruments delivered pursuant thereto to the extent related to the amount and percentage interest identified below of all of such outstand-

ing rights and obligations of the Assignor under the respective facilities identified below and (ii) to the extent permitted to be assigned under Applicable Law, all claims, suits, causes of action and any other right of the Assignor (in its capacity as a Lender) against any Person, whether known or unknown, arising under or in connection with the Credit Agreement, any other documents or instruments delivered pursuant thereto or the loan transactions governed thereby or in any way based on or related to any of the foregoing, including, but not limited to, contract claims, tort claims, malpractice claims, statutory claims and all other claims at law or in equity related to the rights and obligations sold and assigned pursuant to clause (i) above (the rights and obligations sold and assigned pursuant to clauses (i) and (ii) above being referred to herein collectively as, the "Assigned Interest"). Such sale and assignment is without recourse to the Assignor and, except as expressly provided in this Assignment and Assumption, without representation or warranty by the Assignor.

1.	Assignor:	
2.	Assignee:	- <u></u>
		[and is an Affiliate/approved fund of

[identify Lender]¹]

Borrower: EMPRESAS PÚBLICAS DE

MEDELLÍN E.S.P.

Administrative Agent: [●]

5. Credit Agreement: The Credit Agreement dated as of [●], 2015, among EMPRESAS PÚBLICAS

DE MEDELLÍN E.S.P., as
Borrower, certain Lenders
parties thereto, and THE BANK
OF TOKYO-MITSUBISHI
UFJ, LTD., as Administrative

Agent.

Assigned Interest:

¹ Select as applicable.

Facility Assigned	Aggregate Amount of Commitment/Loans for all Lenders*		Percentage Assigned of Commitment/Loans ²	
Commitment	\$	\$	%	
Loans \$		\$	%	
	\$	\$	%	

7.	Trade Date:	

[`]Amount to be adjusted by the counterparties to take into account any payment or prepayments made between the Trade Date and the Effective Date.

² Set forth, to at least 9 decimals, as a percentage of the Commitment/Loans of all Lenders thereunder.

Effective Date:	, 20_	[TO
BE INSERTED BY ADMINISTRATIVE AGE	NTAND	WHICH
SHALL BE THE EFFECTIVE DATE OF I	RECORD	ATION
OF TRANSFER IN THE REGISTER THER	REFOR.]	

The terms set forth in this Assignment and Assumption are hereby agreed to:

Exhibit 1

Title:

Standard Terms and Conditions for Assignment and Assumption

- Representations and Warranties.
- 1.1 <u>Assignor</u>. The Assignor (a) represents and warrants that (i) it is the legal and beneficial owner of the Assigned Interest, (ii) the Assigned Interest is free and clear of any lien, encumbrance or other adverse claim and (iii) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and Assumption and to consummate the transactions contemplated hereby; and (b) assumes no responsibility with respect to (i) any statements, warranties or representations made in or in connection with the Credit Agreement or any other Credit Document, (ii) the execution, legality, validity, enforceability, genuineness, sufficiency or value of the Loan Documents, (iii) the financial condition of the Borrower, any

of its Subsidiaries or Affiliates or any other Person obligated in respect of any Loan Document or (iv) the performance or observance by the Borrower, any of its Subsidiaries or Affiliates or any other Person of any of their respective obligations under any Loan Document.

- Assignee. The Assignee (a) represents and warrants that (i) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and Assumption and to consummate the transactions contemplated hereby and to become a Lender under the Credit Agreement, (ii) it meets all requirements of an assignee under the Credit Agreement (subject to receipt of such consents as may be required under the Credit Agreement), (iii) from and after the Effective Date, it shall be bound by the provisions of the Credit Agreement as a Lender thereunder and, to the extent of the Assigned Interest, shall have the obligations of a Lender thereunder, and (iv) it has received a copy of the Credit Agreement, together with copies of the most recent financial statements delivered pursuant to Section 7.01(b) thereof, as applicable, and such other documents and information as it has deemed appropriate to make its own credit analysis and decision to enter into this Assignment and Assumption and to purchase the Assigned Interest on the basis of which it has made such analysis and decision independently and without reliance on the Administrative Agent or any other Lender; and (b) agrees that (i) it will, independently and without reliance on the Administrative Agent, the Assignor or any other Lender, and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under the Loan Documents, and (ii) it will perform in accordance with their terms all of the obligations which by the terms of the Loan Documents are required to be performed by it as a Lender.
- 2. <u>Payments</u>. From and after the Effective Date, the Administrative Agent shall make all payments in respect of the Assigned Interest (including payments of principal, interest, fees and other amounts) to the Assignor for amounts which have accrued to but excluding the Effective Date and to the Assignee for amounts which have accrued from and after the Effective Date.
- 3. General Provisions. This Assignment and Assumption shall be binding upon, and inure to the benefit of, the parties hereto and their respective successors and assigns. This Assignment and Assumption may be executed in any number of counterparts, which together shall constitute one instrument. Delivery of an executed counterpart of a signature page of this Assignment and Assumption by telecopy shall be effective as delivery of a manually executed counterpart of this Assignment and Assumption. This Assignment and Assumption shall be governed by, and construed in accordance with, the law of the State of New York.

³ To be included to the extent consent of the Borrower is required pursuant to Section 11.03(b)

EXHIBIT B

FORM OF INSTRUCTIONS LETTER

INSTRUCTIONS LETTER

Messrs.

Medellín, [Date]

[•] and / or its assignee and successors.

Instructions to complete the Promissory Reference: Note with Blank Spaces No. [●].

Dear Sirs:

EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P., a public utility services company organized and existing under the laws of the República de Colombia, identified with the tax identification number NIT 890904996, with full capacity to issue securities, and domiciled in Medellín (hereinafter the "Borrower"), represented herein by the undersigned JUAN ESTEBAN CALLE RESTREPO, of legal age, domiciled in Medellín, Colombia, identified with citizenship card number 70.566.038, issued on Envigado acting as Chief Executive Officer, duly authorized by the Board of Directors under Minute number 1539 of September 6, 2011 to execute this document, in accordance with article 622 of the Colombian Commercial Code, hereby imparts instructions and irrevocable and permanent powers to [●] and / or its successors or assignees (hereinafter the "Lender") (in English the "Lender"), a foreign financial institution duly organized and existing under the laws of its country, acting in its capacity as Lender under the Loan Agreement (as defined below), to fill each and every one of the of this Instructions Letter (hereinafter the "Promissory Note"), in the event the obligations set forth under the Promissory Note become due and in accordance to the terms set forth below:

1. BACKGROUND AND CAUSE.

On [•] [•] 20 [•], the Borrower and the Lender executed the Loan Agreement (hereinafter the "Loan Agreement") borrow [•] DOLLARS OF THE UNITED STATES OF AMERICA (USD\$[●]) (hereinafter the "Term Loan"), with a maximum deadline for payment of [●], counted from [●] as set forth in the Loan Agreement.

As a condition precedent to grant and disburse the Loan, it was agreed that EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P. should have issued and delivered the Promissory Note whereby the Term Loan is evidenced.

AUTHORIZATION TO FILL THE PROMISSORY NOTE

Should (i) EMPRESAS PÚBLICAS DE MEDELLÍN and payable under the Loan, including principal repayments, interest payments or other amounts due under the Loan on the date on which such payment is due under the Loan Agreement or (ii) any other Events of Default (in English "Events of Default") described therein leading to the acceleration of the amounts due and payable occur, and such amounts are not paid; [•] or its assignees, endorsees or successors may fill in the blank spaces of the Promissory Note identified above, without prior notice, presentation, protest, notice of default or any other notice, notification or additional requirement of any nature, in accordance with this Instructions Letter.

CARTA DE INSTRUCCIONES

Medellín, [Fecha]

Señores.

[•] y/o sus cesionario y causahabientes

Instrucciones para diligenciar el Pagaré con Referencia: Espacios en Blanco No. [.].

Estimados Señores:

EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P., empresa de servicios públicos domiciliarios, organizada y existente bajo las leyes de la República de Colombia, identificada con NIT 890904996 y domiciliada en Medellín (en adelante el "Deudor"), representada en este acto por el suscrito JUAN ESTEBAN CALLE RESTREPO, mayor de edad, domiciliado en Medellín, Colombia, identificado con la cédula de ciudadanía número 70.566.038, expedida en Envigado, actuando en calidad de Representante Legal, debidamente autorizado por la Junta Directiva mediante Acta No. 1539 del 6 de septiembre de 2011 para suscribir el presente documento, conforme al artículo 622 del Código de Comercio Colombiano, por medio de la presente carta imparte instrucciones y facultades irrevocables y permanentes a [•], con domicilio en [•], y/o sus cesionarios o causahabientes (en adelante el "Acreedor") (en inglés the "Lender"), una entidad financiera extranjera debidamente constituida y actualmente existente bajo las leyes de su país, en su carácter de Acreedor bajo el Contrato de Crédito (tal y como se define más adelante), para blank spaces left in the Promissory Note identified at the heading llenar todos y cada uno de los espacios en blanco dejados en el Pagaré identificado en el encabezamiento de esta Carta de Instrucciones (en adelante, el "Pagaré"), cuando se hagan exigibles las obligaciones contenidas en dicho Pagaré en los términos que se indican a continuación:

1. ANTECEDENTES Y CAUSA.

El día [●] de [●] de 20[●] el Acreedor y el Deudor suscribieron el Contrato de Crédito (en adelante, el "Contrato de Crédiwhereby, the Lender agreed to lend, and the Borrower agreed to |to") para entregar a título de mutuo o préstamo la suma de [•] DE DÓLARES DE LOS ESTADOS UNIDOS DE AMÉRICA (US\$[●])(en adelante el "Préstamo"), con un plazo máximo para su pago de [●] , contados a partir de [•] tal y como se establece en el Contrato de Crédito.

> Como condición para el otorgamiento y desembolso del Préstamo, se estableció que EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P. debía haber otorgado y entregado el Pagaré mediante el cual se evidencia el Préstamo.

AUTORIZACIÓN PARA LLENAR EL PAGARÉ.

En el evento en que (i) EMPRESAS PÚBLICAS DE ME-E.S.P. fail in whole or in part to pay any outstanding sum due DELLÍN E.S.P. incumpla en todo o en parte su obligación de pago de cualquiera de las cuotas de capital, intereses o demás sumas debidas bajo el Préstamo en las fechas previstas de conformidad con el Contrato de Crédito u (ii) ocurra cualquier otro de los Eventos de Incumplimiento (en inglés "Events of Default") allí descritos que conlleven al vencimiento anticipado de las sumas adeudadas y éstas no fueren pagadas; [●] o sus cesionarios, endosatarios o causahabientes, podrá llenar los espacios en blanco del Pagaré arriba identificado, sin necesidad de previo aviso, presentación, protesto, constitución en mora, cualquier otro tipo de aviso, notificación o requisito adicional de cualquier naturaleza de conformidad con esta Carta de Instrucciones

3. VALUE OR AMOUNT OF THE PROMISSORY NOTE.

EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P. expressly and irrevocably authorizes that [•] and / or its successors or assignees fill in the blank spaces of the Promissory Note mechanically (with a typewriter) or manually in accordance with the instructions below:

- 3.1 The blank space included under Section 1.1. of the Promissory Note shall be completed with the outstanding amounts of principal (in letters and numbers) to the date in which the Promissory Note is completed and that are due on such date by EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P. to [•] under the Loan (in English, the "Loan"), including amounts due and outstanding due to maturity or any acceleration in accordance to the Loan Agreement.
- 3.2 The blank spaces included in Section 1.2. of the Promissory Note shall be completed as follows: blank spaces identified with number (1) with the outstanding amounts of interest (in letters and numbers) calculated over the principal amount set forth under section 1.1 of the Promissory Note in accordance to the provisions set forth under the Loan Agreement, blank spaces identified with number (2) with the date as of which such interests are accrued and blank spaces identified with number (3) with the date up to which such interests are accrued. As set forth under the Promissory Note, interests referred to under section 1.2 of the Promissory Note shall bear default interests, it being understood that any outstanding interest obligations that remain unpaid for more than one (1) year may (upon the satisfaction of the rules set forth in Article 886 of the Commercial Code) be subject to interest on interest under Colombian law as currently in effect.
- 3.3 The blank spaces included in Section 1.3. of the Promissory Note shall be completed as follows: blank spaces identified with number (1) with the outstanding amounts of the default interests (in letters and numbers) calculated over the principal amounts, interest payments and any other amounts due and outstanding, either due to maturity, acceleration or otherwise, in accordance with the provisions of the Loan Agreement, blank spaces identified with number (2) with the date as of which such interests are accrued and blank spaces identified with number (3) with the date up to which such interests are accrued.

4. MATURITY.

The blank space included under Section 3 of the Promissory Note corresponding to the maturity date, shall be completed including the date in which the Promissory Note is filled or completed.

5. DATE AND CITY OF ISSUANCE.

[●] is entitled to include as date and city of issuance the date and city in which it decides to fill or complete the Promissory Note.

6. SUCCESSORS AND ASSIGNEES.

Blanks spaces can be completed by the successors or assignees of the Loan in accordance with Section [•] of the Loan Agreement.

The Promissory Note completed in accordance to the provisions set forth herein, shall be directly enforceable without any further requirements.

Sincerely,

The Borrower,

EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P.

Name: JUAN Identification:

Capacity:

ESTEBAN CALLE C.C.

Chief Executive Officer

RESTREPO 70.566.038 3. VALOR O IMPORTE DEL TÍTULO.

EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P. autoriza expresa e irrevocablemente que [•] y/o sus sucesores o sus causa-habientes llenen los espacios en blanco del Pagaré de manera mecánica (a máquina) o a mano de conformidad con las instrucciones que se indican a continuación:

- 3.1 El espacio en blanco identificado con el numeral 1.1. del Pagaré se llenará con las sumas de capital (en letras y números) que a la fecha en que el Pagaré sea llenado y que EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P. adeude, en tal fecha, a favor del [•] por concepto del Préstamo (en inglés, el "Loan"), incluyendo las sumas adeudadas y pendientes de pago por virtud de vencimiento del plazo o cualquier aceleración de conformidad con el Contrato de Crédito.
- 3.2 Los espacios en blanco incluidos en el numeral 1.2. del Pagaré se llenarán así: los espacios identificados con el numeral (1) con las sumas devengadas por concepto de los intereses remuneratorios (en letras y números) calculados sobre el capital indicado en el numeral 1.1. del Pagaré de conformidad con lo previsto en el Contrato de Crédito, los espacios identificados con el numeral (2) con la fecha a partir de la cual se devengan dichos intereses y los espacios identificados con el numeral (3) con la fecha hasta la cual se devengan dichos intereses. Tal como se indica en el Pagaré, los intereses remuneratorios indicados en el punto 1.2 del mismo causarán intereses de mora. tomando en cuenta que cualquier obligación no satisfecha con respecto al pago de intereses debidos por más de un 1 año (sujeto a la satisfacción de las normas previstas en el artículo 886 del Código de Comercio) estarán sujetas al pago de interés sobre interés bajo la ley colombiana vigente.
- 3.3 Los espacios en blanco incluidos en el numeral 1.3. del Pagaré se llenarán así: los espacios identificados con el numeral (1) con las sumas devengadas por concepto de los intereses moratorios (en letras y números) calculados sobre las cuotas de capital o de intereses o cualquier otro monto vencido y pendiente de pago, bien sea por cumplimiento del plazo, por aceleración o de cualquier otra forma, de conformidad con lo previsto en el Contrato de Crédito, los espacios identificados con el numeral (2) con la fecha a partir de la cual se devengan dichos intereses y los espacios identificados con el numeral (3) con la fecha hasta la cual se devengan dichos intereses.

4. VENCIMIENTO.

El espacio en blanco en el numeral 3 del Pagaré correspondiente a la fecha de vencimiento, se llenará con la fecha del día en que se llene o complete el Pagaré.

5. FECHA Y CIUDAD DE EMISIÓN.

[●] está facultado para colocar como fecha y la ciudad de emisión el día y la ciudad en que se decida llenar o completar el Pagaré.

6. SUCESORES O CAUSAHABIENTES.

Los espacios en blanco pueden ser llenados por los sucesores o causahabientes del Préstamo de conformidad con la Sección [ullet] del Contrato de Crédito.

El Pagaré así llenado será exigible inmediatamente y prestará mérito ejecutivo sin más requisitos.

Atentamente.

El Deudor,

EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P.

Por: _____Nombre: JUAN ESTEBAN CALLE RESTREPO

Identificación: C.C. 70.566.038 Cargo: Representante Legal

EXHIBIT C

FORM OF LOAN REQUEST

LOAN REQUEST

 $[[\bullet][\bullet], [\bullet]]$

[•] [•]

Attn: [●]
Phone: +1 [●]
Facsimile: +1 [●]

Email: [•]

Ladies and Gentlemen:

Reference is made to the Credit Agreement dated as of [•], 2015 (the "Agreement") by and among EMPRE-SAS PÜBLICAS DE MEDELLÍN E.S.P. (the "Borrower") THE BANK OF TOKYO-MITSUBISHI UFJ, LTD., as Administrative Agent and the Lenders party thereto. Unless otherwise defined herein, terms defined in the Agreement shall have the same meaning in this Loan Request.

The Borrower hereby requests the Lenders to make Loans to the Borrower in the principal amount of U.S. $\{\bullet\}$ (the "Loan Amount") on $[\{\bullet\}]$, $[\bullet]$, $[\bullet]$] (the "Disbursement Date") under the Agreement, as follows:

LOAN AMOUNT: [•]

DISBURSEMENT DATE: [[●][●], [●]]

The Borrower hereby instructs the Lenders to disburse the Loan to account No. [●] with [●], ABA No. [●], Attn.: EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P.

The Borrower hereby certifies that the following statements are true on the date hereof, and will be true on the Disbursement Date:

- (a) the representations and warranties made by the Borrower in Section 6 of the Agreement are true and correct in all material respects both before and immediately after giving effect to the Loans and the application of the proceeds thereof; provided that representations and warranties qualified as to materiality are true and correct on and as of such date; and
- (b) no Event of Default has occurred and is continuing both before and immediately after giving effect to the Loans and the application of the proceeds thereof.

IN WITNESS WHEREOF, the undersigned have executed this Loan Request and delivered the same as of the date first written above.

EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P.

By:		
Name:		
Title:		

EXHIBIT D

FORM OF PROMISSORY NOTE

FORM OF PROMISSORY NOTE⁴

Blank Promissory Note No. [•]

EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P. a public utility services Company organized and existing under the laws of the República de Colombia , identified with the tax identification number NIT 890904996, with full capacity to issue securities, and domiciled in Medellín (hereinafter the "Borrower"), represented herein by the undersigned JUAN ESTEBAN CALLE RESTREPO, of legal age, domiciled in Medellín, Colombia, Colombia, identified with citizenship card number 70.566.038, issued on Envigado acting as Chief Executive Officer, duly authorized by the Board of Directors under Minute number 1539 of September 6, 2011 to execute this document, hereby promises to pay unconditionally and irrevocably, to the order of [•], domiciled in [•] (hereinafter the "Lender"), or its registered assignee in due course, including the endorsee, the amounts indicated below:

FORMATO DE PAGARÉ

Pagaré en Blanco No. [•]

EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P., empresa de servicios públicos domiciliarios, organizada y existente bajo las leyes de la República de Colombia, identificada con NIT 890904996 y domiciliada en Medellín (en adelante el "Deudor"), representada en este acto por el suscrito JUAN ESTEBAN CALLE RESTREPO, mayor de edad, domiciliado en Medellín, Colombia, identificado con la cédula de ciudadanía número 70.566.038, expedida en Envigado, actuando en calidad de Representante Legal, debidamente autorizado por la Junta Directiva mediante Acta No. 1539 del 6 de septiembre de 2011 para suscribir el presente documento, declara que pagará de manera incondicional e irrevocable, a la orden del [•], con domicilio en [•] (en adelante el "Acreedor"), o a cualquier cesionario legítimo, incluyendo el endosatario, las sumas que se indican a continuación:

⁴ English translation is for convenience. The Spanish version will be executed and delivered on the Agreement Date.

	Amounts in 0.5. Dollars	1. Sullias en Dolales de los Estados Utildos de America.
(US	The amount of United States dollars SD\$) corresponding to the principal ount disbursed and due under the Loan.	1.1 La suma de dólares de los Estados Unidos de América (US\$) por concepto de capital desembolsado y adeudado bajo el Préstamo.
est	The amount of (1) United States lars (USD\$) corresponding to the inters accrued under the loan, as of the (2) day of, and until the day of (3) of 20_; and	1.2 La suma de (1) dólares de los Estados Unidos de América (US\$) por concepto de intereses remuneratorios devengados bajo el préstamo, a partir del día (2) de de 20 y hasta el día (3) de de 20 ; y
est	The amount of (1) United States dols (USD\$) corresponding to default inters accrued and unpaid, as of the (2) day of, and until the day of (3) of 20	1.3 La suma de (1) dólares de los Estados Unidos de América (US\$) por concepto de intereses de mora causados y pendientes de pago, a partir del día (2) de de 20 y hasta el día (3) de de 20
1.	The amounts referred to under section 1 above shall bear interest as follows:	 Las sumas indicadas en el numeral 1 anterior devengarán intereses así:
bea loa	The amount referred to under section 1.1 above shall ar default interests at the interest rate applicable to the n plus two percent (2%) annually, subject to the maxim rate permitted under applicable law.	1.1 La suma indicada en el numeral 1.1 anterior devengará intereses de mora a la tasa de interés anual remuneratoria aplicable al préstamo adicionada en dos por ciento (2%) anual, sujeto a la tasa máxima de interés permitida por la ley aplicable.
1.2 bea 2 p or a E.S set der ma fac Co	The amount referred to under section 1.2 above shall ar interests at the interest rate applicable to the loan plus ercent (2%) annually, from the date of the judicial lawsuit agreement with EMPRESAS PÚBLICAS DE MEDELLÍN S.P. following maturity, subject to applicability of the rules forth in Article 886 of the Commercial Code, it being unstood that any outstanding interest obligations that rein unpaid for more than one (1) year may (upon the satistion of the rules set forth in Article 886 of the Commercial de) be subject to interest on interest under Colombian law currently in effect.	1.2 La suma indicada bajo el numeral 1.2 devengará intereses a la tasa de interés anual remuneratoria aplicable al préstamo adicionada en dos por ciento (2%) anual, desde la fecha de la demanda judicial o por acuerdo posterior al vencimiento con EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P., tomando en cuenta que cualquier obligación no satisfecha con respecto al pago de intereses debidos por más de un 1 año (sujeto a la satisfacción de las normas previstas en el artículo 886 del Código de Comercio) estarán sujetas al pago de interés sobre interés bajo la ley colombiana vigente.
2.	All payments under this Promissory Note shall be made by EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P. as Borrower, in the city of New York, United States of America, in immediately available funds, without any deduction, withholding or set-off (except as required by applicable law), on the following date:	 Todos los pagos bajo el presente Pagaré deben ser realizados por EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P. como Deudor en la ciudad de Nueva York, Estados Unidos de América, en fondos inmediatamente disponibles, libres de retenciones y sin deducción o compensación (salvo por lo previsto por ley aplicable) alguna en la siguiente fecha:
3.	The Borrower expressly agrees that the obligations set forth under this Promissory Note correspond to foreign exchange operations and therefore shall be paid in the agreed currency.	 El Deudor expresamente acepta que las obligaciones in- corporadas en este Pagaré corresponden a operaciones cambiarias y, en consecuencia, serán pagadas en la divisa acordada.
4.	The Borrower, as borrower, irrevocably waives the pre- sentation, protest, notice of default or any other notice, notification or additional requirement of any kind, for the collection of this Promissory Note.	 El Deudor, como deudor, renuncia irrevocablemente a la presentación, protesto, constitución en mora o cualquier otro tipo de aviso, notificación o requisito adicional de cual- quier naturaleza para el cobro de este Pagaré.
5.	The Borrower agrees to pay all the costs, fees and expenses incurred to the collection or enforcement of this Promissory Note.	 El Deudor, acepta que serán de su cargo los gastos y hon- orarios profesionales que se generen por la cobranza de este Pagaré.
6.	The Borrower agrees to pay taxes (including stamp taxes) that may be due under this Promissory Note, as provided in Section 4 of the Credit Agreement and subject to the limitations found therein.	6. El Deudor acepta que serán de su cargo todos los impues- tos que pueda causar el presente Pagaré, incluyendo, si resultare aplicable, el impuesto de timbre, conforme a lo previsto en la Sección 4 del Contrato de Crédito y a los términos y condiciones ahí previstos.
7.	The Borrower expressly agrees that in the event of extension, renewal or modification of its obligation under this Promissory Note, this security shall remain valid and in effect until the date agreed under such extension, renewal or modification.	7. El Deudor acepta expresamente que en caso de prórroga, novación o modificación de la obligación a su cargo con- tenida en este Pagaré, el presente Pagaré continuará vi- gente hasta la fecha pactada en dicha prórroga, novación o modificación.
8.	This Promissory Note shall be governed by and construed in accordance with the laws of the Republic of Colombia and the undersigned Borrower, EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P., hereby agrees that the laws that govern its creation are the laws of the Republic of Colombia, place where this Promissory Note has been executed by EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P.	8. El presente Pagaré se encuentra regido por, y será interpretado de conformidad con, las leyes de la República de Colombia y el suscrito Deudor, EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P., expresamente acuerda que las leyes que rigen su creación son las de las leyes de la República de Colombia, lugar donde ha sido firmado por EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P.
9.	This Promissory Note shall be executed in the competent court in the Republic of Colombia.	 El presente Pagaré podrá ser ejecutado ante los tribu- nales competentes en la República de Colombia.

10. This Promissory Note is executed in both the English and Spanish language, both versions of which shall bind the Borrower; provided, however, that in the case of doubt as to the proper interpretation or construction of this promissory note, the Spanish version shall pre- vail.	obligando ambas versiones al Deudor. En el evento en que exista duda acerca de la adecuada interpretación del
It is hereby understood that this Promissory Note was filled or completed day of year, in the city of, in accordance with the Instructions Letter attached herein.	Se deja constancia que este Pagaré se llenó o completó el día del mes de del año, en la ciudad de, de conformidad con la carta de instrucciones que se adjunta al mismo.
The Borrower,	El Deudor,
EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P.	EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P.
By: Name: JUAN ESTEBAN CALLE RESTREPO Identification: C.C. 70.566.038 Capacity: Chief Executive Officer	Por:

EXHIBITE

FORM OF OFFICER'S CERTIFICATE OFFICER'S CERTIFICATE

[•], 2015

The undersigned, [•] of EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P., a public utility services company organized and existing under the laws of the República de Colombia (the "Borrower"), hereby certifies that he is an Authorized Officer of the Borrower and hereby certifies on behalf of the Borrower that:

This Certificate is furnished pursuant to Section 5.01(b) of the Credit Agreement dated as of $[\bullet][\bullet]$, $[\bullet]$, (the "Agreement"), among the Borrower, The Bank of Tokyo-Mitsubishi UFJ, Ltd., as Administrative Agent and the Lenders party thereto. Unless otherwise defined herein, terms defined in the Agreement shall have the same meaning in this Certificate.

The undersigned named below (a) has been duly elected, is duly qualified as, and is an officer of the Borrower as of the Closing Date, holding the respective office below set opposite to his name, and the signature below set opposite to his name is his genuine signature, (b) is an Authorized Officer of the Borrower.

<u>Name</u>	<u>Officer</u>	<u>Signature</u>
Juan Esteban Calle Restrepo	Chief Executive Officer	

Attached hereto as Exhibit E-1 is a true and correct copy of the articles of incorporation, *estatutos sociales* or other applicable organizational documents of the Borrower as in effect on the date hereof together with all amendments thereto.

Attached hereto as Exhibit E-2 is a true and correct copy of a duly authorized secretary's certificate certifying that the Board of Directors of the Borrower duly adopted the resolutions at a meeting on $[\bullet]$, $[\bullet]$, at which a quorum was present and acting throughout. The resolutions have not been revoked, modified, amended or rescinded and are in full force and effect. Except as attached hereto as Exhibit E-2, no resolutions have been adopted by the Board of Directors of the Borrower which deal with the execution, delivery or performance of any of the Loan Documents to which it is a party.

Attached hereto as Exhibit E-3 is a true and correct copy of the Borrower's *Certificados de Existencia y Representación Legal*.

[Remainder of Page Intentionally Left Blank]

IN WITNESS WHEREOF, each of the undersigned has executed this Certificate this day of $[ullet]$, $[ullet]$.
By: Name: [●] Title: [●]

- I, the undersigned, *Secretary General* of the Borrower, do hereby certify that I am an Authorized Officer of the Borrower and hereby certify on behalf of the Borrower that:
- [ullet] is the duly elected and qualified [ullet] of the Borrower and his respective signature above is his genuine signature.

IN	WITNESS WHEREOF	, I have	hereunto	set	my
hand this _	day of [●], [●].				

By:		
Name: [●]		
	_	

Title: Secretary General

Exhibit E-1 to Officer's Certificate

Organizational Documents

[EPM to provide]

Exhibit E-2 to Officer's Certificate

Resolutions

[EPM to provide]

Exhibit E-3 to Officer's Certificate

Good Standing Certificate

[EPM to provide]

EXHIBIT F

FORM OF LEGAL OPINION OF MOSQUERA ABOGADOS

[•], 2015

To the Lenders that are parties to the Credit Agreement referred to below, and The Bank of Tokyo-Mitsubishi UFJ, LTD, as Administrative Agent for such Lenders (the "Administrative Agent")

Ladies and Gentlemen:

We have acted as Colombian counsel to Empresas Públicas de Medellín S.A. E.S.P. in connection with the transaction contemplated by the credit agreement dated as of [•], 2015 (the "Credit Agreement"), among Empresas Públicas de Medellín S.A. E.S.P. (the "Borrower"), the financial institutions referred to as "Lenders" in the Credit Agreement (the "Lenders") and the Administrative Agent.

This opinion is being furnished to you pursuant to Section 5.01(d) of the Credit Agreement. Capitalized terms used herein and not otherwise defined herein shall have the meanings assigned to them in the Credit Agreement.

In rendering the opinions set forth below, we have examined originals, certified copies or forms otherwise identified to our satisfaction of the following documents, corporate records or instruments:

- a) Executed copy of the Credit Agreement.
- b) Executed copies of the Notes and their corresponding Instructions Letters.
- c) Executed copy of the Fee Letters.

- d) Copy of Acuerdo No. 69 of 1997 and 12 of 1998 from the Concejo de Medellín, including the powers of the general manager of EPM and his general authority to execute agreements on behalf of the Borrower, in its sole capacity as legal representative of the Borrower.
- e) Copy of Resolutions No. 1348 of May 17, 2012, No. 3071 of September 2, 2014 and No. 4783 of December 22, 2015 from the Ministry of Finance and Public Credit necessary for approving execution of the final version of the Opinion Documents and authorizing their execution; and
- f) Photocopy of current by-laws (estatutos) of the Borrower, contained under the Acuerdo Municipal No. 12 of 1998 of the Municipality of Medellín as amended.

The documents listed in paragraphs (a) through (c) are referred to herein as the " $\underline{\text{Opinion}}$ $\underline{\text{Documents}}$ ".

In addition, we have also reviewed the originals, or certified, conformed or reproduced copies, of all records, agreements, instruments and documents as we have deemed relevant or necessary as the basis for the opinions hereinafter expressed.

In rendering the opinions expressed below, we have assumed, without any independent investigation or verification of any kind: (i) the genuineness of all signatures; (ii) the authenticity of all agreements and documents submitted to us as originals and the conformity to the originals of such agreements and documents submitted to us as copies; (iii) the due organization, existence and good standing of all parties, other than the Borrower; (iv) the capacity, power and authority of all parties, other than the Borrower, to execute the Opinion Documents and the due authorization, execution and delivery of the Opinion Documents by all parties thereto, other than the Borrower; and (v) that the Opinion Documents are legal, valid, binding and enforceable against the parties thereto in accordance with their respective terms under, and comply in all respects in form and substance with, the laws of the respective jurisdiction chosen to govern the same (except as set forth in our opinion below).

The opinions rendered herein are limited in all respects to the laws of Colombia as they stand at the date hereof and as they are currently interpreted. We do not express any opinion of any jurisdiction other than Colombia.

Based upon the foregoing, and in reliance thereon, and subject to the limitations, qualifications and exceptions set forth herein, we are of the following opinion:

- The Borrower is a public utility services company duly organized as a *empresa industrial y comercial del Estado*, validly existing and in good standing under the laws Republic of Colombia ("Colombia"), and has all requisite corporate power and authority and all governmental licenses, authorizations, permits, consents and approvals necessary to execute and deliver the Opinion Documents and to comply with its obligations thereunder.
- 2. Each Opinion Document to which the Borrower is a party has been duly executed and delivered by the Borrower and constitutes the legal, valid and binding obligation of the Borrower, enforceable against the Borrower in accordance with its terms, except as enforceability may be limited by applicable bankruptcy, insolvency, moratorium, reorganization, fraudulent conveyance, transfer or similar laws of general applicability affecting the enforcement of creditors' rights generally.
- 3. The execution and delivery by the Borrower of each of the Opinion Documents and the performance of its obligations thereunder (a) are within the Borrower's corporate powers, (b) have been duly authorized by all necessary corporate action of the Borrower, (c) have been duly authorized by the Ministry of Finance and Public Credit of Colombia in accordance with Colombian public indebtedness regulations and applicable laws, and (d) do not and will not (1) contravene any of the Borrower's organizational documents, or (2) violate any laws of general application in Colombia or otherwise applicable to the Borrower.
- 4. Neither the execution and delivery of the Opinion Documents, nor the consummation of any other of the transactions therein contemplated, nor the fulfillment of the terms thereof will conflict with, result in a breach or violation of, or imposition of any lien, charge or encumbrance upon any property or asset of any of the Borrower pursuant to (i) its articles of incorporation or by-laws or (ii) any statute, law, rule, regulation, judgment, order or decree of Colombia.
- 5. The Notes constitute effective and valid executable negotiable instruments (titulo valor) in Colombia. As a consequence of a payment default under the terms of the Opinion Documents, each Lender shall be entitled to initiate a collecting procedure in a court in Colombia under the respective Note issued on its behalf and upon the terms provided for in its corresponding Instructions Letter.
- No license, consent or approval or other action by, or notice to or filing with, any Governmental Authority is necessary for the due execution, delivery and performance by the Borrower of the Opinion Documents or for the legality, validity or enforceability thereof

- against the Borrower except for (i) the approval of the Ministry of Finance and Public Credit of Colombia, and (ii) the filing of a copy of the Credit Agreement (together with its Schedules and Exhibits) before the Central Bank, which shall be made by the Borrower once said document is executed and prior to the first Loan.
- 7. The choice of New York law as the governing law of the Credit Agreement is, under the laws of Colombia, a valid and effective choice of law and the courts of Colombia would give effect to such choice of law in any action, suit or proceeding arising out of, or otherwise relating to, the Credit Agreement. The submission by the Borrower to the non-exclusive jurisdiction of the courts of the State of New York sitting in New York County, and of the United States District Court for the Southern District of New York, and any appellate court from any thereof and the appointment of CT Corporation System as its authorized agent for service of process and the manner of effecting service of process as described in the Credit Agreement, is legal, valid, binding and enforceable under the laws of Colombia.
- 8. Any final judgment obtained against the Borrower in a foreign court relating to any of the Opinion Documents would be recognized, conclusive and enforceable in the courts of Colombia, without reconsideration of the merits, *provided* that the exequatur proceedings described in Sections (A) and (B) on the qualifications section of this opinion, are fulfilled.
- 9. No income, withholding or other taxes are payable by or on behalf of the Borrower to Colombia or to any political subdivision or taxing authority thereof or therein in connection with the Opinion Documents. The Borrower is permitted to pay the Lenders and the Administrative Agent any additional amounts which may be required under the Opinion Documents to compensate it for any amount so withheld.
- 10. The Borrower is subject to civil and commercial law with respect to its obligations under each Opinion Document and the execution, delivery and performance of the Opinion Documents by the Borrower constitute private and commercial acts rather than public or governmental acts. Except as provided in Section (T) of the qualifications section in this opinion, neither the Borrower nor any of its properties is entitled to any immunity (sovereign or otherwise) from any legal action, suit or proceeding, from the jurisdiction of any court or from set-off or any legal process (whether through service or notice, attachment prior to judgment, attachment in aid of execution of judgment. execution of judgment or otherwise) under the laws of Colombia in respect of its obligations under the Opinion Documents.
- 11. The obligations of the Borrower under each Opinion Document do rank and will rank at least pari passu in priority of payment and in all other respects with all other present and future unsubordinated External Indebtedness of the Borrower (other than indebtedness

- which is mandatorily preferred by virtue of any provision of law of general application in Colombia).
- 12. There are no restrictions or requirements which limit the availability or transfer of foreign currency for the purpose of the performance by the Borrower of its obligations under any Opinion Document. Notwithstanding this, the Borrower shall file the required registrations described in Section P to the qualifications Section of this opinion.
- 13. Each of the Opinion Documents is in proper legal form under the laws of Colombia for enforcement thereof in accordance with its terms against the Borrower in the courts of Colombia, and to ensure the legality, validity, enforceability or admissibility in evidence of each Opinion Document in Colombia, it is not necessary that any Opinion Document or any other document be filed, recorded or registered at this time with any court or other authority in Colombia or that any stamp, transaction, registration or similar tax be paid on or in respect of any Opinion Document, *provided* that the required proceedings described in Section P of the qualifications section of this opinion are fulfilled.
- 14. Neither the Lenders nor the Administrative Agent will be deemed resident, domiciled, carrying on business or subject to taxation in Colombia or be required to be licensed, qualified or entitled to do business in Colombia or to comply with the requirements of any foreign registration or qualification statute in Colombia, in each case by reason only of the execution, delivery, performance or enforcement of the Opinion Documents or any document or instrument referred to therein or issued in connection therewith.
- 15. It is not necessary under the laws of Colombia in order to enable the Lenders or the Administrative Agent to enforce any Opinion Document or to exercise its rights or remedies under any Opinion Document that the Lenders or the Administrative Agent be licensed, qualified or entitled to do business in Colombia or to comply with the requirements of any foreign registration or qualification statute in Colombia.

The opinions set forth above are subject to the following qualifications and exceptions:

(A) A final and conclusive judgment (not subject to appeal) of the courts of the State of New York (or an international award rendered in a forum outside Colombia) for the payment of money rendered against the Borrower in respect of any Opinion Document, would be recognized by the courts of Colombia, subject to obtaining the *Exequatur* of the judgment from the Supreme Court of Colombia (*Corte Suprema de Justicia de la República de Colombia*). Pursuant to Articles 693 through 695 of Colombia's Code of Civil Procedure (*Código de Procedimiento Civil*) (which will be abrogated by article 605 et seq. of Law 1564 of 2012 (*Código General del Proceso*) once it is ruled effective by the Colombian Council of the Judiciary

(Consejo Superior de la Judicatura)), the courts of Colombia will give effect to and enforce a judgment obtained in a court outside Colombia without re-trial or re-examination of the merits of the case provided (i) that there exists a treaty or convention relating to recognition and enforcement of foreign judgments between Colombia and the country (or state in case of any Federal Republic) of origin of the judgment or, in the absence of such treaty, that proper evidence is provided to the Supreme Court of Colombia to the effect that the courts of the country of origin of the subject judgment would recognize and enforce Colombian judgments, and (ii) that the subject judgment fulfills the requirements listed in (B) below.

- (B) In order to enforce a foreign judgment in Colombia, it must first be submitted to "Exequatur" proceedings in accordance with Articles 693 through 695 of Colombia's Code of Civil Procedure (Código de Procedimiento Civil) (which will be abrogated by article 605 et seq. of Law 1564 of 2012 (Código General del Proceso) once it is ruled effective by the Colombian Council of the Judiciary (Consejo Superior de la Judicatura) before the Supreme Court of Colombia which, in addition to the matter referred to in (A)(i) above, shall examine whether the following requirements have been fulfilled:
- (i) that the judgment does not refer to in-rem rights over assets located in Colombia;
- (ii) that the defendant was afforded due service of process in accordance with the laws of the judgment's country of origin if the judgment has been rendered in a contentious matter, which issuance is presumed if the judgment is executory;
- (iii) that the judgment is final and executory in accordance with the laws of the country of origin of the judgment, and that a duly legalized copy of the judgment (together with an official translation into Spanish if the judgment is issued in a foreign language) be filed to the competent court in Colombia with the plaintiff's request for "Exequatur";
- (iv) that the judgment is not contrary to Colombian public order provisions, *except* for rules of procedure;
- (v) that the matter of the judgment is not subject to the exclusive jurisdiction of the courts of Colombia; and
- (vi) that there are no proceedings in Colombia or any final judgments rendered by Colombian courts over the same subject matter.
- (C) A judgment issued in a foreign court ordering the payment of money by the Borrower under an Opinion Document would not conflict with public order laws of Colombia. Proceedings for execution of

- a money judgment by attachment or execution against any assets or property located in Colombia would be within the exclusive jurisdiction of Colombian courts.
- (D) In the course of the "Exequatur" proceedings, both the plaintiff and the defendant are allowed the opportunity to request that evidence be collected in connection with the issues listed in (B) above; and before the judgment is rendered, each party may present final allegations in support of such party's position.
- (E) If, pursuant to the Opinion Documents governed by a law other than Colombian law and submitted to the jurisdiction of courts other than Colombian courts, a party to such Opinion Documents were to initiate judicial actions in Colombian courts, there are no explicit laws and regulations or Colombian court decisions with precedential value pursuant to which it is possible to conclusively determine if a Colombian judge would assume jurisdiction over the case or, instead, determine that the case is subject exclusively to the forum expressly agreed to by the parties under the Opinion Documents.
- (F) Any proceeding to enforce a judicial decision by means of seizure, attachment or execution against assets or property, or against any right or interest in assets or properties located in Colombia, is subject to the exclusive jurisdiction of the Colombian courts.
- (G) Section 8 of the Credit Agreement will not be enforceable in Colombia pursuant to Law 550 of 1995 if the Lenders invoke the occurrence of an Event of Default pursuant to such sections as a result of the Borrower having been admitted to an insolvency proceeding under Law 550 of 1995.
- (H) Civil procedure rules are considered public order laws and therefore cannot be modified or waived by private agreements. To the extent that the parties to the Opinion Documents commence enforcement actions before Colombian courts instead of commencing them at foreign courts (which final rulings may subsequently be enforced in Colombia through *Exequatur* proceedings as described above), any waivers made by the parties to such documents in respect of Colombia's rules of civil procedure may be rendered unenforceable.
- (I) In any proceeding in Colombia, service of notice to the parties thereto must be made in accordance with the provisions of Code of Civil Procedure (Código de Procedimiento Civil) or Colombian General Procedure Code (Código General del Proceso) (as applicable once the latter is ruled effective by the Colombian Council of the Judiciary). Contractual provisions regarding service of notice procedures may not be enforceable with respect to any proceeding in Colombia.

- (J) In any proceeding in Colombia in which a law of a foreign country were to be applied, there should be evidence of the law sought to be applied, through a copy of such law duly issued and promulgated by the competent authorities or, when a written law does not exist, through the deposition or affidavit of two or more lawyers admitted in the relevant jurisdiction regarding such applicable law.
- (K) According to Section 260 of Colombia's Code of Civil Procedure (Código de Procedimiento Civil) (which will be abrogated by article 251 of Law 1564 of 2012 (Código General del Proceso) once it is ruled effective by the Colombian Council of the Judiciary (Consejo Superior de la Judicatura)), in order for a document written in a foreign language to be admissible evidence before a Colombian court, the relevant document must be translated into Spanish either by Colombia's Ministry of Foreign Affairs or by a translator duly appointed by a competent judge.
- (L) According to Article 823 of Colombia's Code of Commerce (*Código de Comercio*) the Spanish version of a document used to evidence obligations or contracts shall prevail over the foreign language version of such document. In the event that a party to an enforcement action considers that the available translation into Spanish of documents written in a foreign language is not accurate, such party has the right to object any such inaccuracy in such enforcement proceedings.
- (M) In accordance with Article 902 of Colombia's Code of Commerce ($\it C\'odigo\ de\ Comercio$), any nullification of a provision of a Colombian document to which the Borrower is a party, would nullify such document entirely if the parties would not have entered into such document in the absence of such nullified provision.
- (N) The enforcement of the Opinion Documents in Colombia may be limited by the applicable statute of limitations. Pursuant to Section 2514 of Colombia's Civil Code ($C\'odigo\ Civil$), a waiver to the statute of limitations can only be granted once the relevant statute of limitations has elapsed.
- (O) According to Colombian laws, the laws applicable to a given agreement are those in existence at the time of execution, even if those laws change in the future, provided that the changes are not related to, or do not affect, public order laws, in which case public order laws become immediately effective. Colombian exchange control regulations are deemed to be public order laws. Therefore, the ability of the Borrower to perform its obligations payable in foreign currency (and the ability of any person to remit the proceeds of any judgment award issued by a court in Colombia in foreign currency out of Colombia) will be subject to foreign exchange regulations in effect at the time of the relevant payment or remittance.

- (P) Under Colombian foreign exchange regulation: (i) the Credit Agreement (and any disbursement and repayment thereof) must be reported to the Colombian Central Bank as foreign indebtedness of the Borrower; (ii) the ability of the Borrower to perform its obligations payable in foreign currency under the Opinion Documents from Colombia (and the ability of any person to remit the proceeds of any judgment award issued by a court in Colombia in a foreign currency out of Colombia) will be subject to the Colombian foreign exchange rules and regulations in effect at the time; (iii) the Borrower may have to amend the initial registration of the Agreement if there is any amendment to its financial terms and conditions or any transfer in total or in part to other Lender/s.
- (Q) The provisions of the Opinion Documents which treat certain determinations as conclusive may be subject to review in a proceeding in Colombia to determine the correctness of such determinations.
- (R) The indemnification provisions set forth in Section 11.08 of the Credit Agreement may be limited by the judicial determination of legal costs, fees and judicial amounts as determined by Colombian courts in accordance with Article 365 of Law 1564 of 2012 (Código General del Proceso).
- (S) Under Colombian law, charging interest on interest (whether accrued or unpaid) is not permitted unless those interests are charged as permitted under Article 886 of Colombia's Code of Commerce (*Código de Comercio*), meaning these are to be explicitly stated as agreed between the parties to be charged once their time due exceeds one (1) year.
- Pursuant to Articles 15 and 16 of Colombia's Civil Code (Código Civil), the waiver of rights is permissible provided that said waiver only affects the rights of the waiving party. Under Colombian law any immunity from proceedings which might be available in the future cannot be validly waived in advance. The Borrower and its properties are not subject to immunity from the jurisdiction of courts or from set-off or any legal process subject to the terms, conditions, limitations or exceptions under: (i) article 1677 of the Colombian Civil Code (Código Civil) and Article 684 of Colombia's Code of Civil Procedure (Código de Procedimiento Civil) (which will be abrogated by article 594 of Law 1564 of 2012 (Código General del Proceso) once it is ruled effective by the Colombian Council of the Judiciary (Consejo Superior de la Judicatura)).
- (U) Under Colombian laws there are no equitable remedies or injunctive relief, except for fundamental constitutional rights, specific performance of contracts and precautionary measures and remedies in unfair trade practices actions.

We are qualified to practice law in Colombia and do not purport to be experts on, or to express any opinion herein concerning, any laws other than the laws of Colombia

as in effect on the date hereof. We assume no obligation to update this opinion with respect to changes occurring subsequent to the date hereof.

This opinion is addressed to you and is solely for your benefit and the benefit of your assigns and participants, and only in connection with the transactions contemplated by the Opinion Documents. This opinion may not be relied upon by you or the Lenders for any other purpose. This opinion is delivered as of its date and without any undertaking to advise the Lenders or any other person of any changes of law or fact that occur after the date of this opinion, even though any such changes may affect a legal analysis or conclusion, or a confirmation of information, expressed in this opinion.

Very truly yours,

MOSQUERA ABOGADOS

EXHIBIT G

FORM OF LEGAL OPINION OF MILBANK, TWEED, HADLEY & McCLOY

[•], 2015

To the Lenders that are parties to the Credit Agreement referred to below, and The Bank of Tokyo-Mitsubishi UFJ, LTD, as Administrative Agent for such Lenders (the "Administrative Agent")

Ladies and Gentlemen:

We have acted as special New York counsel to Empresas Públicas de Medellín E.S.P (the "Borrower") in connection with the Credit Agreement dated as of [●], 2015 (the "Credit Agreement"), among the Borrower, the financial institutions referred to as "Lenders" in the Credit Agreement (the "Lenders") and the Administrative Agent. Terms defined in the Credit Agreement have the same respective defined meanings when used herein.

In rendering the opinions expressed below, we have examined

- (a) an executed counterpart of the Credit Agreement;
- (b) an executed counterpart of the Lenders' Fee Letters, dated as of [●], 2015, between the Borrower and the Administrative Agent;
- (c) an executed counterpart of the Administration Fee Letter, dated as of [•], 2015, between the Borrower and the Administrative Agent; and

(d) such other records of the Borrower and such other agreements, instruments, certificates and other documents as we have deemed necessary as a basis for the opinions expressed below.

For the purposes of this opinion letter, (i) "Opinion Documents" means the document described in clauses (a) through (c) above, (ii) "Specified Agreement" means the agreement specified in Schedule I and (iii) "Applicable Law" means, collectively, any Federal law of the United States, any law of the State of New York and any rule or regulation thereunder that, in each case, in our experience, is customarily recognized to apply to transactions of the kind contemplated by the Opinion Documents.

In our examination, we have assumed the genuineness of all signatures, the authenticity of all documents submitted to us as originals and the conformity with authentic original documents of all documents submitted to us as copies. When relevant facts were not independently established, we have relied upon representations made in or pursuant to the Opinion Documents or in certificates delivered by or on behalf of the Borrower pursuant thereto. We have also assumed (i) that the Opinion Documents have been duly authorized, executed and delivered by all of the parties thereto, that all signatories thereto have been duly authorized and that all such parties are duly organized and validly existing and have the power and authority (corporate or limited liability company or other) to execute, deliver and perform the same, (ii) (except, to the extent set forth below, as to the Borrower) that the Opinion Documents constitute a legal, valid, binding and enforceable obligation of all of the parties thereto, and (iii) that all authorizations, approvals or consents of (including without limitation all foreign exchange controls), and all filings or registrations with, any governmental or regulatory authority or agency of Colombia (including the central bank of Colombia) required for the making and performance by the Borrower have been obtained or made and are in effect.

Based upon and subject to the foregoing and subject also to the comments and qualifications set forth below, and having considered such questions of law as we have deemed necessary as a basis for the opinions expressed below, we are of the opinion that:

- The Opinion Documents constitute a legal, valid and binding obligation of the Borrower, enforceable against the Borrower in accordance with their terms, except as may be limited by bankruptcy, insolvency, reorganization, fraudulent conveyance or transfer. moratorium or other similar laws relating to or affecting the rights of creditors generally, and subject to the possible judicial application of foreign laws or governmental action affecting the rights of creditors generally, and except as the enforceability of the Opinion Documents is subject to the application of general principles of equity (regardless of whether considered in a proceeding in equity or at law), including without limitation (i) the possible unavailability of specific performance, injunctive relief or any other equitable remedy and (ii) concepts of materiality, reasonableness, good faith and fair dealing.
- (2) The execution and delivery by the Borrower of the Opinion Documents do not, and the

performance by the Borrower of its obligations under the Opinion Documents will not, (i) violate any Applicable Law or (ii) breach or constitute a default under the Specified Agreement.

- (3) No authorization, approval or other action by, and no notice to or filing with, any governmental or regulatory body or instrumentality under Applicable Law is required for the due execution, delivery or performance by the Borrower of the Opinion Documents.
- (4) The Borrower is not required to, and immediately after giving effect to the making of the Loans and the use of the proceeds thereof in accordance with the Credit Agreement the Borrower will not be required to, register as an "investment company" as defined in the Investment Company Act of 1940, as amended.
- (5) The making of the Loans and the application of the proceeds thereof in accordance with the Credit Agreement will not violate Regulations U or X as promulgated by the Board of Governors of the Federal Reserve System.

The foregoing opinions are also subject to the following comments and qualifications:

- (A) The enforceability of provisions in the Opinion Documents to the effect that terms may not be waived or modified except in writing may be limited under certain circumstances.
- (B) The enforceability of provisions in the Opinion Documents providing for indemnification, contribution or exculpation may be limited by (i) laws rendering unenforceable indemnification contrary to law or the public policy underlying such law, and (ii) laws limiting the enforceability of provisions exculpating or exempting a party from, or requiring indemnification of a party for, liability for its own action or inaction, to the extent the action or inaction involves gross negligence, recklessness, willful misconduct or unlawful conduct.
- (C) We express no opinion as to (i) the effect of the laws of any jurisdiction in which any Lender is located (other than New York) that limits the interest, fees or other charges it may impose for the loan or use of money or other credit, (ii) Section 9.02 insofar as such section relates to the subject matter jurisdiction of the United States District Court for the Southern District of New York to adjudicate any controversy related to the Credit Agreement, (iii) the waiver of inconvenient forum set forth in Section 9.02 of the Credit Agreement (or any similar provision in the Opinion Documents) with respect to proceedings in any federal court of the United States sitting in New York City, (iv) Section 11.09 of the Credit Agreement (or any similar provision of the Opinion Documents) or (v) Section 9.04 of the Credit Agreement to the extent it relates to immunity acquired after the date of execution and delivery of the Credit Agreement.
- (D) We express no opinion as to any Federal or state securities laws except as expressly set forth in paragraphs (4) and (5) above.

For purposes of our opinion in (E) paragraph 2(ii) above, (1) we have assumed that (i) there are no agreements or understandings, written or oral, or other course of dealing, among the parties to the Specified Agreement that would alter the plain meaning thereof, (ii) the Specified Agreement has not been modified and (iii) the Specified Agreement would be interpreted as written in accordance with its plain meaning, (2) we express no opinion as to any provision of the Specified Agreement to the extent that such opinion would depend upon the making of any determination of an accounting matter or any computation in respect of a financial undertaking or condition, (3) we express no opinion as to any breach of any cross default or analogous provision under the Specified Agreement to the extent that such breach results from a default (however defined or described) under an agreement other than an Opinion Document and (4) we express no opinion with respect to any provision of any agreement (other than the Specified Agreements and the Opinion Documents) that is incorporated in or otherwise referred to in the Specified Agreements.

The foregoing opinions are limited to matters involving the Federal laws of the United States and the law of the State of New York, and we do not express any opinion as to the law of any other jurisdiction. Without limiting the foregoing, we do not hold ourselves out as experts on, or purport to advise on, the laws of Colombia and have rendered no opinion with respect to the foregoing.

This opinion letter is provided to you by us as special New York counsel to the Borrower pursuant to the Credit

Agreement and may not be relied upon by any other person or for any purpose other than in connection with the transactions contemplated by the Credit Agreement without our prior written consent in each instance. You may, however, deliver a copy of this opinion to your accountants, attorneys, and other professional advisors, to governmental regulatory agencies having jurisdiction over you, to permitted transferees of the Credit Agreement in connection with such transfer and to participants in connection with their purchase of a participation interest in the Credit Agreement, and such transferees (but not such participants) may rely on this opinion as if it were addressed and had been delivered to them on the date of this opinion, unless statements in this opinion would be affected by the status of the transferee.

Very truly yours,

[]

Schedule I

Specified Agreement

The Indenture in respect of the 7.625% Senior Notes due 2024, dated as of September 10, 2014, among the Borrower, The Bank of New York Mellon, as Trustee, Security Registrar, Paying Agent, Transfer Agent and Calculation Agent, The Bank of New York Mellon (Luxembourg) S.A., as Luxembourg Transfer Agent and The Bank of New York Mellon, acting through its London Branch as Co-Paying Agent.

AMENDMENT TO THE LOAN AGREEMENT

Dated as of December [], 2015

Amendment Agreement to Loan Agreement (Loan# 2217/ OC-CO, signed on December 14, 2011), dated as of December [], 2015, by and between Empresas Públicas de Medellín E.S.P. (the "Borrower") and Inter-American Development Bank ("IDB").

WITNESSETH:

WHEREAS, the Borrower and the IDB are parties to that certain Loan Agreement, dated as of December 14, 2011 (the "Loan Agreement");

WHEREAS, the Borrower and the IDB have agreed that certain terms of the Loan Agreement are to be amended, as herein specified.

NOW, THEREFORE, in consideration of the mutual covenants set forth herein and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the parties hereto agree as follows:

- Definitions. Capitalized terms used but not defined in this Amendment to the Loan Agreement (the "Amendment Agreement") have the meanings assigned to them in the Loan Agreement. The rules of interpretation set forth in Section 1.2 (Interpretation) of the Loan Agreement shall apply to this Amendment Agreement.
- 2. Amendment to the Loan Agreement.
 - a. **Definitions.** Section 1.1 (*Definitions*) of the Loan Agreement is hereby amended by the insertion of the following term:

"Loan to Trust's loan portfolio Ratio means, with respect to the Trust and as of any relevant determination date, the ratio obtained by dividing (a) the balance of the outstanding principal amount of the Loan by (b) the balance of the Trust's accounts receivable (i.e. cuentas por cobrar), provided that for purposes of calculating such ratio, the balance of the Trust's accounts receivable as of any relevant determination date shall be measured in Dollars at the Colombian Peso Market Exchange Rate (i.e. Tasa de Cambio Representativa del Mercado or TRM) in effect on such date published by the Superintendence of Finance at www. superfinanciera.gov.co."

b. Financial Ratio. Section 6.2.1 of the Loan Agreement is hereby amended by replacing where it reads "Trust Debt to Contribution Ratio", as follows: "Financial Ratio. Permit at any time the Loan to Trust's loan portfolio Ratio to be greater than 1.0:4.0."

c. **Audited Annual Financial Statements**. Section 6.3.1.3.2 of the Loan Agreement is hereby amended by replacing where it reads "Trust Debt to Contribution Ratio", as follows:

"setting forth in reasonable detail all information necessary to calculate (and providing the calculations necessary to determine) the Loan to Trust's loan portfolio Ratio during the applicable period and as at the last day of the period covered, as relevant, by the Financial Statements."

d. *Unaudited Quarterly Financial Statements*. Section 6.3.2.2.3 of the Loan Agreement is hereby amended by replacing where it reads "Trust Debt to Contribution Ratio", as follows:

"setting forth in reasonable detail all information necessary to calculate (and providing the calculations necessary to determine) the Loan to Trust's loan portfolio Ratio during the applicable period and as at the last day of the period covered, as relevant, by such Financial Statements."

e. Form of Certificate of Authorized Representative. Exhibit 9 (Form of Certificate of Authorized Representative) is hereby amended by replacing where it reads "Trust Debt to Contribution Ratio", as follows:

"We hereby submit the following information used to calculate and determine the Loan to Trust's loan portfolio Ratio during the applicable period and as at the last day of the period covered, as relevant, by such Financial Statement."

- 3. Apart from the amendment expressly provided herein no other waivers, amendments or consents are given or implied under this letter and the Financing Documents shall remain unchanged and in full force and effect.
- 4. This letter is not and shall not be construed as or deemed to constitute (i) a permanent waiver of any condition, requirement, Default or event of Default set forth in the Financing Documents; (ii) a waiver or impairment of any right, remedy, power or privilege of the IDB under any provision of any Financing Document; (iii) a waiver of any obligation under any provision of any Financing Document; (iv) a novation of any Financing Document, other than those expressly referred to herein.

- 5. The granting of the consent to amend the Loan Agreement by IDB does not establish a course of dealing or otherwise obligate IDB to agree to any waiver or amendment in the future.
- 6. This Amendment Agreement may be executed in several counterparts, each of which is an original, and all of which together constitute one and the same Amendment Agreement.
- 7. The provisions of Section 8.10 (Applicable Law and Jurisdiction) of the Loan Agreement shall be

- incorporated herein *mutatis mutandis* as if set forth herein in full.
- 8. This Amendment Agreement binds and benefits the respective successors and assigns of the parties hereto.

(Signature Page Follows)

IN WITNESS WHEREOF, the parties, acting through their duly Authorized Representatives, have caused this Amendment Agreement to be signed, as of the date first above written.

INTER-AMERICAN DEVELOPMENT BANK

Name: Luiz Ros	
Title: Manager – Opportu	unities for the Majority Sector
	EMPRESAS PÚBLICAS DE MEDELLÍN E.S.P
Name: Juan Esteban Ca Title: Gerente General	lle Restrepo

CONTRATO DE EMPRÉSTITO INTERNO PARA OPERACIÓN DE CREDITO TRANSITORIO CELEBRADO ENTRE LA CENTRAL HIDROELECTRICA DE CALDAS S.A E.S.P Y BANCO DAVIVIENDA S.A.

Entre los suscritos, de una parte CENTRAL HIDROELECTRICA DE CALDAS S.A ESP, Empresa de servicios públicos mixta, conformada como sociedad anónima comercial, sometida al régimen legal aplicable a las empresas de servicios públicos y a las normas especiales que rigen las empresas del sector eléctrico, reformados sus estatutos según consta en escritura pública No. 2497 del 04 de Abril de 2009, otorgada en la Notaria Segunda del Circulo de Manizales, Caldas, y en escritura pública

No. 279 del 14 de Marzo de 2013, otorgada en la Notaria Quinta del Circulo de Manizales, Caldas, con domicilio principal en la ciudad de Manizales, representado en este acto por **MARTHA LIBIA CASTAÑO TOVAR** mayor de edad, identificada con cédula de ciudadanía No 30.291.119 de Manizales (Caldas), obrando en su calidad de segundo suplente del Gerente, de acuerdo con lo establecido en el certificado de existencia y representación legal expedido por la Cámara de Comercio de Central Hidroeléctrica

de Caldas, y debidamente facultada según consta en el Acta No.613 de la Junta Directiva del 9 de Julio de 2014, quien en adelante se denominará **EL PRESTATARIO** y el **BANCO DAVIVIENDA S.A**, establecimiento bancario legalmente constituido con domicilio principal en Bogotá D.C, representado en esta acto por ALEJANDRO ROLDAN PATIÑO identificado con la cédula de ciudadanía No 75.078.951 de Manizales, obrando en su carácter de Gerente de la Sucursal Caldas, como consta en el certificado de existencia y representación legal, quien en adelante se denominará **EL PRESTAMISTA**, hemos celebrado el siguiente Contrato de Empréstito Interno para operación de crédito transitorio, en adelante el Contrato de Empréstito, que se regirá por las cláusulas que se indican a continuación:

PRIMERA: Objeto y Cuantía. EL PRESTAMISTA ha acordadoparacon ELPRESTATARIO otorgarle un empréstito interno transitorio hasta por la suma de CINCUENTA MIL MILLONES DE PESOS (\$50.000.000.000,00) MONEDA LEGAL COLOMBIANA. El presente crédito será cancelado con recursos provenientes de crédito con plazo mayor a un año respecto del cual ya existe oferta en firme.

SEGUNDA: Plazo y Condiciones del Empréstito. Las condiciones para el presente Contrato de Empréstito interno de corto plazo transitorio son las siguientes: A) Moneda. EL PRESTAMISTA desembolsará a favor de EL PRESTATARIO el empréstito en moneda legal colombiana y será reembolsado por EL PRESTATARIO en esta misma moneda. B) Plazo Total y Amortización. EL PRESTATARIO pagará el presente Empréstito a EL PRESTAMISTA en un plazo que no podrá superar los tres meses contados a partir de la fecha de cada desembolso, pagadero en una cuota de conformidad con la carta de instrucciones para el desembolso del crédito Transitorio. EL PRESTATARIO cancelará el valor total del crédito, sumado capital e intereses al vencimiento del plazo aquí dispuesto, el cual en ningún caso podrá superar doce meses contados a partir de la firma del presente contrato. C) Financiación: EL PRESTAMISTA financiará el crédito con recursos propios. En consecuencia, durante el plazo del empréstito, EL PRESTATARIO pagará sobre saldos adeudados bajo el presente contrato de empréstito, intereses corrientes liquidados a la tasa IBR (3 Meses) TV certificada por el Banco de la República, o la entidad que haga sus veces, adicionada en 2.34% puntos porcentuales (IBR T.V. + 2.34%). El interés así estipulado se convertirá en su equivalente trimestre vencido y será pagadero al vencimiento del plazo aquí estipulado. En cada TRIMESTRE se ajustará el interés corriente teniendo en cuenta la IBR vigente, o la tasa que la reemplace, a la fecha de inicio de cada período de causación de intereses, incrementado en el mismo número de puntos porcentuales anteriormente indicados y se pagarán el día del vencimiento o el día hábil siguiente en caso de que el día del vencimiento no corresponda a un día hábil bancario. Los intereses remuneratorios serán calculados con base en meses de treinta (30) días y años de trescientos sesenta (360) días. El IBR al que se refiere este título es el publicado por el Banco de la República o la entidad que se designe y corresponde a la referencia de corto plazo denominada en pesos colombianos, que refleja el costo del dinero en el mercado monetario. En caso de que por cualquier razón la IBR no se publique, este crédito y sus correspondientes intereses, se liquidaran con base en la tasa de intervención de política monetaria del Banco de la República vigente a la fecha de iniciación de cada periodo de liquidación adicionada en 2.34 puntos trimestre vencidos. D) Intereses moratorios. Si el pago del capital adeudado no se efectúa en la fecha prevista para el vencimiento, EL PRESTATARIO reconocerá y pagará intereses moratorios sobre el monto del capital en mora y por cada día de retardo, desde el día de la mora y hasta el día en que dicho monto sea pagado efectivamente, a la tasa máxima permitida por la ley, sin perjuicio de las correspondientes acciones legales de cobro. E) Desembolso. EL PRESTATARIO podrá solicitar uno o varios desembolsos hasta por el monto total del empréstito, para lo cual contará con plazo hasta 3 meses contados a partir del 20 de noviembre de 2015. Vencido este plazo, se entenderá que no se utilizará el saldo pendiente del crédito y por consiguiente EL PRESTAMISTA no estará obligado a entregar dichos recursos, salvo acuerdo escrito entre las partes. El desembolso de la totalidad de los recursos objeto del empréstito se efectuará de conformidad con el siguiente procedimiento: 1. EL PRESTATARIO solicitará por escrito el desembolso del empréstito a EL PRESTAMISTA. mínimo con un (1) día hábil bancario de antelación a la fecha en que el mismo se requiera, indicando la cuenta o cuentas en las que el dinero deba ser abonado. 2. Previo a cada desembolso, EL PRESTATARIO enviara carta de instrucciones para el desembolso a EL PRESTAMISTA, en donde se establecerá las condiciones financieras tales como: cuantía, la tasa de interés y forma de amortización y vencimientos, y fechas de pago de las cuotas de capital e intereses, de acuerdo con las estipulaciones consignadas en este contrato. 3. Una vez comunicado, EL PRESTAMISTA se obliga a abonar el dinero en la fecha indicada para el desembolso por EL PRESTATARIO en la entidad financiera y número de cuenta que designe, según condiciones de firma. PARAGRAFO: EL PRESTARIO respalda los desembolsos objeto del presente contrato de crédito, en un pagaré abierto que tiene previamente firmado con EL PRESTAMISTA. F) Prepago. EL PRESTATARIO en cualquier momento de la vigencia del presente Contrato de Empréstito podrá a pre-pagarlo total o parcialmente y, EL PRESTAMISTA no podrá cobrar penalización, prima o costo adicional.

TERCERA. Destinación. EL PRESTATARIO destinará los recursos objeto del presente Contrato de Empréstito como adelanto a operación de largo plazo para la ejecución de Plan de Inversiones 2015-2018.

CUARTA. Exigibilidad Anticipada. EL PRESTAMISTA dando previo aviso por escrito a EL PRESTATARIO y sin necesidad de requerimiento judicial alguno, podrá declarar vencido el plazo de manera anticipada y exigir el pago inmediato de la totalidad del empréstito junto con los intereses adeudados, en los siguientes eventos: a) Mora o incumplimiento en el pago de cualquiera de las cuotas de capital o de los intereses de ésta obligación; b) Si EL PRESTATARIO incumple total o parcialmente cualquiera otra de las obligaciones diferentes al pago del Contrato de Empréstito y que no sea subsanada en un plazo de quince (15) días contados a partir de la comunicación escrita de EL PRESTAMISTA a EL PRESTATARIO; c) Incumplimiento de

EL PRESTATARIO en la destinación del empréstito; d) La existencia de acciones judiciales de cualquier naturaleza que afecten sustancialmente la capacidad financiera de **EI PRESTAMISTA**, de modo tal que impidan el debido y oportuno cumplimiento de las obligaciones surgidas de este Contrato de Empréstito.

QUINTA. **Declaraciones** del Prestatario. PRESTAMISTA suscribe el presente contrato, consideración a las siguientes declaraciones que efectúa EL PRESTATARIO a la firma del presente Contrato de Empréstito: EL PRESTATARIO declara que: 5.1 Estatus: Es una empresa de servicios públicos mixta válidamente existente y regulada por las leyes de Colombia, debidamente autorizada por la ley, las autoridades y los órganos colegiados competentes para celebrar el presente Contrato de Empréstito. Declara, igualmente, que los recursos a los que hace referencia el presente Contrato de Empréstito sólo se emplearán en la destinación señalada en este contrato 5.2 Autorizaciones: EL PRESTATARIO tiene todas las autorizaciones, registros y permisos legales, administrativos, gubernamentales y cualesquiera otra autorizaciones requeridos para: 1. permitirle el cumplimiento de sus obligaciones de pago para con el prestamista, en relación con el presente contrato de empréstito y la celebración y ejecución del contrato de empréstito. 2. Declara que las obligaciones de pago son validas, legalmente obligatorias y vigentes. 3. asegura que las personas que celebren y ejecuten el presente contrato en nombre del PRESTATARIO han sido debidamente autorizadas por los órganos pertinentes, para firmar los títulos de deuda, las notificaciones, comunicaciones y certificaciones en conexión con este. 4. Asegura que los términos y condiciones del presente contrato no desconocen las leyes aplicables, ni viola otras obligaciones o contratos celebrados. 5.3 Obligaciones vinculantes: Las obligaciones que asume el PRESTATARIO en desarrollo del presente contrato de los títulos de deuda son obligaciones legales y valederas de acuerdo a los términos del mismo 5.4. No violación a las leves ni a otros acuerdos: La ejecución de este contrato y cualquier otro documento ejecutado o que ha de ejecutarse bajo este contrato, y el cumplimiento del mismo no contraviene ninguna ley vigente u otra regulación, ni ningún otro contrato, convenio o endeudamiento. 5.5. EL PRESTATARIO no está incumpliendo contratos celebrados con terceras personas, ni existen reclamaciones ni procesos en su contra con relación a dichos contratos, de tal manera que tuvieran o pudieran tener un efecto adverso sobre su habilidad para cumplir sus obligaciones de pago bajo el presente contrato de empréstito. 5.6.Información Completa, Verdadera y Exacta: (i) Toda la información suministrada por EL PRESTATARIO a EL PRESTAMISTA es verdadera, completa y exacta en todos sus aspectos materiales y a la fecha de la entrega de la misma no tiene conocimiento de ningún hecho material o datos que no se hayan divulgado a EL PRESTAMISTA que si fueren divulgados pudieren tener un efecto adverso en la decisión de EL PRESTAMISTA de celebrar la presente operación con EL PRESTATARIO; (ii) que no tiene procesos de ninguna naturaleza, que de ser fallados en su contra. pueden conducir al incumplimiento de las obligaciones de pago adquiridas bajo el presente contrato de empréstito, ni de sanciones que puedan producir el mismo efecto. El PRESTATARIO se compromete a cumplir a partir de la firma del presente contrato a: 1. inmediatamente notificar a el **PRESTAMISTA** sobre cualquier litigio, arbitramento, demanda, requerimiento administrativo o decisión en contra, cuyas pretensiones y/o condena equivalgan individual o conjuntamente al diez por ciento (10%) del presupuesto de la correspondiente vigencia fiscal. 3. **EL PRESTATARIO** entiende los términos y condiciones del presente contrato de empréstito y los desembolsos a realizar en desarrollo del mismo y desea asumir las obligaciones establecidas con respecto a los mismos.

SÉXTA. Cesión. **EL PRESTAMISTA** no podrá ceder, endosar o traspasar el presente Contrato de Empréstito, ni el (los) pagaré(s) que se suscriban en desarrollo del mismo, sin el concepto previo y escrito de **EL PRESTATARIO**.

SÉPTIMA. Perfeccionamiento. Este Contrato de Empréstito se perfecciona con la firma de las partes. **EL PRESTATARIO** se obliga a publicar el presente Contrato de Empréstito en la Gaceta Oficial del Municipio de Medellín.

OCTAVA. Comunicaciones. Todo aviso, comunicación o solicitud que las partes deban dirigirse en virtud del presente Contrato de Empréstito se efectuarán por escrito y se considerarán realizadas desde el momento en que se reciba el documento correspondiente por el destinatario en la respectiva dirección que a continuación se indica: EL PRESTATARIO: Km 2 salida a Chinchiná Estación Uribe y EL PRESTAMISTA: Cr 22 No. 21-05 Piso 2.

NOVENA. Vencimiento en días feriados. Todo pago o cumplimiento de cualquiera otra obligación derivada del presente Contrato de Empréstito que deba efectuarse en sábado o en un día dominical o feriado, o un día de cierre bancario según la ley, se entenderá válidamente realizado en el primer día hábil bancario siguiente, sin que por esta circunstancia se cause recargo alguno.

DÉCIMA. Gastos por cobro judicial o extrajudicial. En caso de cobro judicial serán a cargo de **EL PRESTATARIO** las sumas que determine el juez competente. En caso de cobro extrajudicial, **EL PRESTAMISTA** presentará a **EL PRESTATARIO** para su pago, una relación detallada y justificada de los gastos respectivos.

DECIMA PRIMERA. Impuesto de Timbre. El presente Contrato de Empréstito y el pagaré que se suscriba en desarrollo del mismo están exentos del Impuesto de Timbre Nacional por constituir una operación de crédito público, de conformidad con lo dispuesto en el numeral 14 del artículo 530 del Estatuto Tributario, modificado por el artículo 8 de la ley 488 de 1998.

DECIMASEGUNDA. Inhabilidades o Incompatibilidades. EL PRESTAMISTA declara no hallarse incurso en ninguna causal de inhabilidades e incompatibilidades establecidas en la ley, para firmar y cumplir el presente Contrato de Empréstito.

DECIMA TERCERA. Requisitos previos al primer desembolso y los siguientes. Para que EL PRESTAMISTA realice el primer desembolso en desarrollo

del presente Contrato de Empréstito, es necesario que **EL PRESTATARIO** entregue:

Contrato de Empréstito debidamente firmado.

Carta de Instrucciones enviada por **EL PRESTATARIO** a EL **PRESTAMISTA** en donde se relacionan las condiciones financieras tales como: cuantía, la tasa de interés y forma de amortización y vencimientos, y fechas de pago de las cuotas de capital e intereses, de acuerdo con las estipulaciones consignadas en este contrato.

Documentos que acreditan el nombramiento, posesión y las autorizaciones otorgadas a quien firma el presente documento en nombre de **EL PRESTATARIO**.

DECIMA CUARTA. Apropiaciones Presupuestales. Los pagos que debe hacer **EL PRESTATARIO** en razón de este Contrato de Empréstito, quedan subordinados a las apropiaciones que al efecto se hagan en su presupuesto; por tal motivo **EL PRESTATARIO** se obliga a incluir, mientras existan obligaciones a su cargo derivadas de este empréstito, las partidas necesarias en su presupuesto anual de gastos o realizará las modificaciones presupuestales del caso, y enviará a **EL PRESTAMISTA** anualmente, o cuando corresponda, copia auténtica de la parte pertinente del presupuesto de rentas y gastos, o si fuere necesario, de sus modificaciones, una vez aprobados y expedidos por las autoridades competentes, cuando **EL PRESTAMISTA** así lo requiera.

DECIMA QUINTA. Aplicación de los pagos. Los pagos que se efectúen en desarrollo del presente Contrato de Empréstito, se aplicarán en el siguiente orden: primero a intereses de mora, si los hubiere, segundo a intereses corrientes, tercero a capital y por último al prepago de la obligación.

DECIMA SEXTA. Modificaciones. El presente contrato de empréstito no podrá ser modificado, salvo mutuo acuerdo escrito entre las partes.

DÉCIMA SEPTIMA. **DECLARACIONES** FINALES En virtud de lo dispuesto por la ley 1328 del 2009 EL PRESTATARIO manifiesta 1) que ha leído y revisado todas y cada una de las cláusulas del presente contrato, que conoce y comprende su contenido, cuáles son sus deberes y obligaciones, derechos, los costos y gastos inherentes a la operación de Crédito Público y mediante su suscripción acepta el contenido en su integridad, 2) que los costos, gastos, intereses y tarifas que se causan en desarrollo del presente Contrato de Empréstito le han sido informados por EL PRESTAMISTA, que los conoce y acepta, y que ha tenido la oportunidad de compararlos con los de otras instituciones que ofrecen el servicio financiero al que se refiere el presente Contrato de Empréstito; 3) Que le ha sido suministrada una copia del presente contrato, que ha sido informado sobre los órganos y medios de los que dispone para presentar peticiones, solicitudes o reclamos; 4) y que ha recibido una adecuada educación sobre la manera de instrumentar los productos y/o servicios ofrecidos, sobre el mercado y sobre el tipo de actividad que desarrolla EL PRESTAMISTA. PARAGRAFO: EL PRESTATARIO declara que le han sido informadas las políticas de cobro prejurídico y jurídico de EL PRESTAMISTA, las cuales igualmente se pueden consultar en Internet, dando cumplimiento a lo dispuesto por la Superintendencia Financiera de Colombia.

Para constancia de todo lo anterior se firma en Manizales a los nueve días (9) días del mes de Noviembre del año dos mil quince (2015).

EL PRESTATARIO,

MARTHA LIBIA CASTAÑO TOVAR

C.C. 30.291.119

Segundo Suplente del Gerente

CENTRAL HIDROELÉCTRICA DE CALDAS S.A. E.S.P

NIT 890.800.128-6

EL PRESTAMISTA,

ALEJANDRO ROLDAN PATIÑO CC 75.078.951 GERENTE-SUCURSAL CALDAS BANCO DAVIVIENDA S.A. NIT 860.034.313-

RESOLUCIÓN 3952 de 2015

(Diciembre 29)

Por medio del cual se realizan unos ajustes a la resolución De la Secretaría del Medio Ambiente y se dictan otras disposiciones.

LA SECRETARÍA DE GESTIÓN HUMANA Y SERVICIO A LA CIUDADANÍA.

En ejercicio de sus facultades legales, especialmente las conferidas en el Decreto Nacional 1950 de 1973 y la Ley 909 de 2004, sus Decretos reglamentarios y los Decretos Municipales 883 y 912 de 2015,

CONSIDERANDO:

Que es necesario precisar que la Administración Municipal tiene una Planta Global de Empleos, de acuerdo con lo definido en el artículo 115 de la Ley 489 de 1998 en concordancia con el artículo 209 de la Constitución Política.

Que la condición de planta global que caracteriza la estructura de empleos de la Administración Municipal permite traslados en forma horizontal, respetando su perfil, asignación salarial y la esencia de las funciones del empleo objeto a trasladar.

Que ha dispuesto la Ley que es función del nominador ubicar los cargos en las distintas dependencias de la organización, de conformidad con la estructura orgánica, las necesidades del servicio y los planes y programas que se deban ejecutar.

Que el artículo 19 de la Ley 909 de 2004, "Por la cual se expiden normas que regulan el empleo público, la carrera administrativa, gerencia pública y se dictan otras disposiciones", trata de la estructura del empleo público, definiéndolo como el conjunto de funciones, tareas y responsabilidades que se asignan a una persona y las competencias requeridas para llevarlas a cabo, con el propósito de satisfacer los Planes de Desarrollo y los fines del Estado.

Que los artículos 2.2.5.9.2 y 2.2.5.9.3 del Decreto 1083 de 2015, expresan que un traslado se puede realizar por necesidad del servicio, siempre que ello no implique condiciones menos favorables para el empleado y se trate de empleos con funciones afines, de la misma categoría y con requisitos mínimos similares. Así mismo, el traslado se podrá hacer por necesidad del servicio, siempre que ello no implique condiciones menos favorables para el empleado o cuando sea solicitado por los servidores interesados, siempre que el movimiento no cause perjuicios al servicio

Que el artículo 2.2.12.1 del Decreto 1083 de 2015, "Reformas de las plantas de empleos", expresa que la modificación de la Planta de Empleos debe fundamentarse en necesidades del servicio o en razones de Modernización de la Administración, y con base, entre otros, en los siguientes aspectos: Literal 2: Cambios en la misión y objeto social o en las funciones generales de la Entidad, Literal 3: Traslado de funciones o competencias de un organismo a otro., 4: Supresión fusión o creación de dependencias o modificación de sus funciones, Literal 5: Mejoramiento o introducción de procesos, producción de bienes o prestación de servicios y Literal 10: Mejoramiento de los niveles de eficiencia, eficacia y economía de la Entidad.

Que mediante el Decreto 883 de junio 3 de 2015, se adecúa la Estructura de la Administración Municipal de Medellín, las funciones de sus organismos, dependencias y entidades descentralizadas, se modifican unas entidades descentralizadas y se dictan otras disposiciones.

Que en el mencionado Decreto se creó en la estructura administrativa del nivel central de la Administración Municipal, entre otras, la Secretaría de Medio Ambiente, la cual para cumplir las responsabilidades, funciones, procesos, planes, programas y proyectos a ella asignados, requiere contar con una planta de personal, que permita la operación, integración y el fortalecimiento, para alcanzar los fines del Estado y la ejecución del Plan de Desarrollo, manteniendo lo consagrado en el artículo 115 de la Ley 489 de 1998.

Que ha dispuesto la Ley que es función del nominador ubicar los cargos en las distintas dependencias de la organización, de conformidad con la estructura orgánica, las necesidades del servicio y los planes y programas que se deban ejecutar.

Que mediante el Decreto 912 de junio 5 de 2015, se delegó en la Secretaría de Gestión Humana y Servicio a la Ciudadanía, la administración y control de la planta global de empleos del sector central, la organización con carácter permanente o transitorio de las unidades, equipos y grupos internos de trabajo y la redistribución del personal del Municipio de Medellín del nivel central, con el fin de atender las necesidades del servicio y cumplir con eficacia y eficiencia, los objetivos, políticas y programas de las dependencias centrales.

Que es necesario realizar algunos movimientos al interior de la Secretaría del Medio Ambiente , a fin de apoyar el diseño y la implementación del sistema de información ambiental de Medellin, el sistema de información de alertas tempranas SIATA, el sistema de información Geografico, los indicadores ambientales, proyectos ambientales, centro de gestión Documental ambiental, y los demás componentes que se diseñen y consideren necesarios para su buen funcionamiento, con el fin de conocer de forma oportuna y veraz el estado de los recursos naturales y el ambiente.

Que mediante el Decreto 883 de 2015 se le confiere a la Secretaría de Medio Ambiente las siguientes funciones:

Dirigir y coordinar con las demás dependencias, entidades descentralizadas y autoridades ambientales, la formulación e implementación del Plan de Mitigación y adaptación al cambio climático municipal, teniendo en cuenta los lineamientos del Plan de Ordenamiento Territorial, las políticas establecidas por la entidad y las normas vigentes.

Liderar la implementación de políticas nacionales, regionales y locales de la dimensión ambiental y los recursos naturales renovables con los diferentes actores.

Implementar estrategias y acciones de gestión del recurso hídrico en las cuencas aportantes externas a Medellín, bajo criterios de sostenibilidad urbano regional, promoviendo y viabilizando la conservación de áreas estratégicas para el mantenimiento en cantidad, calidad y regulación del recurso, con el concurso y autorización de los entes territoriales respectivos.

Implementar acciones de reforestación, regeneración natural asistida, turismo ecológico, cercado y amojonamiento, para la protección del recurso hídrico.

Adelantar las acciones tendientes a generar, recuperar, restituir y hacer mantenimiento a la Estructura Ecológica Principal.

Participar en la formulación e implementación de los instrumentos complementarios al Plan de Ordenamiento Territorial referidos a la protección y conservación de los recursos naturales.

Participar en la identificación de paisajes culturales y su potencial inclusión en el Subsistema de Patrimonio Cultural Inmueble.

Gestionar con las dependencias competentes la adquisición de predios, en pro de la conservación, preservación y recuperación de los recursos naturales e incorporar la Política Nacional para Humedales Interiores de Colombia, establecida en Ley 357 de 1997 o en la que la modifique o sustituya.

Impulsar el sostenimiento, operación, desarrollo y fortalecimiento técnico, científico y social del Sistema de Alerta Temprana de Medellín y su región vecina –SIATA, en coordinación con el DAGRD.

Promover las disposiciones legales, necesarias para la vigilancia, la preservación y la defensa de patrimonio ecológico del municipio.

Dirigir la formulación, ejecución y seguimiento del Plan Ambiental Municipal (PAM) en armonía con las directrices establecidas por planeación municipal y el Ministerio de Ambiente y Desarrollo Sostenible.

Liderar la formulación, revisión, concertación y seguimiento de los asuntos ambientales del Plan de Ordenamiento Territorial.

Dirigir el Sistema de Gestión Ambiental de Medellín (SIGAM), como herramienta estratégica para la armonización de la gestión ambiental.

Adoptar los planes, programas y proyectos para la conservación, recuperación y protección de los recursos naturales renovables y mejoramiento de las condiciones ambientales, así como liderar la elaboración de los planes que de acuerdo a la competencia corresponda al Municipio de Medellín en materia ambiental.

Liderar el Sistema de Información Ambiental de Medellín (Siamed).

Colaborar con la Corporación Autónoma Regional y el Área Metropolitana del Valle de Aburrá, en la elaboración de los planes regionales y en la ejecución de programas, proyectos y acciones necesarias para la conservación de los recursos naturales renovables y el mejoramiento de las condiciones ambientales.

Actuar con sujeción a la distribución legal de competencias para que se cumplan las funciones de control y vigilancia del medio ambiente y los recursos naturales renovables, asignadas a las instancias municipales.

Dirigir las actividades relacionadas con el manejo integral del Sistema Orográfico Municipal, según lo dispuesto en el Plan de Ordenamiento Territorial.

Liderar la consecución de recursos externos del orden internacional, nacional y regional, con el fin de cumplir con los objetivos de la Secretaría.

Definir las políticas, planes y programas tendientes a la investigación, para la conservación, mejoramiento, promoción, valoración y uso sostenible de los recursos naturales renovables y los servicios ambientales del Municipio.

Definir y concertar los planes, programas y proyectos de formación y educación para mejorar la gestión ambiental y el uso responsable de recursos naturales renovables y fomentar una cultura de respeto por el patrimonio natural.

Planear, presupuestar y coordinar con la Secretaría de Infraestructura las necesidades de dotación de equipamientos, identificando los requerimientos de infraestructura física para los diferentes programas y proyectos que deban ser priorizados, implementados y ejecutados en el marco del Plan de Desarrollo y su articulación con el Plan de Ordenamiento Territorial y sus instrumentos complementarios.

Coordinar con la Secretaría de Suministros y Servicios, la realización de los contratos necesarios para obtener los bienes y servicios requeridos y ejercer la supervisión técnica de los mismos.

Implementar y ejecutar las políticas y directrices que orienten los procesos de apoyo.

Implementar y ejecutar las políticas y directrices que establezca la Secretaría de Gestión Humana y Servicio a la Ciudanía, en materia de gestión de trámites, procedimientos administrativos, peticiones, quejas, reclamos y sugerencias.

Implementar y ejecutar las políticas y directrices que orienten el Plan de Ordenamiento Territorial.

Implementar, mantener y mejorar el Sistema Integral de Gestión para los procesos que dirige o participa, de acuerdo con las directrices definidas por la entidad.

La Subsecretaría Gestión Ambiental tendrá las siguientes funciones:

Dirigir y articular las políticas, programas y proyectos del orden nacional y local y el Sistema de Gestión Ambiental de Medellín (SIGAM), con las diversas dependencias de la Administración municipal y sus entes descentralizados, de acuerdo con las normas vigentes.

Liderar la formulación del componente ambiental del Plan de Desarrollo de Medellín y del Plan de Acción y Presupuesto por Resultados de la Secretaría del Medio Ambiente, articulados con el Plan Ambiental de Medellín (PAM).

Participar en la formulación, revisión y concertación del componente ambiental del Plan de Ordenamiento Territorial.

Implementar y hacer seguimiento en coordinación con las autoridades ambientales al Plan de Mitigación y adaptación al cambio Climático Municipal y a las políticas definidas para tal fin de acuerdo con las disposiciones legales.

Liderar el diseño y la implementación del Sistema de Información Ambiental de Medellín, con el fin de conocer de forma oportuna y veraz el estado de los recursos naturales y el ambiente, así como la gestión ambiental del municipio, para orientar la toma de decisiones e informar a la comunidad.

Liderar la formulación, ejecución y seguimiento del Plan Ambiental de Medellín (PAM), en armonía con las políticas ambientales.

Coordinar e implementar el Sistema de Gestión Ambiental de Medellín (SIGAM), como herramienta estratégica, para la armonización de la gestión ambiental municipal.

Liderar la implementación de políticas, planes y programas de investigación, para la conservación, mejoramiento, promoción, valoración y uso sostenible de los recursos naturales renovables y servicios ambientales del municipio.

Fomentar y/o desarrollar estudios e investigaciones relacionados con la dimensión ambiental - urbano rural - que propendan por el manejo adecuado de los recursos naturales y del medio ambiente.

Establecer los lineamientos que orienten la gestión de la educación ambiental mediante acciones conjuntas y coordinadas con los diversos actores que tengan responsabilidades y competencias en la educación y la cultura ambiental.

Administrar y fortalecer el Sistema de Información Ambiental de Medellín (Siamed).

Promover y fortalecer en los medios de comunicación social, el desarrollo de estrategias de formación de actitudes y valores de protección y recuperación ambiental de los recursos naturales, acorde con las políticas nacionales.

Realizar el seguimiento al estado de los recursos naturales y del ambiente como herramienta para la revisión y ajuste del Plan Ambiental de Medellín.

Realizar acompañamiento técnico y ambiental a la actividad empresarial e industrial para la implementación de medidas de producción y consumo sostenible.

Coordinar y ejecutar los proyectos y acciones necesarias para mejorar las condiciones de la fauna, mediante actividades de educación y el fomento de la protección, con el fin de evitar el maltrato e inadecuada tenencia de los animales, en cumplimiento de la Política Pública de Protección Animal del Municipio de Medellín.

Diseñar, coordinar y ejecutar los proyectos relacionados con la gestión integral de los residuos sólidos de acuerdo a la normativa vigente, para apoyar el direccionamiento de las políticas de manejo de residuos en la ciudad.

Diseñar y coordinar las estrategias de mejoramiento de la calidad del aire y la prevención y corrección de la contaminación auditiva, visual y electro magnética, así como establecer la redes de monitoreo respectivos.

Implementar y ejecutar las políticas y directrices que orienten los procesos de apoyo.

Implementar y ejecutar las políticas y directrices que establezca la Secretaría de Gestión Humana y Servicio a la Ciudanía, en materia de gestión de trámites, procedimientos administrativos, peticiones, quejas, reclamos y sugerencias.

Implementar y ejecutar las políticas y directrices que orienten el Plan de Ordenamiento Territorial en el ámbito de sus funciones.

Implementar, mantener y mejorar el Sistema Integral de Gestión para los procesos que dirige o participa, de acuerdo con las directrices definidas por la entidad.

La Subsecretaria de Recursos Naturales Renovables, tendrá las siguientes funciones:

Definir y ejecutar los planes, programas y proyectos ambientales relacionados con los recursos naturales, que garanticen la conservación, recuperación y protección de los recursos naturales renovables.

Ejecutar acciones que garanticen la conservación de los recursos naturales en el Municipio de Medellín, orientados a la protección de la flora y fauna silvestre, aislamiento y reforestación de áreas protectoras y demás acciones que así lo requieran.

Liderar las acciones que garanticen el manejo hidráulico de las corrientes, minimizando el impacto que se pudiera generar por avenidas torrenciales de las quebradas que hacen parte del recurso hídrico del Municipio de Medellín.

Desarrollar las acciones necesarias para el mejoramiento de la calidad de los recursos naturales y fomentar la apropiación de los espacios públicos verdes.

Gestionar y coordinar las acciones necesarias para la mitigación de riesgos asociados a eventos que pudieran presentarse en las quebradas del Municipio de Medellín.

Articular con las entidades competentes, la política de gestión estratégica del ciclo del agua como recurso natural, bien público y elemento de efectividad del derecho a la vida.

Realizar la identificación, priorización técnica y ambiental para gestionar con la dependencia competente la compra de predios que surten acueductos para la protección de microcuencas, de acuerdo con la normativa vigente.

Mantener actualizado el inventario de la red hídrica del municipio de Medellín en coordinación con el Departamento Administrativo de Planeación, con el fin de garantizar unidad de criterios técnicos en las actuaciones de las diversas dependencias del municipio.

Realizar la gestión ambiental de los Cerros Tutelares de Medellín, en aras de dar cumplimiento a sus planes de manejo.

Liderar y gestionar las acciones necesarias para la protección, conservación y fomento de los espacios públicos verdes, paisajismo y silvicultura del Municipio de Medellín.

Formular, implementar y coordinar, con visión integral, la política de conservación, aprovechamiento y desarrollo sostenible de las áreas protegidas del Municipio.

Implementar acciones en cumplimiento de los planes de descontaminación de aire y de ruido.

Implementar y ejecutar las políticas y directrices que orienten los procesos de apoyo.

Implementar y ejecutar las políticas y directrices que establezca la Secretaría de Gestión Humana y Servicio a la Ciudanía, en materia de gestión de trámites, procedimientos administrativos, peticiones, quejas, reclamos y sugerencias.

Implementar y ejecutar las políticas y directrices que orienten el Plan de Ordenamiento Territorial en el ámbito de sus funciones.

Implementar, mantener y mejorar el Sistema Integral de Gestión para los procesos que dirige o participa, de acuerdo con las directrices definidas por la entidad.

Parágrafo. La Secretaría de Medio Ambiente no ejecutará en forma directa labores de construcción y sostenimiento de obra pública, las cuales serán realizadas exclusivamente mediante contratación pública con terceros especializados.

Que de acuerdo al proceso de optimización de la modernización y según lo consagrado en el Decreto 883 de 2015, se adopta en la Secretaría de Medio Ambiente la planta global de empleos mediante el Resolución 002068 de julio 09 de 2015.

Que con el propósito de fortalecer los procesos de la Secretaría de Medio Ambiente, se hace necesario adoptar la planta de empleos de la dependencia.

Que mediante Resolución 3096 de Octubre 26 de 2015 se conformaron las unidades y equipos internos en la Secretaría de Medio Ambiente.

Que la conformación de las Unidades y los Equipos de Trabajo descongestionan la gestión administrativa y la evaluación del desempeño de los Despachos a fin de puedan direccionar y controlar de manera adecuada el proceso asignado a su dependencia.

Que de acuerdo a la nueva organización al interior de la Secretaría y la conformación de unidades y/o equipos de trabajo, es necesario realizar algunos ajustes a las funciones, a la vez realizar movimientos a fin de hacer más ágiles y eficientes las actividades desarrolladas en los equipos.

Que consecuente con todo lo anterior, y en mérito de lo expuesto

RESUELVE:

Artículo 1º: Ajustar las funciones de la Unidad Administrativa Adscrita al Despacho de la Secretaría de Medio Ambiente la cual quedara de la siguiente manera:

La Unidad Administrativa adscrita al Despacho de la Secretaría de Medio Ambiente tendrá como objetivo básico el siguiente:

Coordinar y controlar los procesos administrativos, de apoyo logístico, de comunicaciones y las relaciones

interinstitucionales incluidos entes de control y demás actividades administrativas requeridas para la ejecución de los programas y proyectos de la dependencia, así mismo las acciones necesarias para asegurar oportunidad y calidad en los servicios, de transporte, dotación, suministros, y demás servicios internos, requeridos para la ejecución de las actividades propias de la dependencia.

Consecuente con el objetivo básico, realizará entre otras, las siguientes responsabilidades:

Coordinar la formulación, ejecución, seguimiento y control a los procesos de apoyo administrativo, financiero, logístico, jurídico, de infraestructura para el funcionamiento de la Secretaía del Medio Ambiente, de contratación y demás requeridos para la operación de programas y proyectos de la Dependencia garantizando oportunidad y calidad para la ejecución de las actividades.

Contribuir al cumplimiento de políticas, programas y proyectos establecidos en el Plan de Desarrollo Municipal vigente para cada periodo de gobierno, a través de la aplicación del Estatuto contractual Vigente y sus Decretos Reglamentarios en desarrollo de los procesos de contratación que adelante la Secretaría del Medio Ambiente.

Prestar apoyo técnico y Administrativo a las Subsecretarías de la Secretaría del Medio Ambiente, con el fin de adelantar procesos contractuales con eficiencia y eficacia.

Adelantar las Gestiones internas y externas requeridas para atender de manera oportuna los requerimientos y solicitudes de los entes de control con oportunidad y calidad

Coordinar con otras dependencias del orden municipal y entidades públicas y privadas los aspectos administrativos relacionadas con la ejecución de programas y proyectos tendientes al mejoramiento de la gestión ambiental del Municipio de Medellín

Realizar seguimiento al cumplimiento y ejecución del presupuesto de la Secretaría, mediante la implementación de actividades de control y elaboración de informes de seguimiento periódicos al Plan de Compras.

Promover y fortalecer en los medios de comunicación social, el desarrollo de estrategias de formación de actitudes y valores de protección y recuperación ambiental de los recursos naturales, acorde con las políticas nacionales.

Coordinar la atención oportuna de los requerimientos, citaciones y convocatorias que realicen el ente de control político, las veedurías ciudadanas y las organizaciones comunitarias.

Aplicar los mecanismos de administración del personal de la Secretaría y hacer las recomendaciones del caso, adoptando las políticas y directrices que en materia de manejo de personal establezca el Municipio.

Coordinar el servicio de transporte, de acuerdo con los programas y necesidades de las diferentes dependencias de la Secretaría del Medio Ambiente, procurando que éste se preste en forma eficiente y oportuna.

Artículo 2º: Trasladar el empleo Código 40701030 Posición 2005438 adscrito Unidad Administrativa, Secretaría de Medio Ambiente para que dependa directamente de la Unidad de Educación y Buenas Prácticas Ambientales Subsecretaría de Gestión Ambiental, Secretaría del Medio Ambiente.

Artículo 3º: Trasladar la servidora SANDRA MARCELA TOVAR ACERO identificada con cédula de ciudadanía número 35.528.171 quien se encuentra en provisionalidad en el empleo Auxiliar Administrativo Código 40701030 posición 2005438 adscrito a la Unidad Administrativa, Secretaría de Medio Ambiente y a su titular MONICA MARIA ZAPATA RUIZ identificada con cédula de ciudadanía número 43.522.07 quien se encuentra en encargo en otro empleo para que dependan directamente de la Unidad de Educación y Buenas Prácticas Ambientales Subsecretaría de Gestión Ambiental, Secretaría del Medio Ambiente.

Artículo 4º: Ajustar las funciones de la Unidad de Educación y Buenas Prácticas Ambientales adscrita al Despacho de la Subsecretaría de Gestión Ambiental, Secretaría del Medio Ambiente la cual quedara de la siguiente manera:

La Unidad de Educación y Buenas Prácticas Ambientales adscrita al Despacho de la Subsecretaría de Gestión Ambiental, Secretaría del Medio Ambiente, tendrá como objetivo básico el siguiente:

Coordinar y orientar la gestión de la educación ambiental mediante acciones conjuntas coordinadas con los diversos actores que tengan responsabilidades y competencias en la educación y la cultura ambiental.

Consecuente con el objetivo básico, realizará entre otras, las siguientes responsabilidades:

Promover la mejora continua de las actividades industriales en relación con el ambiente mediante, la reducción de las pérdidas sistemáticas o accidentales de materiales, en forma de contaminantes (residuos, emisiones o vertidos), y de esta manera aumentar la productividad centrándose principalmente en factores humanos y ambientales de la producción

Realizar acompañamiento técnico y ambiental a la actividad empresarial e industrial para la implementación de medidas de producción y consumo sostenible.

Coordinar y ejecutar los proyectos y acciones necesarias para mejorar las condiciones de la fauna, mediante actividades de educación y de fomento de la protección, con el fin de evitar el maltrato e inadecuada tenencia de los animales, en cumplimiento de la Política Pública de Protección Animal del Municipio de Medellín.

Diseñar y coordinar las estrategias de mejoramiento de la calidad del aire y la prevención y corrección de la contaminación auditiva, visual y electromagnética, así como establecer las redes de monitoreo respectivos.

Fortalecer la educación en tenencia responsable de animales, atención de animales en situación de calle, sensibilización para la entrega en adopción de animales de compañía, administración del sistema de registro de información.

Promover la educación ambiental a través de la escuela ambiental y escuela para servidores municipales, articulado con estrategias como aulas ambientales, mesas ambientales, procedas y semilleros ambientales; estrategias educativas en calidad del aire y ruido, puntos limpios, puntos críticos asociados a residuos y a quebradas. Educación en uso eficiente del agua, la energía y los residuos sólidos, en hogares y empresas; huertas urbanas y rurales, huertos leñeros y estufas eficientes; seguimiento a la implementación de la guía de manejo socio ambiental en obra.

Artículo 5º: Trasladar el empleo Profesional Universitario Código 21902295 Posición 2000592 adscrito al Equipo de Planeación Ambiental SIGAM, Subsecretaría de Gestión Ambiental, Secretaría del Medio Ambiente, para que dependa directamente de la Unidad de Educación y Buenas Prácticas Ambientales, Subsecretaría de Gestión Ambiental, Secretaría del Medio Ambiente,

Artículo 6º: Trasladar el servidor JUAN GUILLERMO MEJIA RIVAS identificado con cédula de ciudadanía número 71.648.609 quien se encuentra en encargo en el empleo Profesional Universitario código 21902295 posición 2000592 adscrito al Equipo de Planeación Ambiental SIGAM, Subsecretaría de Gestión Ambiental, Secretaría del Medio Ambiente y a su titular MARIA MAGDALENA GONZALEZ CORREA identificada con cédula de ciudadanía número 39.267.787 quien se encuentra en encargo en otro empleo para que dependa directamente de la Unidad de Educación y Buenas Prácticas Ambientales Subsecretaría de Gestión Ambiental, Secretaría del Medio Ambiente.

Artículo 7º: Ajustar las funciones de**l Equipo de Residuos Sólidos** adscrito a la Unidad de Buenas Prácticas ambientales, Subsecretaría de Gestión Ambiental, Secretaría del Medio Ambiente la cual quedara de la siguiente manera:

El Equipo de Residuos Sólidos adscrito a la Unidad de Buenas Prácticas ambientales, tendrá como objetivo básico:

Diseñar, coordinar y ejecutar los proyectos relacionados con la gestión integral de los residuos sólidos de acuerdo a la normativa vigente, para apoyar el direccionamiento de las políticas de manejo de residuos en la ciudad.

Consecuente con el objetivo básico, realizará entre otras, las siguientes responsabilidades:

Ejecutar los proyectos priorizados en el PGIRS y relacionados con la GIRS.

Acompañamiento a recicladores articulado a los centros de acopio y compraventas de material reciclable urbanos y rurales; gestión de residuos de construcción y demolición; PMIRS en los sectores institucional, residencial y comercial; seguimiento a puntos limpios.

Desarrollar acciones de educación, capacitación y participación para una gestión adecuada de los residuos sólidos eficiente, eficaz y sostenible.

Desarrollar procedimientos y políticas que generen un adecuado manejo de los residuos sólidos que permitan realizar una gestión ambiental y económicamente adecuada.

Adoptar medidas de minimización de residuos sólidos, a través de la reducción de sus volúmenes de generación y características de peligrosidad.

Promover la iniciativa y participación activa de la población, la sociedad civil organizada, y el sector privado en la adecuada gestión de los residuos sólidos.

Artículo 8º: Trasladar el empleo Profesional Universitario código 21902212 posición 2017086 del Equipo de Soluciones Hidráulicas, Unidad de Ordenamiento del Recurso Hídrico Subsecretaría de Recursos Naturales Renovables, Secretaría de Medio Ambiente, para que dependa directamente Equipo de Residuos Sólidos, Unidad de Buenas Prácticas Ambientales, Subsecretaría de Gestión Ambiental, Secretaría de Medio Ambiente

Artículo 9º: Trasladar el servidor YESID ALEXIS RAMIREZ VALENCIA identificado con cédula de ciudadanía número 8.161.279 quien se encuentra en provisionalidad en el empleo Profesional Universitario código 21902212 posición 2017086 del Equipo de Soluciones Hidráulicas, Unidad de Ordenamiento del Recurso Hídrico Subsecretaría de Recursos Naturales Renovables, Secretaría de Medio Ambiente, para que dependa directamente Equipo de Residuos Sólidos, Unidad de Buenas Prácticas Ambientales, Subsecretaría de Gestión Ambiental, Secretaría de Medio Ambiente.

Artículo 10°: Ajustar las funciones de**l Equipo de Planeación Ambiental SIGAM** adscrito al Despacho de la Subsecretaría de Gestión Ambiental, Secretaría del Medio Ambiente quedara de la siguiente manera:

Equipo de Planeación Ambiental SIGAM adscrito al Despacho de la Subsecretaría de Gestión Ambiental, Secretaría del Medio Ambiente, tendrá como objetivo básico:

Coordinar e implementar el Sistema de Gestión Ambiental de Medellín – SIGAM, como herramienta estratégica para la armonización de la Gestión Ambiental Municipal.

Consecuente con el objetivo básico, realizará entre otras, las siguientes responsabilidades:

Fortalecer la Planeación, la gestión interinstitucional y desarrollar programas y proyectos ambientales, a través de los principales instrumentos que guían la acción ambiental de Medellín, como son: El Sistema de Gestión Ambiental de Medellín –Sigam- y el Plan Ambiental de Medellín -PAM- orientados al cumplimiento de la Política Ambiental establecida para Medellín, amparada en la normatividad vigente

Liderar el comité de Aseo y Ornato del Municipio con el propósito de articular las instancias de la Administración Municipal para intervenir los puntos críticos, lograr intervenciones integrales, y generar acciones de educación ambiental y recuperar espacios públicos para el disfrute de toda la ciudadanía.

Liderar la formulación, ejecución y seguimiento del componente ambiental del Plan de Desarrollo de Medellín y del Plan de Acción y Presupuesto por resultados en lo que tiene que ver con de la Secretaría del Medio Ambiente, articulados con el Plan Ambiental de Medellín – PAM.

Participar en la formulación, revisión y concertación del componente ambiental del Plan de Ordenamiento Territorial

Dirigir y articular las políticas, programas y proyectos del orden Nacional y Local y el Sistema de Gestión Ambiental de Medellín - Sigam, con las diversas dependencias de la Administración Municipal y sus entes descentralizados, de acuerdo con las normas vigentes.

Fomentar y/o desarrollar estudios e investigaciones relacionados con la dimensión ambiental - urbano rural - que propendan por el manejo adecuado de los recursos naturales y del medio ambiente.

Acompañamiento a formulación y seguimiento a los planes, programas y políticas ambientales, como el Plan de Ordenamiento Territorial (POT), Plan Ambiental de Medellin (PAM), y el Plan de Desarrollo Municipal (PDM).

Coordinación del Programa de Planeación Local y Presupuesto Participativo, en referente a la Secretaría del Medio Ambiente, específicamente en la planeación y el seguimiento a la ejecución de los componentes mediambientales

Coordinación del funcionamiento de la estructura operativa del Sigam (CAM, Grupo Líder, Comités Temáticos), y acompañamiento al proceso de Calidad SIG de la Secretaría de Medio Ambiente.

Apoyar la identificación y consolidación de la información requerida para los informes de seguimiento al cumplimiento y ejecución del Plan de Desarrollo, Plan de Acción de la Secretaría de Medio Ambiente,así como la consolidación y elaboración de la información estadística e indicadores asociados a los mismos.

Artículo 11º: Trasladar el empleo Profesional Universitario código 21902344 posición 2005380 de la Unidad de Educación y Buenas Prácticas Ambientales, Despacho de la Subsecretaría de Gestión Ambiental, Secretaría del Medio Ambiente, para que dependa directamente del Equipo de Planeación Ambiental SIGAM, Subsecretaría de Gestión Ambiental, Secretaría de Medio Ambiente.

Artículo 12º: Trasladar la servidora GLORIA INES BENJUMEA PEREZ identificada con cédula de ciudadanía número 32.319.750 quien es titular del empleo Profesional Universitario código 21902344 posición 2005380 de la Unidad de Educación y Buenas Prácticas Ambientales, Despacho de la Subsecretaría de Gestión Ambiental, Secretaría del Medio Ambiente, para que dependa directamente del Equipo de Planeación Ambiental SIGAM Subsecretaría de Gestión Ambiental, Secretaría de Medio Ambiente

Artículo 13º: Trasladar el empleo Profesional Universitario código 21902324 posición 2016512 de la Unidad de Educación y Buenas Prácticas Ambientales, Subsecretaría de Gestión Ambiental, Secretaría del Medio Ambiente, para que dependa directamente del Equipo de Planeación Ambiental SIGAM, Subsecretaría de Gestión Ambiental, Secretaría de Medio Ambiente.

Artículo 14º: Trasladar el servidor HECTOR JULIO CANO JARAMILLO identificado con cédula de ciudadanía número 71.386.959 quien se encuentra en provisionalidad en el empleo Profesional Universitario código 21902324 posición 2016512 de la Unidad de Educación y Buenas Prácticas Ambientales, Subsecretaría de Gestión Ambiental, Secretaría del Medio Ambiente, para que dependa directamente del Equipo de Planeación Ambiental SIGAM, Subsecretaría de Gestión Ambiental, Secretaría de Medio Ambiente.

Artículo 15°: Trasladar el empleo Técnico Administrativo código 36702132 posición 2005381 adscrito a la Unidad Administrativa, Secretaría del Medio Ambiente, para que dependa directamente del Equipo de Planeación Ambiental SIGAM, Subsecretaría de Gestión Ambiental, Secretaría de Medio Ambiente

Artículo16º: Trasladar la servidora ALEYDA DE LAS MERCEDES OSORNO ALVAREZ identificada con cédula de ciudadanía número 42.777.454 quien es titular del empleo Técnico Administrativo código 36702132 posición 2005381 adscrito a la Unidad Administrativa, Secretaría del Medio Ambiente, para que dependa directamente del Equipo de Planeación Ambiental SIGAM, Subsecretaría de Gestión Ambiental, Secretaría de Medio Ambiente.

Artículo 17°: Ajustar las funciones de**l Equipo de Gestión de la Información** adscrito al Despacho de la Subsecretaría de Gestión Ambiental, Secretaría del Medio Ambiente quedara de la siguiente manera:

El Equipo de Gestión de la Información, adscrito a la despacho de la Subsecretaria de tendrá como objetivo básico:

Liderar, administrar y fortalecer el diseño y la implementación del Sistema de Información Ambiental de Medellín, con el fin de conocer de forma oportuna y veraz el estado de los recursos naturales renovables y el ambiente, así como la gestión ambiental del Municipio, para orientar la toma de decisiones e informar a la comunidad.

Consecuente con el objetivo básico, realizará entre otras, las siguientes responsabilidades:

Liderar el diseño e implementación de los Componentes del Sistema de Información Ambiental, que permita determinar las potencialidades y limitaciones de los recursos naturales, de manera que sirvan de insumo para planificar su ordenamiento y manejo desde la Secretaría.

Coordinar la elaboración y presentación de los informes de los componentes del Sistema de Información Ambiental, con el fin de conocer de forma oportuna y veraz el estado de los recursos naturales y el ambiente, así como la gestión ambiental del municipio, para orientar la toma de decisiones.

Coordinar, apoyar y participar en las diferentes investigaciones de tipo estadístico que se desarrollan en la Secretaría de Medio Ambiente en cumplimiento de su misión, de tal forma que cumplan con el rigor técnico y científico que exigen este tipo de estudios.

Coordinar el centro de información e investigaciones ambientales de Medellin, Sistema de Infornación Geografica, la oficina de soporte SIAMED y realizar el mantenimiento y mejoras permanentes al sistema de información ambiental para Medellin y los susbsistemas implementados, por implementar y gestión de las plataformas del OAM y SIATA.

Artículo 18º: Trasladar el empleo Profesional Universitario código 21902298 posición 2005312 de la Unidad de Estructura Ecológica, Subsecretaría Recursos Naturales Renovables, Secretaría de Medio Ambiente, para que dependa directamente del Equipo de estudios y diseños, Unidad de Ordenamiento del Recurso Hídrico Subsecretaría de Recursos Naturales Renovables, Secretaría de Medio Ambiente.

Artículo 19°: Trasladar la servidora YULI DAYANY RODRIGUEZ DAZA identificada con cédula de ciudadanía numero 43.996.610 quien se encuentra en provisionalidad en el empleo Profesional Universitario código 21902298 posición 2005312, y su titluar SANDOVAL HERNANDEZ ANA CATALINA identificada con cédula de ciudadanía numero 43.755.118 que se encuentra en encargo en otro empleo, de la Unidad de Estructura Ecológica, Subsecretaría Recursos Naturales Renovables, Secretaría de Medio Ambiente, para que dependa directamente del Equipo de estudios y diseños, Unidad de Ordenamiento

del Recurso Hídrico Subsecretaría de Recursos Naturales Renovables, Secretaría de Medio Ambiente

Artículo 20°: Trasladar el empleo Profesional Universitario código 21902356 posición 2016517 del Equipo de estudios y diseños adscrito a la Unidad de Ordenamiento del Recurso Hídrico, Subsecretaría de Recursos Naturales Renovables, Secretaría de Medio Ambiente, para que dependa directamente de la Unidad de Estructura Ecologica, Subsecretaría de Recursos Naturales Renovables, Secretaría de Medio Ambiente.

Artículo 21º: Trasladar la servidora VILLA GRAJALES ANA MARIA identificada con cédula de ciudadanía numero 43.252.423 quien se encuentra en provisionalidad en el empleo Profesional Universitario código 21902356 Posición 2016517 del Equipo de estudios y diseños adscrito a la Unidad de Ordenamiento del Recurso Hídrico Subsecretaría de Recursos Naturales Renovables, Secretaría de Medio Ambiente, para que dependa directamente de la Unidad de Estructura Ecologica, Subsecretaría de Recursos Naturales Renovables, Secretaría de Medio Ambiente.

Artículo 22º: Trasladar el empleo Profesional Universitario código 21902212 posición 2017124 del Equipo de residuos sólidos, Unidad de Buenas Prácticas ambientales Subsecretaría de Gestón Ambiental, Secretaría de Medio Ambiente, para que dependa directamente del equipo de soluciones Hidraulicas, Unidad de Ordenamiento del Recurso Hídrico Subsecretaría de Recursos Naturales Renovables.

Artículo 23º: Trasladar la servidora GARCIA ARCILA NELIDA AMPARO identificada con cédula de ciudadanía numero 43.501.311 quien se encuentra en provisionalidad en el empleo Profesional Universitario código 21902212 posición 2017124 del Equipo de residuos sólidos, Unidad de Buenas Prácticas ambientales Subsecretaría de Gestón Ambiental, Secretaría de Medio Ambiente, para que dependa directamente del equipo de soluciones Hidraulicas, Unidad de Ordenamiento del Recurso Hídrico Subsecretaría de Recursos Naturales Renovables..

Artículo 24°: Las Unidades, Equipos, y servidores adscritos a la Secretaría de Medio Ambiente no mencionados en el presente Acto Administrativo, continuarán en las mismas Dependencias y con las mismas responsabilidades.

Artículo25º: Los manuales específicos de funciones y competencias laborales de los empleos contemplados en esta resolución, serán los existentes para estos empleos en el Municipio de Medellín.

Artículo 26º: La presente resolución rige a partir de la fecha de su publicación en la gaceta oficial y deroga todas las disposiciones que le sean contrarias.

PUBLÍQUESE Y CÚMPLASE

Dado en Medellín, a los veintinueve (29) días del mes diciembre de 2015

ALEJANDRA VANEGAS VALENCIA

RESOLUCIÓN 3635 DE 2015 (Diciembre 4)

Por medio del cual se modifican unos Actos Administrativos en el Municipio de Medellín

LA SECRETARÍA DE GESTIÓN HUMANA Y SERVICIO A LA CIUDADANÍA

En ejercicio de sus facultades legales, especialmente las conferidas en el artículo 315 de la Constitución Política, el artículo 29 de la Ley 1551 de 2012, el Decreto Nacional 1083 de 2015, la Ley 909 de 2004 y sus Decretos reglamentarios y los Decretos Municipales 883 y 912 de 2015, y

CONSIDERANDO:

Que la Administración Municipal tiene una Planta Global de Empleos, de acuerdo con lo definido en el artículo 115 de la Ley 489 de 1998, en concordancia con el artículo 209 de la Constitución Política.

Que la condición de planta global que caracteriza la estructura de empleos de la Administración Municipal, permite realizar traslados en forma horizontal, respetando su perfil, asignación salarial y la esencia de las funciones del empleo objeto a trasladar, de acuerdo con lo establecido en los artículos 2.2.5.9.2. y 2.2.5.9.3. del Decreto Nacional 1083 de 2015.

Que es función del nominador ubicar los cargos en las distintas dependencias de la organización, de conformidad con la estructura orgánica, las necesidades del servicio y los planes y programas que se deban ejecutar.

Que el artículo 19 de la Ley 909 de 2004, "Por la cual se expiden normas que regulan el empleo público, la carrera administrativa, gerencia pública y se dictan otras disposiciones", trata de la estructura del empleo público, definiéndolo como el conjunto de funciones, tareas y responsabilidades que se asignan a una persona y las competencias requeridas para llevarlas a cabo, con el propósito de satisfacer los Planes de Desarrollo y los fines del Estado.

Que los artículos 2.2.5.9.2. y 2.2.5.9.3. del Decreto Nacional 1083 de 2015, expresan que un traslado se puede realizar por necesidad del servicio, siempre que ello no implique condiciones menos favorables para el empleado y se trate de empleos con funciones afines, de la misma categoría y con requisitos mínimos similares. Así mismo, el traslado se podrá hacer por necesidad del servicio, siempre que ello no implique condiciones menos favorables para el empleado o cuando sea solicitado por los servidores interesados, siempre que el movimiento no cause perjuicios al servicio.

Que el artículo 2.2.12.1. del Decreto 1083 de 2015, "Reformas de las plantas de empleos", expresa que la modificación de la Planta de Empleos debe fundamentarse en necesidades del servicio o en razones de Modernización de la Administración, y con base, entre otros, en los siguientes aspectos: Literal 2: Cambios en la misión y objeto social o en las funciones generales de la Entidad, Literal 3: Traslado de funciones o competencias de un organismo a otro., 4: Supresión fusión o creación de dependencias o modificación de sus funciones, Literal 5: Mejoramiento o introducción de procesos, producción de bienes o prestación de servicios y Literal 10: Mejoramiento de los niveles de eficiencia, eficacia y economía de la Entidad.

Que mediante el Decreto 883 de junio 3 de 2015, se adecúa la Estructura de la Administración Municipal de Medellín, las funciones de sus organismos, dependencias y entidades descentralizadas, se modifican unas entidades descentralizadas y se dictan otras disposiciones.

Que mediante el Decreto 912 de junio 5 de 2015, se delegó en la Secretaría de Gestión Humana y Servicio a la Ciudadanía la administración y control de la planta global de empleos del sector central, la organización con carácter permanente o transitorio de las unidades, equipos y grupos internos de trabajo y la redistribución del personal del Municipio de Medellín del nivel central, con el fin de atender las necesidades del servicio y cumplir con eficacia y eficiencia, los objetivos, políticas y programas de las dependencias centrales.

Que con el proceso de implementación de la optimización a la modernización, se expidieron entre otras, las Resoluciones 2017, 2021, 2022, 2043, 2046, 2061, 2062, 2063, 2064, 2067, 2068, 2069 y 2098 de julio de 2015, mediante las cuales se conformaron las plantas de empleos de las Secretarías de Salud, Hacienda, Las Mujeres, Gobierno y Derechos Humanos, Privada, Infraestructura Física, Seguridad, Departamento Administrativo de Gestión del Riesgo de Desastres, Cultura Ciudadana, Inclusión Social y Familia, Medio ambiente, Educación y Hacienda, respectivamente.

Que mediante Resolución 2066 de 2015, se adoptó la Planta de Empleos de la Secretaría General y se trasladaron de las dependencias en mención para esa Secretaría, varias plazas de empleos cuyo perfil contempla la formación disciplinar en Derecho, a fin de fortalecer el direccionamiento jurídico de la entidad.

Que como consecuencia de lo anterior, los servidores que fueron trasladados para la Secretaría General no fueron incluidos en las plantas de empleos de las Secretarías de origen.

Que no obstante lo anterior y teniendo en cuenta que se viene adelantando el proceso de implementación de la nueva estructura de la Administración Municipal, en el que es necesario aprovechar el conocimiento y experticia

de los servidores en sus respectivos procesos, se evaluó nuevamente la necesidad del servicio en cuanto al apoyo que debe brindarse al proceso de Gestión Jurídica, y se determinó no realizar la notificación de la Resolución 2066 de 2015 a los servidores que serían trasladados.

Que en el Artículo 1° de la Resolución 3218 del 11 de noviembre de 2015, se derogó la Resolución 2066 de 2015, mediante la cual se realizó el traslado de varias plazas de empleos del nivel profesional con sus respectivos servidores, de las Secretarías de Salud, Hacienda, Las Mujeres, Gobierno y Derechos Humanos, Privada, Infraestructura Física, Seguridad, Departamento Administrativo de Gestión del Riesgo de Desastres, Cultura Ciudadana, Inclusión Social y Familia, Medio ambiente, Educación y Hacienda para la Secretaría General.

Que en virtud de dicha derogatoria, se hace necesario asignar los empleos y servidores que no harán parte de la Secretaría General en las dependencias a donde pertenecían antes de que entrara en vigencia la Resolución 2066 de 2015.

Que en mérito de lo expuesto.

RESUELVE:

Artículo 1°: Modificar la Resolución 2017 de 2015 mediante la cual se conformó la planta de empleos de la Secretaría de Salud, en el sentido de que los siguientes empleos con sus respectivos servidores hacen parte de dicha Secretaría, así:

No.	Posición	Cédula	Apellidos y Nombres	Código Empleo	Nombre del Empleo	Tipo Vinculación	Estado	Subsecretaría
1	2012219	98461042	MARTINEZ CHAVARRIAGA OSCAR ARLEY	21902036	PROFESIONAL UNIVERSITARIO	PROVISIONALIDAD	Titular	SUBSECRETARIA DE SALUD PUBLICA
2	2012283	39443412	SANCHEZ GOMEZ LUZ STELLA	21902036	PROFESIONAL UNIVERSITARIO	CARRERA ADTIVA.	Titular	SUBSECRETARIA ADMINISTRATIVA Y FINANCIERA DE SALUD

Artículo 2°: Modificar la Resolución 2021 de 2015, por medio de la cual se conformó la planta de empleos de la Secretaría de Juventud, en el sentido de que el siguiente empleo con su respectivo servidor hace parte de dicha Secretaría, así:

No	Posición	Cédula	Apellidos y Nombres	Código Empleo	Nombre del Empleo	Tipo Vinculación	Estado	Unidad Organizativa - Subsecretaría
1	2016178	8,013,486	MESA GUZMAN RODRIGO ALFONSO	21902036	PROFESIONAL UNIVERSITARIO	PROVISIONALIDAD	Titular	UNIDAD ADMINISTRATIVA - SECRETARIA DE LA JUVENTUD

Artículo 3°: Modificar la Resolución 2022 de 2015, por medio de la cual se conformó la planta de empleos de la Secretaría de las Mujeres, en el sentido de que el siguiente empleo con su respectivo servidor hace parte de dicha Secretaría, así:

1	No.	Posición	Cédula	Apellidos y Nombres	Código Empleo	Nombre del Empleo	Tipo Vinculación	Estado	Unidad Organizativa - Subsecretaría
	1	2016399	98537290	VALENCIA HINCAPIE LUIS ALBERTO	21902133	PROFESIONAL UNIVERSITARIO	CARRERA ADTIVA.	Encargo	EQUIPO DE APOYO ADMINISTRATIVO - SECRETARÍA DE LAS MUJERES

Artículo 4°: Modificar la Resolución 2043 de 2015, por medio de la cual se conformó la planta de empleos de la Secretaría de Gobierno y Derechos Humanos, en el sentido de que los siguientes empleos con sus respectivos servidores hacen parte de dicha Secretaría, así:

No	Posición	Cédula	Apellidos y Nombres	Código Empleo	Nombre del Empleo	Tipo Vinculación	Estado	Unidad Organizativa - Subsecretaría
1	2013501	71668226	MESA RUBIANO JUAN GUILLERMO	22203021	PROFESIONAL ESPECIALIZADO	CARRERA ADTIVA.	Titular	UNIDAD DE PREVENCIÓN, PROTECCIÓN Y PROMOCIÓN DE DERECHOS HUMANOS - SUBSECRETARÍA DE DERECHOS HUMANOS



No.	Posición	Cédula	Apellidos y Nombres	Código Empleo	Nombre del Empleo	Tipo Vinculación	Estado	Unidad Organizativa - Subsecretaría
1	2013501	32541992	RENDON ARIAS FABIOLA ESTELA	22203021	PROFESIONAL ESPECIALIZADO	CARRERA ADTIVA.	Encargo	UNIDAD DE PREVENCIÓN, PROTECCIÓN Y PROMOCIÓN DE DERECHOS HUMANOS - SUBSECRETARÍA DE DERECHOS HUMANOS
2	2016270	71747884	GALLEGO RODRIGUEZ VICTOR HUGO	21902036	PROFESIONAL UNIVERSITARIO	CARRERA ADTIVA.	Encargo	SUBSECRETARIA DE GOBIERNO LOCAL Y CONVIVENCIA

Artículo 5°: Modificar la Resolución 2046 de 2015, por medio de la cual se conformó la planta de empleos de la Secretaría Privada, en el sentido en que el siguiente empleo con su respectivo servidor hace parte de dicha Secretaría, así:

No.	Posición	Cédula	Apellidos y Nombres	Código Empleo	Nombre del Empleo	Tipo Vinculación	Estado	Unidad Organizativa - Subsecretaría
1	2012013	32209279	PINZON MUÑOZ INDIRA COLOMBIA	21902036	PROFESIONAL UNIVERSITARIO	CARRERA ADTIVA.	Comisión En Otra Entidad	SECRETARIA PRIVADA
	2012013	43261135	RESTREPO ZULUAGA SANDRA VERONICA	21902036	PROFESIONAL UNIVERSITARIO	CARRERA ADTIVA.	Encargo	SECRETARIA PRIVADA

Artículo 6°: Modificar la Resolución 2061 de 2015, mediante la cual se conformó la planta de empleos de la Secretaría de Infraestructura Física, en el sentido de que los siguientes empleos con sus respectivos servidores hacen parte de dicha Secretaría, así:

No.	Posición	Cédula	Apellidos y Nombres	Código Empleo	Nombre del Empleo	Tipo Vinculación	Estado	Unidad Organizativa - Subsecretaría
1	2004879	71,626,060	RESTREPO RIVAS LUIS FERNANDO	21902036	PROFESIONAL UNIVERSITARIO	CARRERA ADTIVA.	Titular	SECRETARIA DE INFRAESTRUCTURA FISICA
ı	2004879	71,876,496	RESTREPO JARAMILLO OSCAR EDILBERTO	21902036	PROFESIONAL UNIVERSITARIO	CARRERA ADTIVA.	Encargo	SECRETARIA DE INFRAESTRUCTURA FISICA
2	2012037	71,656,805	AGUILAR MUÑOZ JOHNY	21902036	PROFESIONAL UNIVERSITARIO	CARRERA ADTIVA.	Titular	SECRETARIA DE INFRAESTRUCTURA FISICA
3	2016591	98,580,846	RIVERA ZAPATA ROBERT ALIRIO	21902036	PROFESIONAL UNIVERSITARIO	CARRERA ADTIVA.	Encargo	SUBSECRETARIA DE CONSTRUCCION Y MANTENIMIENTO DE LA INFRAESTRUCTURA FISICA
4	2016588	43,265,268	MOSCOSO GARCIA SANDRA KATERINE	21902036	PROFESIONAL UNIVERSITARIO	PROVISIONALIDAD	Titular	SUBSECRETARIA DE PLANEACION DE LA INFRAESTRUCTURA FISICA

Artículo 7°: Modificar la Resolución 2062 de 2015, mediante la cual se conformó la planta de empleos de la Secretaría de Seguridad, en el sentido de que el siguiente empleo con su respectivo servidor hace parte de dicha Secretaría, así:

No.	Posición	Cédula	Apellidos y Nombres	Código Empleo	Nombre del Empleo	Tipo Vinculación	Estado	Unidad Organizativa - Subsecretaría
1	2015985	34985978	GUZMAN MORENO MATILDE EUGENIA	21902036	PROFESIONAL UNIVERSITARIO	CARRERA ADTIVA.	Encargo	EQUIPO GESTION DE SEGURIDAD TERRITORIAL - SUBSECRETARIA OPERATIVA DE LA SEGURIDAD

Artículo 8°: Modificar la Resolución 2063 de 2015, mediante la cual se conformó la planta de empleos del Departamento Administrativo de Gestión del Riesgo de Desastres, en el sentido de que el siguiente empleo con su respectivo servidor hace parte de dicho Departamento Administrativo, así:

No.	Posición	Cédula	Apellidos y Nombres	Código Empleo	Nombre del Empleo	Tipo Vinculación	Estado	Unidad Organizativa - Subsecretaría
1	2016568	71,603,334	CANO JIMENEZ MAURICIO	22203021	PROFESIONAL ESPECIALIZADO	CARRERA ADTIVA.	Encargo	DEPTO. ADMTIVO. DE GESTION DEL RIESGO DE DESASTRES

Artículo 9°: Modificar la Resolución 2064 de 2015, por medio de la cual se conformó la planta de empleos de la Secretaría de Cultura Ciudadana, en el sentido de que los siguientes empleos con sus respectivos servidores hacen parte de dicha Secretaría, así:

No.	Posición	Cédula	Apellidos y Nombres	Código Empleo	Nombre del Empleo	Tipo Vinculación	Estado	Unidad Organizativa - Subsecretaría
1	2012060	70056080	CONGOTE ZULETA JOSE DE JESUS	20606063	LIDER DE PROGRAMA	CARRERA ADTIVA.	Titular	SECRETARIA DE CULTURA CIUDADANA
2	2000628	42756814	ARENAS RUA CECILIA INES	22203006	PROFESIONAL ESPECIALIZADO	CARRERA ADTIVA.	Encargo	UNIDAD ADMINISTRATIVA - SECRETARIA DE CULTURA CIUDADANA

Artículo 10°: Modificar la Resolución 2067 de 2015, por medio de la cual se conformó la planta de empleos de la Secretaría de Inclusión Social y Familia, en el sentido de que los siguientes empleos con sus respectivos servidores hacen parte de dicha Secretaría, así:

No.	Posición	Cédula	Apellidos y Nombres	Código Empleo	Nombre del Empleo	Tipo Vinculación	Estado	Unidad Organizativa - Subsecretaría
1	2016337	70600814	CALLE POSADA CARLOS CESAR DE JESUS	21902036	PROFESIONAL UNIVERSITARIO	PROVISIONALIDAD	Titular	UNIDAD ADMINISTRATIVA - SECRETARÍA DE INCLUSIÓN SOCIAL Y FAMILIA
2	2009714	43,620,867	FERNANDEZ VELEZ ERICA EVIDALIA	21902036	PROFESIONAL UNIVERSITARIO	CARRERA ADTIVA.	Titular	UNIDAD ADMINISTRATIVA - SECRETARÍA DE INCLUSIÓN SOCIAL Y FAMILIA
2	2009714	71,591,198	JIMENEZ LOPEZ DE MESA GUSTAVO	21902036	PROFESIONAL UNIVERSITARIO	PROVISIONALIDAD	Titular	UNIDAD ADMINISTRATIVA - SECRETARÍA DE INCLUSIÓN SOCIAL Y FAMILIA

Artículo 11°: Modificar la Resolución 2068 de 2015, por medio de la cual se conformó la planta de empleos de la Secretaría de Medio Ambiente, en el sentido de que el siguiente empleo hace parte de dicha Secretaría, así:

No.	Posición	Cédula	Apellidos y Nombres	Código Empleo	Nombre del Empleo	Tipo Vinculación	Estado	Unidad Organizativa - Subsecretaría
1	2013667	V-Def	Vacante Definitiva	21902036	PROFESIONAL UNIVERSITARIO	V-Def	V-Def	UNIDAD ADMINISTRATIVA - SECRETARIA
								DEL MEDIO AMBIENTE

Artículo 12°: Modificar la Resolución 2069 de 2015, por medio de la cual se conformó la planta de empleos de la Secretaría de Educación, en el sentido de que los siguientes empleos hacen parte de dicha Secretaría, así:

No	. Posición	Cédula	Apellidos y Nombres	Código Empleo	Nombre del Empleo	Tipo Vinculación	Estado	Unidad Organizativa - Subsecretaría
1	2008872	15379693	MUNOZ OSORIO JUAN FERNANDO	21902036	PROFESIONAL UNIVERSITARIO	CARRERA ADTIVA.	Titular	SUBSECRETARIA ADMINISTRATIVA Y FINANCIERA DE EDUCACION
2	2013713	3602588	PEMBERTHY GUTIERREZ LEONARDO ANTONIO	21902036	PROFESIONAL UNIVERSITARIO	CARRERA ADTIVA.	Titular	SUBSECRETARIA ADMINISTRATIVA Y FINANCIERA DE EDUCACION



No.	Posición	Cédula	Apellidos y Nombres	Código Empleo	Nombre del Empleo	Tipo Vinculación	Estado	Unidad Organizativa - Subsecretaría
2	2013713	70950751	OSORIO AGUDELO IVAN DE JESUS	21902036	PROFESIONAL UNIVERSITARIO	CARRERA ADTIVA.	Encargo	SUBSECRETARIA ADMINISTRATIVA Y FINANCIERA DE EDUCACION
2	2013715	2013715 71746160 GARCIA ARANGO GUSTAVO ADOLFO		21902036	PROFESIONAL UNIVERSITARIO	CARRERA ADTIVA.	Titular	SUBSECRETARIA ADMINISTRATIVA Y FINANCIERA DE EDUCACION
3	2013715	3583757	DUQUE CASTRILLON HERNAN DE JESUS	21902036	PROFESIONAL UNIVERSITARIO	CARRERA ADTIVA.	Encargo	SUBSECRETARIA ADMINISTRATIVA Y FINANCIERA DE EDUCACION
	2013714	71790873	RICO PALACIO DUVAN DE JESUS	21902036	PROFESIONAL UNIVERSITARIO	CARRERA ADTIVA.	Titular	SUBSECRETARIA ADMINISTRATIVA Y FINANCIERA DE EDUCACION
4	2013714	42989822	RODRIGUEZ AGUIRRE CLARA INES	21902036	PROFESIONAL UNIVERSITARIO	CARRERA ADTIVA.	Encargo	SUBSECRETARIA ADMINISTRATIVA Y FINANCIERA DE EDUCACION

Artículo 13°: Modificar la Resolución 2098 de 2015, mediante la cual se conformó la planta de empleos de la Secretaría de Hacienda, en el sentido de que los siguientes empleos con sus respectivos servidores hacen parte de dicha Secretaría, así:

No.	Posición	Cédula	Apellidos y Nombres	Código Empleo	Nombre del Empleo	Tipo Vinculación	Estado	Unidad Organizativa - Subsecretaría
1	2002229	71590466	ALVAREZ ALVAREZ REINALDO DE JESUS	21902036	PROFESIONAL UNIVERSITARIO	CARRERA ADTIVA.	Encargo	SECRETARIA DE HACIENDA
2	2016488	43510974	RIVERA RESTREPO MABEL ELENA	21902036	PROFESIONAL UNIVERSITARIO	PROVISIONALIDAD	Titular	SUBSECRETARIA DE INGRESOS
3	2002234	70127543	CARDONA ALZATE GONZALO ALONSO	21902036	PROFESIONAL UNIVERSITARIO	CARRERA ADTIVA.	Titular	SECRETARIA DE HACIENDA
4	2001615	32515614	CORREA URIBE MARTA ELENA	21902036	PROFESIONAL UNIVERSITARIO	CARRERA ADTIVA.	Titular	SUBSECRETARIA DE TESORERIA
5	2000351	70513806	GALLEGO CHICA PAULO IGNACIO	21902036	PROFESIONAL UNIVERSITARIO	CARRERA ADTIVA.	Titular	SUBSECRETARIA DE TESORERIA

Artículo 14°: Las funciones, requisitos, categoría salarial y demás especificaciones técnicas de los empleos contemplados en los artículos anteriores serán los existentes en el manual de funciones y competencias laborales del Municipio de Medellín.

Artículo 15°: Esta Resolución rige a partir de la fecha de su publicación en la Gaceta Oficial del Municipio de Medellín y deroga las disposiciones que le sean contrarias.

PUBLÍQUESE Y CÚMPLASE

Dado en Medellín, a los cuatro (4) días del mes diciembre de 2015

ALEJANDRA VANEGAS VALENCIA

Secretaria de Gestión Humana y Servicio a la Ciudadanía

RESOLUCIÓN 3838 DE 2015 (Diciembre 18)

Por medio del cual se aclaran los Artículos 405 y 406 de la Resolución 3728 de 2015, los Artículos 143 y 149 de la Resolución 3730 de Diciembre 14 de 2015 y se adiciona el Artículo 95 de éste último Acto Administrativo.

LA SECRETARIA DE GESTIÓN HUMANA Y SERVICIO A LA CIUDADANÍA,

En uso de sus atribuciones legales, y

CONSIDERANDO:

Que mediante el Decreto 912 de junio 5 de 2015, se delegó en la Secretaría de Gestión Humana y Servicio a la Ciudadanía, la administración y control de la planta global de empleos del sector central, la organización con carácter permanente o transitorio de las unidades, equipos y grupos internos de trabajo y la redistribución del personal del Municipio de Medellín del nivel central, con el fin de atender las necesidades del servicio y cumplir con eficacia y eficiencia, los objetivos, políticas y programas de las dependencias centrales.

Que mediante Resolución 3728 de Diciembre 14 de 2015, se ajustan los manuales de funciones y competencias laborales para los empleos del nivel Profesional (Profesional Universitario, Profesional Universitario Área Salud, Profesional Especializado Área Salud, Médico General, Médico Especialista, Enfermero, Enfermero Especialista, Odontólogo, Odontólogo Especialista) en el Municipio de Medellín.

Que en el artículo 405 de la Resolución en mención, se homologa el empleo Profesional Universitario Área Salud, códigos 23702002, posición 23702005, posiciones 2012212 y 2012215 respectivamente, por el empleo Profesional Universitario, código 21902369, debiendo homologarse por el empleo Profesional Universitario código 21902390.

Que así mismo, en el artículo 406 de la misma Resolución, se determina cual será el manual de funciones del empleo Profesional Universitario, código 21902369, modificado en el artículo 405 de dicho Acto Administrativo, debiendo el 21902390 el código correcto.

Que mediante Resolución 3730 de diciembre 14 de 2015, se ajustaron los manuales de funciones y competencias laborales para los empleos del nivel Asesor y Profesional (Líder de Programa, Líder de Proyecto, Profesional Especializado, Comisario de Familia e Inspector de Policía de 1ª Categoría) en el Municipio de Medellín.

Que en el artículo 95 de dicha Resolución, se determinó el manual de funciones y competencias laborales del empleo Líder de Programa código 20606178.

Que por ser un empleo de carrera administrativa del más alto nivel requiere para su desempeño de unas calidades y competencias, así como de experticia en las funciones del mismo por lo cual los requisitos de estudio y experiencia no pueden ser compensados.

Que se hace necesario aclarar los artículos 405 y 406 de la Resolución 3728 de diciembre 14 de 2015, en el sentido de que el empleo Profesional Universitario Área Salud, códigos 23702002, posición 23702005, posiciones 2012212 y 2012215 respectivamente, es homologado al empleo Profesional Universitario código 21902390 y no al empleo 21902369 como se mencionó en dicha Resolución.

Que así mismo, es necesario adicionar los requisitos de estudio y experiencia del empleo Líder de Programa código 20606178, a fin de garantizar la idoneidad de los servidores que se desempeñen en este empleo.

Que mediante Resolución 3730 de Diciembre 14 de 2015, Por medio de la cual se ajustan los manuales de funciones y competencias laborales para los empleos del nivel Asesor y Profesional (Líder de Programa, Líder de Proyecto, Profesional Especializado, Comisario de Familia e Inspector de Policía de 1ª Categoría) en el Municipio de Medellín.

Que en los artículos 143 y 149 de la Resolución en mención, se determinan cuáles serán los manuales de funciones y competencias laborales para los empleos Líder de Programa código 20606109 y 20606015 respectivamente.

Que el ejercicio de los empleos en mención implica la administración y el manejo directo de bienes, dineros y/o valores del Estado y por tal razón la naturaleza de los mismos es de Libre Nombramiento y Remoción.

Que se hace necesario aclarar los artículos 143 y 149 de la Resolución 3730 de diciembre 14 de 2015, en el sentido de que la naturaleza de dichos empleos es de Libre Nombramiento y Remoción y no de Carrera Administrativa, como quedó registrado.

Que en mérito de lo expuesto,

RESUELVE:

Artículo 1º: Aclarar el artículo 405 y 406 de la Resolución 3728 de diciembre 14 de 2015, en cuanto a que el empleo Profesional Universitario Área Salud, códigos 23702002, 23702005, posiciones 2012212 y 2012215

respectivamente, es homologado por el empleo Profesional Universitario código 21902390 y no por el empleo Profesional Universitario código 21902369 como quedó registrado.

Artículo 2°: Adicionar el artículo 95 de la Resolución 3730 de diciembre 14 de 2015, en el cual se determina el manual de funciones y competencias laborales del empleo Líder de Programa código 20606178, al cual se le deberá agregar en los requisitos de formación académica y experiencia, lo siguiente:

EQUIVALENCIA:

No le aplican las equivalencias establecidas en las normas vigentes de Carrera Administrativa.

Artículo 3º: Aclarar los Artículos 143 y 149 de la Resolución 3730 de 2015, en cuanto a que la naturaleza de los empleos Líder de Programa códigos 20606109 y 20606015, respectivamente es de Libre Nombramiento y Remoción y no de Carrera Administrativa como quedó registrado.

Artículo 4°: La presente Resolución rige a partir de la fecha de su publicación en la Gaceta Oficial del Municipio de Medellín y deroga las disposiciones que le sean contrarias.

PUBLIQUESE Y CÚMPLASE

Dado en Medellín, a los diez y ocho (18) días del mes diciembre de 2015

ALEJANDRA VANEGAS VALENCIA

Secretaria de Gestión Humana y Servicio a la Ciudadanía

RESOLUCIÓN 3945 DE 2015 (Diciembre 29)

Por medio del cual se realizan unos traslados en la Administración Municipal, y se dictan otras disposiciones.

EL ALCALDE DE MEDELLÍN,

En uso de sus facultades constitucionales y legales, especialmente las conferidas por el artículo 209 del artículo 315 de la Constitución Política, la Ley 909 de 2004 y sus decretos reglamentarios, el Decreto Nacional 1083 de 2015 y los Decretos Municipales 883 y 912 de 2015, y

CONSIDERANDO:

Que el Municipio de Medellín tiene una Planta Global de empleos, de acuerdo con lo definido en el artículo 115 de

la Ley 489 de 1998 "Por la cual se dictan normas sobre la organización y funcionamiento de las entidades del orden nacional, se expiden las disposiciones, principios y reglas generales para el ejercicio de las atribuciones previstas en los numerales 15 y 16 del artículo 189 de la Constitución Política y se dictan otras disposiciones", en concordancia con el artículo 209 de la Constitución Política.

Que la condición de planta global que caracteriza la estructura de empleos del Municipio de Medellín permite traslados en forma horizontal, respetando su perfil, asignación salarial y la esencia de las funciones de los empleos a trasladar.

Que ha dispuesto la Ley que es función del nominador ubicar los cargos en las distintas dependencias de la organización, de conformidad con la estructura orgánica, las necesidades del servicio y los planes y programas que se deban ejecutar.

Que el Decreto 1083 de 2015, "Por medio del cual se expide el Decreto Único Reglamentario del Sector de Función Pública", establece entre otros los criterios para realizar los Traslados o permutas de empleos.

Que en el Decreto antes citado, en su Título 5 "Administración de personal y situaciones administrativas de los empleados públicos de las entidades del orden nacional y Territorial", Capítulo 9 "Los Movimientos", artículo 2.2.5.9.1 Movimientos de personal. El movimiento del personal en servicio se puede hacer por: 1.Traslado o permuta, 2. Encargo, y 3. Ascenso.

Que además en el artículo 2.2.5.9.2 Traslado o permuta. Indica que hay traslado cuando se provee, con un empleado en servicio activo, un cargo vacante definitivamente, con funciones afines al que desempeña, de la misma categoría, y para el cual se exijan requisitos mínimos similares. También hay traslado cuando la administración hace permutas entre empleados que desempeñen cargos de funciones afines o complementarias, que tengan la misma categoría y para los cuales se exijan requisitos mínimos similares para su desempeño. Los traslados o permutas podrán hacerse dentro de la misma entidad o de un organismo a otro, con el lleno de los requisitos previstos en el presente decreto. Cuando se trate de traslados o permutas entere organismos, la providencia deberá ser autorizada por los jefes de las entidades en donde se produce. Los reglamentos de las carreras especiales, en lo referente a los traslados y permutas, se ajustarán a lo que se dispone en éste Título.

Que a su vez en el artículo 2.2.5.9.3 Condiciones del traslado. El traslado se podrá hacer por necesidades del servicio, siempre que ello no implique condiciones menos favorables para el empleado. Podrá hacerse también cuando sea solicitado por los empleados interesados, y siempre que el movimiento no cause perjuicios al servicio.

Que en el Titulo 12 "Reformas de las Plantas de Empleos", del mismo Decreto 1083, en su artículo 2.2.12.1, expresa que la modificación de la Planta de Empleos debe fundamentarse en necesidades del servicio o en razones de Modernización de la Administración, y con base, entre otros, en los siguientes aspectos: Literal 2: Cambios en la misión y objeto social o en las funciones generales de la Entidad, Literal 3: Traslado de funciones o competencias de un organismo a otro., 4: Supresión fusión o creación de dependencias o modificación de sus funciones, Literal 5: Mejoramiento o introducción de procesos, producción de bienes o prestación de servicios y Literal 10: Mejoramiento de los niveles de eficiencia, eficacia y economía de la Entidad.

Que mediante el Decreto 883 de junio 3 de 2015, se adecúa la Estructura de la Administración Municipal de

Medellín, las funciones de sus organismos, dependencias y entidades descentralizadas, se modifican unas entidades descentralizadas y se dictan otras disposiciones.

Que en el mencionado Decreto se creó en la estructura administrativa del nivel central de la Administración Municipal, entre otras, la Secretaría de Gobierno y Derechos Humanos, la cual para cumplir las responsabilidades, funciones, procesos, planes, programas y proyectos a ella asignada, requiere contar con una planta de personal, que permita la operación, integración y el fortalecimiento, para alcanzar los fines del Estado y la ejecución del Plan de Desarrollo, manteniendo lo consagrado en el artículo 115 de la Ley 489 de 1998, el cual expresa que la condición de planta global que caracteriza la estructura de empleos de la Administración Municipal permite un traslado en forma horizontal, respetando su perfil, asignación salarial y la naturaleza de las funciones, a fin de fortalecer los procesos estratégicos, misionales y de apoyo de la dependencia.

Que mediante el Decreto 912 de junio 5 de 2015, se delegó en la Secretaría de Gestión Humana y Servicio a la Ciudadanía, la administración y control de la planta global de empleos del sector central, la organización con carácter permanente o transitorio de las unidades, equipos y grupos internos de trabajo y la redistribución del personal del Municipio de Medellín del nivel central, con el fin de atender las necesidades del servicio y cumplir con eficacia y eficiencia, los objetivos, políticas y programas de las dependencias centrales.

Que la Administración Municipal requiere profesionales analíticos, capaces de elaborar planes estratégicos, integrar los programas y desarrollar los proyectos para la toma de decisiones y optimización de los recursos de las mismas, buscando además que sus servidores apliquen los conocimientos adquiridos con el fin de atender las necesidades del servicio y cumplir con eficacia y eficiencia los objetivos y funciones de la entidad.

Que la permuta de los empleos Profesional Universitario, se ajusta perfectamente al desarrollo de los objetivos establecidos para cada uno de ellos, además ajustándose a lo establecido por la Ley 1098 de 2006, en cuanto a los perfiles de los servidores que conformarán las Comisarías de Familia.

Que el traslado a realizar en el presente Acto Administrativo, está fundamentado en la solicitud elevada por correo electrónico de fecha 23 de diciembre de 2015, suscrito por la doctora Claudia Patricia Wilches Mesa, Secretaria de Despacho de la Secretaría de Gobierno y Derechos Humanos, quien requiere se realice la permuta de empleos y sus respectivos servidores.

Que consecuente con todo lo anterior, y en mérito de lo expuesto,

RESUELVE:

Artículo 1º: Trasladar una (1) plaza del empleo Profesional Universitario **código** 21902049, posición 2009366,

adscrito a la Unidad Comisarías de Familia, Subsecretaría Gobierno Local y Convivencia, Secretaría de Gobierno y Derechos Humanos, para que dependa directamente de la Unidad Convivencia Ciudadana de la misma Subsecretaría y Secretaría.

Artículo 2: Trasladar al servidor ANDRÉS FELIPE BURITICÁ VARGAS, identificado con la cédula de ciudadanía número 98.669.842, quien se desempeña en calidad de Provisionalidad en el empleo Profesional Universitario código 21902049, posición 2009366, adscrito a la Unidad Comisarías de Familia, Subsecretaría Gobierno Local y Convivencia, Secretaría de Gobierno y Derechos Humanos, y su titular CIELO MARIA SERNA LONDOÑO, identificada con la cédula de ciudadanía número 43.519.572, quien se encuentra encargada en otro empleo, para que dependa directamente de la Unidad Convivencia Ciudadana de la misma Subsecretaría y Secretaría.

Artículo 3º: Trasladar una (1) plaza del empleo Profesional Universitario **código** 21902049, posición 2004493, adscrito a la Unidad Convivencia Ciudadana, Subsecretaría Gobierno Local y Convivencia, Secretaría de Gobierno y Derechos Humanos, para que dependa directamente de la

Unidad Comisarías de Familia, de la misma Subsecretaría y Secretaría.

Artículo 4: Trasladar al servidor BYRNE RAMÍREZ URREGO, identificado con la cédula de ciudadanía número 98.671495, quien se desempeña en calidad de Provisionalidad en el empleo Profesional Universitario código 21902049, posición 2004493, adscrito a la Unidad Convivencia Ciudadana, Subsecretaría Gobierno Local y Convivencia, Secretaría de Gobierno y Derechos Humanos, y su titular NATALIA DEL PILAR CAMPUZANO ROJAS, identificada con la cédula de ciudadanía número 43.259.185, quien se encuentra encargada en otro empleo, para que dependa directamente de la Unidad Comisarías de Familia, de la misma Subsecretaría y Secretaría.

Artículo 5°: El manual de funciones y competencias labores del empleo trasladado en la presente Resolución, será el existente para este empleo en el Municipio de Medellín.

Artículo 6°: Esta Resolución rige a partir de la fecha de su publicación en la Gaceta Oficial del Municipio de Medellín y deroga las disposiciones que le sean contrarias.

PUBLÍQUESE Y CÚMPLASE

Dado en Medellín, a los veintinueve (29) día del mes de diciembre de 2015

ALEJANDRA VANEGAS VALENCIA

Secretaria de Gestión Humana y Servicio a la Ciudadanía



Alcaldía de Medellín



Alcaldía de Medellín